UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K/A

(Amendment No. 1)

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the **Securities Exchange Act of 1934**

Date of Report (Date of earliest event reported): October 4, 2006

Great Lakes Dredge & Dock Corporation

(Exact name of Registrant as specified in its charter)

	Delaware	001-33225	20-5336063
	(State or other jurisdiction of Incorporation or Organization)	(Commission File Number)	(I.R.S. Employer Identification No.)
	incorporation of Organization)	The Number)	identification 140.)
		2122 York Road	
		Oak Brook, Illinois 60523	
		(Address of Principal Executive Offices)	
		(630) 574-3000	
	(Re	gistrant's telephone number, including area c	ode)
Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:			
	Written communications pursuant to Rule 425 under the	Securities Act (17 CFR 230.425)	
	Soliciting material pursuant to Rule 14a-12 under the Ex	xchange Act (17 CFR 240.14a-12)	
	Pre-commencement communications pursuant to Rule 1	4d-2(b) under the Exchange Act (17	CFR 240.14d-2(b))
	Pre-commencement communications pursuant to Rule 1	3e-4(c) under the Exchange Act (17	CFR 240.13e-4(c))

Preliminary Note:

On October 4, 2006, Great Lakes Dredge & Dock Corporation (the "Company") filed a Current Report on Form 8-K (the "Original Form 8-K"). Attached as Exhibit 10.1 to the Original Form 8-K was a copy of the Company's International Letter of Credit Agreement (the "Agreement"), dated September 29, 2006, by and among the Company and Wells Fargo HSBC Trade Bank. The schedules and exhibits to the Agreement were inadvertently not included with the Agreement, as filed with the Original Form 8-K. A complete, executed copy of the Agreement, including all schedules and exhibits, is being filed as Exhibit 10.1 to this Current Report on Form 8-K/A (the "Amendment").

The Company is filing the Amendment for the sole purpose of adding the schedules and exhibits to the Agreement. In all other respects, the Original Form 8-K remains the same. The information previously reported in the Original Form 8-K is hereby incorporated by reference into the Amendment except to the extent such information is amended by this Amendment.

Item 9.01 - Financial Statements and Exhibits

(d) Exhibits

The following exhibit is furnished herewith:

10.1 International Letter of Credit Agreement, dated September 29, 2006, by and among Great Lakes Dredge & Dock Corporation and Wells Fargo HSBC Trade Bank.

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

GREAT LAKES DREDGE & DOCK CORPORATION

Date: August 17, 2010 /s/ Deborah A. Wensel

Deborah A. Wensel Senior Vice President and Chief Financial Officer

EXHIBIT INDEX

Number Exhibi

10.1 International Letter of Credit Agreement, dated September 29, 2006, by and among Great Lakes Dredge & Dock Corporation and Wells Fargo HSBC Trade Bank.

INTERNATIONAL LETTER OF CREDIT AGREEMENT

Dated as of September 29, 2006

among

GREAT LAKES DREDGE & DOCK CORPORATION,
GREAT LAKES DREDGE & DOCK COMPANY, LLC,

and

WELLS FARGO HSBC TRADE BANK, N.A.

TABLE OF CONTENTS

A DELCLE I	CERTAIN DEFINITIONS AND OTHER INTERPRETATE PROVISIONS	PAGE
ARTICLE I. 1.1	CERTAIN DEFINITIONS AND OTHER INTERPRETIVE PROVISIONS Definitions	1
		1
1.2	Other Interpretive Provisions	1
1.3	Accounting Terms	2
1.4	Times of Day	2
ARTICLE II.	LETTERS OF CREDIT	2
2.1	Commitment	2
2.2	International Borrowing Base	3
2.3	Letters of Credit	4
2.4	Letter of Credit Requests	5
2.5	Agreement to Repay Letter of Credit Drawings	5
2.6	Conflict Between Applications and Agreement	7
2.7	Increased Costs; Increased Capital	7
2.8	Default Rate	8
2.9	Facility Subject to Ex-Im Bank Rules	8
2.10	Currency Equivalents	9
ARTICLE III.	CASH COLLATERAL, PREPAYMENTS AND OTHER PAYMENTS	9
3.1	Deposit of Cash Collateral and Required Prepayments	9
3.2	[RESERVED]	9
3.3	Place of Payment or Prepayment	9
3.4	Prepayment Premium or Penalty	10
3.5	Taxes	10
3.6	Reduction or Termination of the Commitment	10
ARTICLE IV.	FEES	10
4.1	Facility Fee	10
4.2	Letter of Credit Fees	10
4.3	Fees Not Interest; Nonpayment	11
ARTICLE V.	PERMITTED PURPOSES OF LETTERS OF CREDIT	11
ARTICLE VI.	REPRESENTATIONS AND WARRANTIES	11
6.1	Organization and Qualification	11

6.2	Financial Statements; Positive Tangible Net Worth	11
6.3	Litigation; Contingent Liabilities	12
6.4	Default	12
6.5	Title to Assets; Ownership	12
6.6	Authorization, Validity, Etc.	13
6.7	Intentionally Omitted	13
6.8	Taxes	13
6.9	Conflicting or Adverse Agreements or Restrictions	13
6.10	Information	13
6.11	No Consent	14
6.12	Environmental Matters	14
6.13	Debt	14
6.14	Compliance with Laws	14
6.15	Suspension and Debarment, Etc.	15
ARTICLE VII.	CONDITIONS	15
7.1	Representations True and No Defaults	15
7.2	Terms of Sale	15
7.3	Governmental Approvals	15
7.4	Letter of Credit Documents	16
7.5	Required Initial Documents and Certificates	16
7.6	Ex-Im Bank Acknowledgment, Etc.	18
7.7	Post-Closing Lien Search	18
ARTICLE VIII.	AFFIRMATIVE COVENANTS	18
8.1	Financial Statements and Information	18
8.2	Books and Records	19
8.3	Insurance	19
8.4	Inspection of Property and Records; Audits of Collateral; Etc.	20
8.5	Notice of Certain Matters	20
8.6	Security and Further Assurances	22
8.7	Establishment of Cash Collateral Account	22
8.8	Maintenance of Property	22
8.9	Corporate Existence; Payment of Taxes and Claims; Compliance With Laws	22
8.10	Assignment of International Letter of Credit Proceeds	23
8.11	Environmental Matters	23
8.12	Controlling Affiliates	23
8.13	Assembly of Export Order Summaries	23
8.14	Delivery of Domestic Loan Documents	24
8.15	Reaffirmation of Borrower	24
ARTICLE IX.	NEGATIVE COVENANTS	24

24

9.1

Liens

9.2	Merger, Consolidation, Acquisitions, Sales of Assets, Etc.	24
9.3	Conduct of Business	24
9.4	Investments	24
9.5	Financial Covenants	24
9.6	Debt	26
9.7	Affiliate Transactions	26
9.8	Restricted Payments	26
9.9	Suspension and Debarment, Etc.	26
9.10	Debt of Eligible Joint Ventures	27
ARTICLE X.	EVENTS OF DEFAULT; REMEDIES	27
10.1	Events of Default	27
10.2	Other Remedies	31
10.3	Collateral Account	31
10.4	Exit Date Event of Default	31
10.5	Remedies Cumulative	31
ARTICLE XI.	MISCELLANEOUS	32
11.1	Waivers, Amendments	32
11.2	Reimbursement of Expenses	32
11.3	Notices	32
11.4	Governing Law	33
11.5	Survival of Representations, Warranties and Covenants	33
11.6	Counterparts	33
11.7	Separability	33
11.8	Limitation of Liability	34
11.9	Set-off	34
11.10	Sale or Assignment	34
11.11	Interest	34
11.12	Indemnification	35
11.13	Payments Set Aside	36
11.14	Credit Agreement Controls	36
11.15	Amendment of Financial Covenants	36
11.16	Waiver of Jury Trial	36
11.17	USA Patriot Act Notice	36
11.18	Time of the Essence	37
11.19	Final Agreement	37
11.20	Release of Collateral and Guarantors	37
11.21	Additional Guarantors	37

Exhibit A - Definitions
Exhibit B - Form of Note

Exhibit C - Form of Weekly International Borrowing Base Certificate, together with Annexes I and II

Exhibit D - Form of Compliance Certificate
Exhibit E - Form of Letter of Credit Request

Exhibit F - Form of Guaranty

Exhibit G - Form of Joinder to Guaranty

Exhibit H - Form of Standby Letter of Credit Agreement

Exhibit I - Form of Reaffirmation, Ratification and Assumption Agreement

Exhibit J - Form of Deposit Account Control Agreement

Schedule 1.1 - Existing Letters of Credit

Schedule 6.1 - Subsidiaries Schedule 6.3 - Litigation

Schedule 6.5 - Capital Stock of Borrower; Control Affiliates

Schedule 8.3 - Insurance

Schedule 8.7 - Export Orders Excluded from the Cash Collateral Account

INTERNATIONAL LETTER OF CREDIT AGREEMENT

This International Letter of Credit Agreement is entered into as of September 29, 2006 among GREAT LAKES DREDGE & DOCK CORPORATION, a Delaware corporation, the Guarantors from time to time party hereto and WELLS FARGO HSBC TRADE BANK, N.A. (together with its successors and assigns, the "Bank").

The Borrower has requested that the Bank provide an international letter of credit facility, and the Bank is willing to do so on and subject to the terms and conditions set forth herein.

In consideration of the mutual covenants herein contained, the parties hereto covenant and agree as follows:

ARTICLE I. CERTAIN DEFINITIONS AND OTHER INTERPRETIVE PROVISIONS.

- 1.1 <u>Definitions</u>. Capitalized terms used in this Agreement and not otherwise defined herein shall have the meanings given to them in <u>Exhibit A</u> to this Agreement.
- 1.2 Other Interpretive Provisions. With reference to this Agreement and each other International Loan Document, unless otherwise specified herein or in such other International Loan Document:
 - (a) The definitions of terms herein shall apply equally to the singular and plural forms of the terms defined. Whenever the context may require, any pronoun shall include the corresponding masculine, feminine and neuter forms. The words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation." The word "will" shall be construed to have the same meaning and effect as the word "shall." Unless the context requires otherwise, (i) any definition of or reference to any agreement, instrument or other document (including any Organization Document) shall be construed as referring to such agreement, instrument or other document as from time to time amended, restated, supplemented or otherwise modified (subject to any restrictions on such amendments, restatements, supplements or modifications set forth herein or in any other International Loan Document), (ii) any reference herein to any Person shall be construed to include such Person's successors and assigns, (iii) the words "herein," "hereof" and "hereunder" and words of similar import when used in any International Loan Document, shall be construed to refer to such International Loan Document in its entirety and not to any particular provision thereof, (iv) all references in an International Loan Document to Articles, Sections, Exhibits and Schedules shall be construed to refer to Articles and Sections of, and Exhibits and Schedules to, the International Loan Document in which such references appear, (v) any reference to any law shall include all statutory and regulatory provisions consolidating, amending, replacing or interpreting such law and any reference to any law or regulation shall, unless otherwise specified, refer to such law or regulation as amended, modified or supplemented from time to time, and (vi) the words "asset" and "property" shall be construed to have the same meaning and effect and to refer to any and all tangible and intangible assets and properties, including cash, securities, accounts and contract r

- (b) In the computation of periods of time from a specified date to a later specified date, the word "<u>from</u>" means "<u>from and including</u>;" the words "<u>to</u>" and "<u>until</u>" each mean "<u>to but excluding</u>;" and the word "<u>through</u>" means "<u>to and including</u>."
- (c) Section headings herein and in the other International Loan Documents are included for convenience of reference only and shall not affect the interpretation of this Agreement or any other International Loan Document.

1.3 Accounting Terms.

- (a) All accounting terms not specifically defined herein shall be construed in accordance with GAAP. The Borrower shall maintain a system of accounting established and administered in accordance with sound business practices to permit preparation of financial statements in conformity with GAAP (except as may be otherwise required pursuant to <u>Section 1.3(b)</u>), and each of the financial statements described below shall be prepared from such system and records.
- (b) Except as otherwise provided herein, if any changes in accounting principles from those used in the preparation of the most recent financial statements referred to in Section 8.1 are hereafter required or permitted by the rules, regulations, pronouncements and opinions of the Financial Accounting Standards Board of the American Institute of Certified Public Accountants (or successors thereto or agencies with similar functions) and are adopted by the Borrower with the agreement of its independent certified public accountants and such changes result in a change in the method of calculation of any of the financial covenants, standards or terms found in Section 9.5 or in the related definitions of terms used therein, the parties hereto agree, subject to Section 11.15, to enter into negotiations in order to amend such provisions so as to reflect equitably such changes with the desired result that the criteria for evaluating the Borrower's financial condition shall be the same after such changes as if such changes had not been made, provided that, subject to Section 11.15, no change in GAAP that would affect the method of calculation of any of the financial covenants, standards or terms shall be given effect in such calculations until such provisions are amended, in a manner satisfactory to the Bank, so as to reflect such change in accounting principles.
- 1.4 <u>Times of Day</u>. Unless otherwise specified, all references herein to times of day shall be references to Central time (daylight or standard, as applicable).

ARTICLE II. LETTERS OF CREDIT.

2.1 Commitment.

(a) Upon the terms and conditions and relying upon the representations and warranties herein set forth, the Bank agrees to issue standby Letters of Credit for the account of the Borrower up to an aggregate face amount not exceeding at any one time outstanding the lesser of (i) \$20,000,000.00 (such amount, as it may be reduced from time to time pursuant to Section 3.6, being the Bank's "Commitment") or (ii) the International Borrowing Base.

(b) The Borrower shall execute and deliver to the Bank, to evidence its reimbursement obligations in respect of Letters of Credit, a Note which shall be (i) dated of even date herewith; (ii) in the principal amount of the Commitment; and (iii) in substantially the form attached hereto as <u>Exhibit B</u>, with the blanks appropriately completed.

2.2 International Borrowing Base.

(a) As soon as available, and in any event by 11 a.m. (Central time) each Wednesday (or, if Wednesday is not a Business Day, then on the following Business Day), the Borrower shall furnish the Bank an international borrowing base detailed scheduled report in the form of Exhibit C, which is dated as of the last Business Day of the previous calendar week (the "Weekly International Borrowing Base Certificate"); provided, however, in order to receive Credit Accommodations hereunder, the Borrower shall have delivered a Weekly International Borrowing Base Certificate at least within the past seven (7) calendar days. The Bank may, on demand, inspect the documentation supporting any Weekly International Borrowing Base Certificate. With respect to the Weekly International Borrowing Base Certificate delivered for the fourth week of every calendar month, such Weekly International Borrowing Base Certificate shall include (i) a list of the Eligible Export-Related Accounts Receivable and (ii) a list of Export Orders that are included in the International Borrowing Base, as at the end of the preceding month, such list to be in such form and containing such information and detail as the Bank may reasonably request, and as is satisfactory to the Bank, to comply with the requirements of the Ex-Im Bank Guaranty, including, without limiting the generality of the foregoing, as to Eligible Export-Related Accounts Receivable, aging thereof in the customary manner. The inclusion of any receivable in the Weekly International Borrowing Base Certificate shall constitute a representation and warranty by the Borrower that such receivable is an Eligible Export-Related Accounts Receivable.

(b) Two Business Days after the receipt of any such Weekly International Borrowing Base Certificate (unless the Bank prior to such date has notified the Borrower in writing of its determination of a different International Borrowing Base) the International Borrowing Base amount stated in any such Weekly International Borrowing Base Certificate shall automatically be deemed the International Borrowing Base amount until the earlier of (i) the date of the next determination or (ii) the date the Bank notifies the Borrower in writing of its determination of a different International Borrowing Base and the basis for such determination. The date of any such notification (if any) by the Bank is herein called a "Determination Date" and any increase or reduction in the International Borrowing Base by the Bank shall be effective as of such date. Each determination of the International Borrowing Base shall be made by the Bank in its reasonable business judgment, based on the most recent Weekly International Borrowing Base Certificate furnished by the Borrower and other information available to the Bank.

2.3 Letters of Credit.

- (a) Subject to and upon the terms and conditions herein set forth, including, without limitation, the applicable terms and conditions set forth in Article VII hereof, the Bank agrees that it will, following its receipt of a Letter of Credit Request, issue or cause to be issued for the account of the Borrower one or more irrevocable standby letters of credit which can be drawn down by a Buyer only if the Borrower, any Guarantor or an Eligible Joint Venture fails to perform any of its obligations under an Export Order (each a "Letter of Credit" and collectively, the "Letters of Credit"); provided that the Bank shall be under no obligation to issue or cause to be issued any Letter of Credit if at the time of such issuance:
 - (i) any order, judgment or decree of any Governmental Authority or arbitrator shall purport by its terms to enjoin or restrain the Bank or Wells Fargo from issuing such Letter of Credit or any requirement of law applicable to the Bank or Wells Fargo or any request or directive (whether or not having the force of law) from any Governmental Authority with jurisdiction over the Bank or Wells Fargo shall prohibit, or request that the Bank or Wells Fargo refrain from, the issuance of letters of credit generally; or
 - (ii) subject to any Cover requirements set forth in <u>Section 10.3</u> of this Agreement, (x) 25% of the Stated Amount of such Letter of Credit plus (y) 25% of the aggregate Stated Amount of all outstanding Letters of Credit plus (z) 100% of the aggregate Unpaid Drawings then outstanding (after giving effect to all Unpaid Drawings reimbursed prior to or concurrently with the issuance of such Letter of Credit) exceeds the International Borrowing Base then in effect; or
 - (iii) if the Stated Amount of such Letter of Credit plus all Letter of Credit Outstandings (after giving effect to all Unpaid Drawings reimbursed prior to or concurrently with the issuance of such Letter of Credit) exceeds the Commitment; or
 - (iv) unless the Bank and Ex-Im Bank shall give their prior written consent in their sole discretion, the expiry date of such Letter of Credit is later than 12 months from the date of issuance of such Letter of Credit; <u>provided however</u>, the Bank may issue or cause to be issued the Diyaar Project Letter of Credit so long as the Diyaar Project Letter of Credit shall have an expiry date of not later than the lesser of (a) 27 months from the date of issuance of such Diyaar Project Letter of Credit and (b) the number of months from Closing Date to the Final Disbursement Date.
- (b) Notwithstanding the foregoing, (i) the Bank may not issue or cause to be issued during the final sixty (60) days of the term of this Agreement any Letters of Credit which expire after the Maturity Date unless the Bank either has decided to renew this Agreement or has obtained the prior written approval of Ex-Im Bank and (ii) any Letter of Credit that shall have an expiration date after the Maturity Date shall be subject to Cover, such Cover to be delivered to the Bank thirty (30) days prior to the Maturity Date.

- (c) The Borrower acknowledges that the Bank will be liable to Wells Fargo for reimbursement of any and all draws under Letters of Credit issued by Wells Fargo and for all other amounts required to be paid under the applicable Application and the Standby Letter of Credit Agreement. Accordingly, the Borrower agrees to pay to the Bank any and all amounts required to be paid under the applicable Application and the Standby Letter of Credit Agreement, when and as required to be paid thereby.
- (d) From and after the Initial Date, all Existing Letters of Credit shall be deemed to have been issued pursuant hereto, and shall be subject to and governed by the terms and conditions hereof.

2.4 Letter of Credit Requests.

- (a) Whenever the Borrower desires that a Letter of Credit be issued for its account or that an existing expiry date shall be extended, the Borrower shall deliver to the Bank its prior written request therefore not later than 12:00 noon (Central time) on at least the second Business Day prior to the requested issuance or extension date, as the case may be. Each such request shall be in the form of Exhibit E attached hereto executed by the Borrower or such other form (including electronic requests) satisfactory to the Bank and Wells Fargo in their sole discretion (together with the Application, a "Letter of Credit Request") and, in the case of the issuance of any Letter of Credit, shall be accompanied by an Application therefor, completed to the satisfaction of the Bank and Wells Fargo, and such other certificates, documents and other papers and information as the Bank and Wells Fargo may reasonably request. Each Letter of Credit shall be denominated in Dollars or in an Approved Currency, shall expire no later than the date specified in Section 2.3, shall not be in an amount greater than is permitted under Section 2.3(a) and shall be in such form as may be approved from time to time by the Bank, Wells Fargo and the Borrower.
- (b) The making of each Letter of Credit Request shall be deemed to be a representation and warranty by the Borrower that such Letter of Credit may be issued in accordance with, and will not violate the requirements of Section 2.3(a) and Article VII of this Agreement. Upon its issuance of any Letter of Credit or the extension of the existing expiry date of any Letter of Credit, as the case may be, the Bank shall promptly notify the Borrower of such issuance or extension, which notice shall be accompanied by a copy of the Letter of Credit actually issued or a copy of any amendment extending the existing expiry date of any Letter of Credit, as the case may be.

2.5 Agreement to Repay Letter of Credit Drawings.

(a) Upon the receipt by the Bank of notice of any Drawing from a beneficiary under a Letter of Credit, the Bank promptly will provide the Borrower with telecopy notice thereof. The Borrower hereby agrees to reimburse the Bank by making payment to the Bank in immediately available funds at the Payment Office, for any payment made by the Bank under any Letter of Credit issued by it or Wells Fargo (each such amount so paid until reimbursed, an "<u>Unpaid Drawing</u>") on the date of such payment, <u>provided</u> that if the Borrower shall have received notice of any Drawing later than 11:00 a.m. (Central

time) on any Business Day, then the Borrower shall make such payment not later than 11:00 a.m. (Central time) on the immediately following Business Day, together with interest on the amount so paid by the Bank at the rate specified herein, to the extent not reimbursed prior to 2:00 p.m. (Central time) on the date of such payment, from and including the date paid but excluding the date reimbursement is made as provided above, such interest to be payable ON DEMAND.

- (b) The Borrower's obligations under this Section 2.5 to reimburse the Bank with respect to Unpaid Drawings (including, in each case, interest thereon) shall be absolute and unconditional under any and all circumstances (except as provided below with respect to the gross negligence, bad faith or willful misconduct of the Bank or Wells Fargo) and irrespective of any setoff, counterclaim or defense to payment which the Borrower may have or have had against the Bank or Wells Fargo, including any defense based upon the failure of any drawing under a Letter of Credit (each a "Drawing") to conform to the terms of the Letter of Credit (other than a defense based upon the gross negligence, bad faith or willful misconduct of the Bank or Wells Fargo in determining whether such Drawing conforms to the terms of the Letter of Credit) or any non-application or misapplication by the beneficiary of the proceeds of such Drawing, including any of the following circumstances:
 - (i) any lack of validity or enforceability of this Agreement or any of the other International Loan Documents;
 - (ii) the existence of any claim, setoff, defense or other right which the Borrower may have at any time against a beneficiary named in a Letter of Credit, any transferee of any Letter of Credit (or any Person for whom any such transferee may be acting), the Bank, Wells Fargo, or any other Person, whether in connection with this Agreement, any Letter of Credit, the transactions contemplated herein or any unrelated transactions (including any underlying transaction between the Borrower or any other Person and the beneficiary named in any such Letter of Credit);
 - (iii) any draft, certificate or any other document presented under the Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect;
 - (iv) the surrender or impairment of any security for the performance or observance of any of the terms of any of the International Loan Documents;
 - (v) the occurrence of any Default or Event of Default; or
 - (vi) any other circumstance which might otherwise constitute a defense available to, or a discharge of, the Borrower; provided that none of the foregoing is attributable to the gross negligence, bad faith or willful misconduct of the Bank or Wells Fargo.

- (c) The Borrower also agrees with the Bank that, in the absence of gross negligence, bad faith or willful misconduct of the Bank or Wells Fargo, the Bank shall not be responsible for, and the Borrower's reimbursement obligations under Section 2.5(a) shall not be affected by, among other things, the validity or genuineness of documents or of any endorsements thereon, even though such documents shall in fact prove to be invalid, fraudulent or forged, or any dispute between the Borrower or any other Person and the beneficiary of any Letter of Credit or any other party to which such Letter of Credit may be transferred, or any claims whatsoever of the Borrower or any other Person against any beneficiary of such Letter of Credit or any such transferee.
- (d) Neither the Bank nor Wells Fargo shall be liable for any error, omission, interruption or delay in transmission, dispatch or delivery of any message or advice, however transmitted, in connection with any Letter of Credit, except for errors or omissions caused by the Bank's or Wells Fargo's gross negligence, bad faith or willful misconduct. The Borrower agrees that any action taken or omitted by the Bank or Wells Fargo under or in connection with any Letter of Credit or the related drafts or documents, if done in the absence of gross negligence, bad faith or willful misconduct and in accordance with the standards of care specified in the Uniform Customs and Practice for Documentary Credits (1994 Revision), International Chamber of Commerce, Publication No. 500 (and any subsequent revisions thereof approved by a Congress of the International Chamber of Commerce and adhered to by the Bank) and, to the extent not inconsistent therewith, the Uniform Commercial Code of the State of New York, shall not result in any liability of the Bank or Wells Fargo to the Borrower or any other Person. IT IS THE INTENT OF THE PARTIES HERETO THAT NEITHER THE BANK NOR WELLS FARGO SHALL HAVE ANY LIABILITY UNDER THIS SECTION 2.5 FOR THE ORDINARY SOLE OR CONTRIBUTORY NEGLIGENCE OF THE BANK OR WELLS FARGO.
- 2.6 <u>Conflict Between Applications and Agreement</u>. To the extent that any provision of any Application related to any Letter of Credit or any provision of the Standby Letter of Credit Agreement is inconsistent with the provisions of this Agreement, the provisions of this Agreement shall control.

2.7 Increased Costs; Increased Capital.

(a) If the Bank determines that compliance with any law or regulation or any guideline or request from any central bank or other Governmental Authority (whether or not having the force of law) issued after the Closing Date affects or would affect the amount of capital required or expected to be maintained by the Bank or any corporation controlling the Bank and that the amount of such capital is increased by or based upon the existence of the Bank's Commitment hereunder and other commitments of this type, then, within fifteen (15) Business Days after written demand by the Bank, the Borrower shall pay to the Bank, from time to time as specified by the Bank, additional amounts sufficient to compensate the Bank or such corporation for such increase. A certificate as to such amounts, showing a calculation of such amounts in reasonable detail, submitted to the Borrower by the Bank shall be presumptive evidence of such amounts.

- (b) Anything in this Section notwithstanding: (i) the Borrower shall not be required to pay to the Bank reimbursement or indemnification with regard to any costs or expenses described in this Section, unless the Bank notifies the Borrower in writing of such costs or expenses within ninety (90) days after the date the Bank obtained knowledge of the circumstances that would cause the Borrower to pay such additional amounts; and (ii) the Bank shall not be permitted to pass through to the Borrower charges and costs under this Section on a discriminatory basis (i.e., which are not also passed through by the Bank to other customers of the Bank similarly situated where such customer is subject to documents providing for such pass through).
- 2.8 <u>Default Rate</u>. Notwithstanding anything set forth herein to the contrary if any Unpaid Drawing is not paid when due, all outstanding Obligations shall bear interest at a fluctuating rate per annum equal to the Default Rate, which interest shall be due and payable ON DEMAND.
- 2.9 Facility Subject to Ex-Im Bank Rules. The Borrower acknowledges that the Bank is willing to make its Commitment available to the Borrower because Ex-Im Bank is willing to guaranty payment of a significant portion of the Obligations pursuant to the Ex-Im Bank Guaranty. Accordingly, in the event of any inconsistency among the International Loan Documents and the Ex-Im Bank Guaranty, the Borrower Agreement, the Fast Track Agreement, the Fast Track Borrower Supplement and the rules and regulations of Ex-Im Bank governing the Working Capital Guaranty Program, the provision that is the more stringent on the Borrower shall control and compliance with any such applicable rule or regulation shall constitute an additional covenant of the Borrower incorporated herein by reference; provided, however, that the following provisions shall be deemed to be waived by the terms of this Agreement: (1) Fast Track Lender Agreement Section 2(e)(ii)(a) with respect to the requirement that the Borrower provide the Bank with a Lien on the domestic revolving loan collateral under the Domestic Credit Agreement; (2) Section 1.01 of the Borrower Agreement and the Ex-Im Bank Guaranty with respect to the definition of Eligible Export-Related Accounts Receivable regarding (i) Accounts Receivable owned by the Eligible Joint Ventures being included in the definition of Eligible Export-Related Accounts Receivable and (ii) affiliate receivables being included in the definition of Eligible Export-Related Accounts Receivable; (3) Section 4.12(b) of the Ex-Im Bank Guaranty with respect to the length of the term of certain letters of credit longer than 12 months; (4) Section 4.02 of the Ex-Im Bank Guaranty, as amended by the Fast Track Lender Agreement, with respect to the requirement that the Bank perfect its security interest in Export-Related Inventory; (5) Section 4.02(a) of the Ex-Im Bank Guaranty, as amended by the Fast Track Lender Agreement, with respect to the collateral securing the Domestic Credit Agreement that will not secure the Obligations; (6) Section 7 of the Fast Track Agreement, Section 2.07(c) of the Borrower Agreement, as amended by the Fast Track Borrower Supplement and Section 2.08(a) of the Borrower Agreement, as amended by the Fast Track Borrower Supplement with respect to the payment of all funds received in payment of the Excluded Export-Related Accounts Receivable into a lockbox and/or the Cash Collateral Account; (7) Section 2.19 of the Borrower Agreement with respect to naming the Bank as loss payee or mortgagee on its property insurance; and (8) Section 2.22 of the Borrower Agreement with respect to the consummation of the Aldabra Transactions. The Borrower and the Bank hereby acknowledge that Holdings and the Persons that own 20% or more of the ownership of Holdings are controlled by venture capital

firms, private equity firms or other investment groups and are therefore not required to guaranty this Agreement.

2.10 <u>Currency Equivalents</u>. For all purposes of this Agreement, (i) the equivalent in Dollars of any Approved Currency shall be determined by using the quoted spot rate at which Wells Fargo offers to exchange Dollars for such Approved Currency two Business Days prior to the date on which such equivalent is to be determined and (ii) the equivalent in any Approved Currency of Dollars shall be determined by using the quoted spot rate at which Wells Fargo offers to exchange such Approved Currency for Dollars two Business Days prior to the date on which such equivalent is to be determined. The equivalent in Dollars of each Letter of Credit made in an Approved Currency shall be recalculated hereunder on each date that it shall be necessary or desirable by any party to determine the amount of the Commitment, the Unpaid Drawings, the Letter of Credit Obligations, the Letter of Credit Outstandings, the Obligations or the face amount of any Letter of Credit on such date.

ARTICLE III. CASH COLLATERAL, PREPAYMENTS AND OTHER PAYMENTS.

3.1 Deposit of Cash Collateral and Required Prepayments.

- (a) The Borrower agrees that if at any time the Borrower or the Bank determines that (x) 25% of the aggregate Stated Amount of all outstanding Letters of Credit plus (z) 100% of the aggregate Unpaid Drawings then outstanding exceeds the International Borrowing Base, the Borrower will, within five Business Days after request from the Bank or actual knowledge thereof, (i) deliver to the Bank cash collateral in an amount equal to such excess, which shall be held by the Bank to the extent and until such excess is eliminated or (ii) furnish additional security to the Bank, in form and amount satisfactory to the Bank and Ex-Im Bank.
- (b) The Borrower agrees that if at any time the aggregate principal amount of the Letter of Credit Outstandings exceeds the amount of the Commitment, the Borrower shall immediately deliver to the Bank cash collateral in an amount equal to such excess, which, subject to Section 3.1(c) below, shall be held by the Bank to the extent and until such excess is eliminated.
- (c) Cash collateral provided to the Bank pursuant to Section 3.1(a) and (b) above shall be first applied by the Bank to the Unpaid Drawings, if any, and the remainder to be held by the Bank in a collateral account.

3.2 [RESERVED].

3.3 <u>Place of Payment or Prepayment</u>. All payments and prepayments of the Obligations made in accordance with the provisions of this Agreement or of the Note, including, without limitation, fees, reimbursements or interest, shall, subject to <u>Section 2.5(a)</u>, be made to the Bank at the Payment Office no later than 2:00 p.m. (Central time) on the date when due, in immediately available funds.

- 3.4 Prepayment Premium or Penalty. Each prepayment pursuant to Section 3.1 shall be without premium or penalty.
- 3.5 Taxes. Unless required by law, all payments (whether of principal, interest, reimbursements or otherwise) under this Agreement or on the Note shall be made by the Borrower without set-off or counterclaim and shall be made free and clear of and without deduction for any present or future tax, levy, impost or any other charge, if any, of any nature whatsoever now or hereafter imposed by any taxing authority, but excluding taxes imposed on or measured by the Bank's net income (including branch profits taxes) and franchise taxes imposed on the Bank (each non-excluded tax, levy, impost or charge, a "Tax"). If the making of such payments is prohibited by law unless such Tax is deducted or withheld therefrom, the Borrower shall pay to the Bank, on the date of each such payment, such additional amounts as may be necessary in order that the net amounts received by the Bank after such deduction or withholding shall equal the amounts which would have been received if such deduction or withholding were not required, provided that the Borrower shall be required to pay such additional amounts only if (A) such withholding Tax is imposed as a result of a change in law after the Closing Date (or, in the case of a transferee of the Bank, if such withholding Tax is imposed as a result of a change in law after the date the transferee became a party hereto) and (B) such withholding Tax is not imposed as a result of the failure of the Bank (or any transferee of the Bank) to provide such properly completed and duly executed documentation prescribed by applicable law as will permit such payments to be made without withholding or at a reduced rate of withholding, but only to the extent the Bank (or such transferee) is legally permitted to provide such documentation.
- 3.6 <u>Reduction or Termination of the Commitment</u>. The Borrower may at any time or from time to time reduce or terminate the Commitment by giving written notice thereof no later than 2:00 p.m. (Central time) not less than two (2) Business Days' prior to such reduction or termination. Any reduction in the Commitment shall be effective on the date specified in the Borrower's notice with respect to such reduction. The Commitment shall automatically terminate on the earlier of (i) the Maturity Date or (ii) in the event of acceleration of the Maturity Date of the Note. Each reduction of the Commitment hereunder shall be irrevocable.

ARTICLE IV. FEES.

- 4.1 <u>Facility Fee</u>. The Borrower shall pay to the Bank on the Closing Date, and on each Loan Facility Anniversary Date, a non-refundable facility fee in an amount equal to the product of (x) the Commitment and (y) the Annual Facility Fee Percentage; <u>provided</u> that on the second Loan Facility Anniversary Date, the facility fee shall be pro rated for a period of time equal to the greater of (i) the remaining term of the Agreement and (ii) six months. The facility fee payable on the Closing Date shall be in the amount of \$200,000.
- 4.2 <u>Letter of Credit Fees</u>. The Borrower agrees to pay the Bank a fee in respect of each Letter of Credit issued for the account of the Borrower (the "<u>Letter of Credit Fee</u>"), in an amount equal to the greater of (i) \$750 or (ii) an amount computed at the applicable rate for standby letters of credit in the Domestic Credit Agreement set forth in <u>Section 3.3(a)</u> thereof <u>minus</u> 0.25%. Letter of Credit Fees shall be due and payable quarterly in arrears on or before the fifth (5th) Business Day following the end of each Fiscal Quarter. Fees due under this

<u>Section 4.2</u> shall be computed on the basis of a year of 360 days. In addition, the Borrower shall pay to the Bank or Wells Fargo, as applicable, solely for its own account as issuer of Letters of Credit, any applicable amendment, transfer, negotiation and other fees as determined in accordance with the Bank's or Wells Fargo's then current fee policy.

4.3 Fees Not Interest; Nonpayment. The fees described in this Agreement represent compensation for services rendered and to be rendered separate and apart from the lending of money or the provision of credit and do not constitute compensation for the use, detention or forbearance of money, and the obligation of the Borrower to pay each fee described herein shall be in addition to, and not in lieu of, the obligation of the Borrower to pay interest, other fees described in this Agreement, and expenses otherwise described in this Agreement. Fees shall be payable when due in Dollars and in immediately available funds. All fees shall be non-refundable, and shall, to the fullest extent permitted by law, bear interest, if not paid within five (5) Business Days of when due, at a rate per annum equal to the Default Rate.

ARTICLE V. PERMITTED PURPOSES OF LETTERS OF CREDIT

The Borrower agrees that the Letters of Credit shall be issued solely to support one or more of the Borrower's, any Guarantor's or any Eligible Joint Venture's obligations under an Export Order. No Letter of Credit may be a Warranty Letter of Credit; provided, however, a performance Letter of Credit that converts to a Warranty Letter of Credit at the end of the performance period may be issued so long as the Bank's liability thereunder and therefor is eliminated (in a manner satisfactory to the Bank in its sole discretion) not later than the date when the warranty period commences.

ARTICLE VI. REPRESENTATIONS AND WARRANTIES.

The Borrower and, as applicable, each Guarantor each represents and warrants that:

- 6.1 <u>Organization and Qualification</u>. It and each of its Subsidiaries (a) is duly organized, validly existing, and in good standing under the laws of the jurisdiction of its organization (except to the extent that such failure to be in good standing under such laws could not reasonably be expected to have a Material Adverse Effect); (b) has the power to own its Properties and to carry on its business as now conducted; and (c) is duly qualified to do business and is in good standing in all jurisdictions in which the failure to so qualify or be in good standing could reasonably be expected to have a Material Adverse Effect. As of the Closing Date, the Borrower has no Subsidiaries other than those listed on <u>Schedule 6.1</u>.
- 6.2 <u>Financial Statements</u>; <u>Positive Tangible Net Worth</u>. The Borrower has furnished the Bank with the following financial statements: (a) its year-end Annual Audited Financial Statements for the fiscal year ended December 31, 2005; and (b) its unaudited consolidated balance sheet, statement of earnings and statement of cash flow as at and for the three month period ended June 30, 2006. These statements have been prepared in conformity with GAAP, except, in the case of unaudited financial statements, for the absence of footnote disclosure and for year-end audit adjustments. Such statements fairly present, in all material respects, the

consolidated financial condition of the Borrower and its Subsidiaries and the consolidated results of its operations as at the dates and for the periods indicated. Since December 31, 2005, there has been no change in any circumstances, facts or conditions nor shall an event have taken place which, individually or in the aggregate, could reasonably be expected to have a Material Adverse Effect. During the period from June 30, 2006 to the Closing Date, the Borrower has had a positive tangible net worth, as determined in accordance with GAAP. For the purpose of this determination, net worth (as determined in accordance with GAAP) must be (i) increased by any Debt of the Borrower and its Subsidiaries subordinated to the indebtedness evidenced by this Agreement and (ii) decreased by all intangible assets (including, without limitation, all patents, licenses, goodwill, subscription lists, capitalized software, organization expenses, covenants not to compete, and investments in and monies due from Affiliates, officers and directors of the Borrower and its Subsidiaries).

- 6.3 <u>Litigation; Contingent Liabilities</u>. Other than as set forth on <u>Schedule 6.3</u>, there is no material action or proceeding pending or, to the knowledge of the Loan Parties, overtly threatened against or involving the Borrower or any Subsidiary before any court, administrative agency or arbitrator the results of which could reasonably be expected to have a Material Adverse Effect. Other than any liability incident to the actions or proceedings disclosed on <u>Schedule 6.3</u>, or provided for or disclosed in the financial statements referred to in <u>Section 6.2</u>, as of the Closing Date, none of the Borrower or any of the Borrower's Subsidiaries has any contingent liabilities which could reasonably be expected to have a Material Adverse Effect.
- 6.4 <u>Default</u>. Neither the Borrower nor any Subsidiary is (i) in default under any agreement or instrument to which any of such Persons is a party or by which any of their respective properties or assets is bound or affected, which default could reasonably be expected to have a Material Adverse Effect, (ii) in default under or in violation of the provisions of any Governmental Requirement (except where such default or violation could not reasonably be expected to have a Material Adverse Effect) or (iii) in default in any material respect with any material provisions of any Export Order (which is included in the International Borrowing Base), which has not been waived by the applicable party thereto. No Event of Default or Default has occurred and is continuing.

6.5 Title to Assets; Ownership.

- (a) The Borrower and each Subsidiary has good and marketable title to, or a valid leasehold interest in, its material Properties, subject to no Liens except as permitted by Section 9.1 hereof.
- (b) As of the Closing Date, except as permitted in Section 9.1 hereof, the Persons identified on Schedule 6.5 directly own, free and clear of all Liens or restrictions on transferability or voting, one hundred percent (100%) of the outstanding shares of Capital Stock of the Borrower and all such shares are validly issued, fully paid and non assessable. As of the Closing Date, there are no outstanding warrants, options, contracts or commitments of any kind entitling any Person to purchase or otherwise acquire (i) any shares of the Capital Stock of the Borrower or (ii) any securities convertible into or exchangeable for any shares of such Capital Stock. As of the Closing Date, no securities are outstanding which are convertible into or exchangeable for any shares of Capital

Stock of the Borrower. As of the Closing Date, the Persons identified on <u>Schedule 6.5</u> as Controlling Affiliates are the only Controlling Affiliates of the Borrower (other than Controlling Affiliates that are not required to execute a Guaranty pursuant to <u>Section 6(c)</u> of the Fast Track Lender Agreement).

6.6 <u>Authorization, Validity, Etc.</u> It has the power and authority to make, execute, deliver and carry out the International Loan Documents to which it is a party and the transactions contemplated therein and to perform its obligations thereunder and all such action has been duly authorized by all necessary proceedings on its part. The International Loan Documents to which any Loan Party is a party have been duly and validly executed and delivered by such Loan Party and constitute valid and legally binding obligations of such Loan Party enforceable against such Loan Party in accordance with their respective terms, except as limited by Debtor Laws. The Liens created by the International Security Documents will constitute valid and perfected Liens on the Collateral described therein superior in right to all other Liens, except for Permitted Liens.

6.7 Intentionally Omitted.

6.8 <u>Taxes</u>. The Borrower and each Subsidiary has timely filed, or has caused to be timely filed, all tax returns and reports required by applicable law, has timely paid, or has caused to be timely paid, all applicable taxes, assessments, deposits and contributions owing by Borrower and each such Subsidiary and will timely pay, or cause to be timely paid, all such items in the future as they became due and payable. The Borrower or any Subsidiary may, however, defer payment of any contested taxes; <u>provided</u>, that Borrower or such Subsidiary (a) in good faith contests Borrower's obligation to pay such taxes by appropriate proceedings promptly and diligently instituted and conducted; (b) notifies the Bank in writing of the commencement of, and any material development in, the proceedings; (c) posts bonds or takes any other steps required to keep the contested taxes from becoming a Lien upon any of the Collateral; and (d) maintains adequate reserves therefore in conformity with GAAP. The Borrower is not aware of any proposed tax assessment by any taxing authority or of any claims by any Governmental Authority for any unpaid taxes which could reasonably expected to have a Material Adverse Effect, in each case except liabilities contested in good faith or for which adequate reserves have been provided.

6.9 <u>Conflicting or Adverse Agreements or Restrictions</u>. Neither the execution, delivery and performance by each Loan Party of the International Loan Documents to which it is a party, any Export Order to which it is a party and which is included in the International Borrowing Base, nor the consummation of the transactions contemplated thereby nor fulfillment of and compliance with the respective terms, conditions and provisions thereof, will conflict with or result in a breach of any of the terms, conditions or provisions of, or constitute a default under, or result in any violation of, or result in the creation or imposition of any Lien on any of the Property of the Borrower or any Subsidiary (other than Liens created under the International Loan Documents and, in connection with the execution, delivery and performance of any Export Order, Permitted Liens) pursuant to, (a) the Organization Documents of such Person; (b) any Governmental Requirement; or (c) the terms, conditions or provisions of any material contract to which such Person is a party or by which it is bound or to which it is subject.

6.10 <u>Information</u>. Neither this Agreement nor any other document, certificate or written statement furnished to the Bank by or on behalf of the Borrower or any Subsidiary in

connection with this Agreement (including, without limitation, any Export Order included in the International Borrowing Base or any Weekly International Borrowing Base Certificate) contains any statement of material fact which is untrue or incorrect in any material respect or omits to state a material fact necessary in order to make the statements contained herein and therein not misleading in any material respect in light of the circumstances under which such information was provided as of the time when delivered. As of the Closing Date, there is no fact currently known to any Loan Party which now or in the future could reasonably be expected to have a Material Adverse Effect and which has not been set forth or referred to in this Agreement, the other International Loan Documents, or such documents, certificates or statements heretofore furnished to the Bank.

- 6.11 No Consent. Except to the extent obtained, no authorization or approval or other action by, and no notice to or filing with, any Person or any Governmental Authority is required for the due execution, delivery, and performance (a) by any Loan Party of any International Loan Document to which it is a party or the borrowings hereunder, in each case as contemplated herein, or the effectuation of the transactions contemplated under any International Loan Document to which it is a party, except (i) filings and recordings to perfect the Liens created under the International Loan Documents and (ii) those which, if not obtained, could not reasonably be expected to have a Material Adverse Effect, or (b) of any Export Order by the parties thereto or the effectuation of the transactions contemplated thereunder, except those which, if not obtained, could not reasonably be expected to have a Material Adverse Effect.
- 6.12 Environmental Matters. The Borrower and each Subsidiary (a) owns, possesses or otherwise holds, and is in compliance with the terms and conditions of, all Governmental Approvals under Environmental Laws necessary for the ownership or lease and operation of its Property and the carrying on of its business as now conducted or proposed to be conducted, except where the failure could not reasonably be expected to have a Material Adverse Effect, and (b) is in compliance with all Governmental Requirements relating to the environment except where the failure could not reasonably be expected to have a Material Adverse Effect. There are no civil, criminal or administrative actions, investigations or proceedings or claims pending or, to the best knowledge of the Borrower, threatened against the Borrower or any Subsidiary, for noncompliance with Environmental Laws or arising out of the presence or release of hazardous materials at, on or under any Property now owned, leased or operated by the Borrower or any Subsidiary which could reasonably be expected to have a Material Adverse Effect.
- 6.13 <u>Debt</u>. As of the Closing Date, the financial statements referred to in Section 6.2 contain a complete and accurate disclosure in all material respects of all Debt of the Borrower and its Subsidiaries outstanding as of the respective dates of such financial statements, except for Guaranties permitted by Section 6.2(f) (or any comparable section) of the Domestic Credit Agreement. As of the Closing Date, none of the Borrower or any of its Subsidiaries has incurred any other Debt since the respective dates of such financial statements, except as would have been permitted under <u>Section 9.6</u> of this Agreement had this Agreement been in effect at the time of the incurrence of any such additional Debt.
- 6.14 <u>Compliance with Laws.</u> The Borrower and each Subsidiary has complied in all material respects with all provisions of all applicable laws and regulations, including those relating to Borrower's ownership of real or personal property, the conduct and licensing of

Borrower's business, the payment and withholding of taxes, ERISA and other employee matters, safety and environmental matters.

6.15 <u>Suspension and Debarment, Etc.</u> Neither the Borrower nor any of its Principals (as defined below) are (A) debarred, suspended, proposed for debarment with a final determination still pending, declared ineligible or voluntarily excluded (as such terms are defined under any of the Debarment Regulations referred to below) from participating in procurement or nonprocurement transactions with any U.S. federal government department or agency pursuant to any of the Debarment Regulations (as defined below) or (B) indicted, convicted or had a civil judgment rendered against the Borrower or any of its Principals for any of the offenses listed in any of the Debarment Regulations. For the purposes hereof, (1) "<u>Principals</u>" shall mean any officer, director, owner, partner, key employee, or other person with primary management or supervisory responsibilities with respect to the Borrower; or any other Person (whether or not an employee) who has critical influence on or substantive control over the transaction covered by this Agreement and (2) the "<u>Debarment Regulations</u>" shall mean (x) the Government wide Debarment and Suspension (Nonprocurement) regulations (Common Rule), 53 Fed. Reg. 19204 (May 26, 1988), (y) Subpart 9.4 (Debarment, Suspension and Ineligibility) of the Federal Acquisition Regulations, 48 C.F.R. 9.400-9.409 and (z) the revised Government wide Debarment and Suspension (Nonprocurement) regulations (Common Rule), 60 Fed. Reg. 33037 (June 26, 1995).

ARTICLE VII. CONDITIONS.

The obligation of the Bank to issue each Letter of Credit is subject to the following conditions:

7.1 Representations True and No Defaults.

- (a) The representations and warranties of the Borrower and the Guarantors contained in the International Loan Documents shall be true and correct in all material respects on and as of the date of the issuance of any Letter of Credit, as though made on and as of such date except for any representation or warranty which is specified as being made as of an earlier date, in which case such representation or warranty shall only speak as to such earlier date; and
 - (b) no Event of Default or Default shall have occurred and be continuing.
- 7.2 Terms of Sale. Letters of Credit shall be issued only against Export Orders, copies of which have been delivered to the Bank, if the Bank or Ex-Im Bank requires, and which are maintained on file with the Borrower. Each Letter of Credit hereunder is subject to the condition precedent that the Bank shall have reviewed and approved in its sole discretion the terms and conditions, including without limitation that there exists satisfactory evidence that the Items will in fact be exported, of the applicable Export Order.
- 7.3 <u>Governmental Approvals</u>. The Borrower and each Guarantor and the Bank shall have obtained all Governmental Approvals required for the making and carrying out of this Agreement, the issuance of Letters of Credit pursuant hereto and the issuance of the Note and the

execution, delivery and performance of the applicable Export Order that is included in the International Borrowing Base. Without limitation of the foregoing, no Letter of Credit shall be issued hereunder (a) in violation of the Ex-Im Bank Guaranty or contrary to the instructions of Ex-Im Bank or (b) following the occurrence of an Event of Default that is continuing or of any event which but for the giving of notice or the lapse of time or both would constitute an Event of Default pursuant to Section 4.10 of the Ex-Im Bank Guaranty.

- 7.4 <u>Letter of Credit Documents</u>. On the date of the issuance of any Letter of Credit, the Bank shall have received from the Borrower a Letter of Credit Request in respect of Letters of Credit to be issued on such issuance date together with such other documents and certificates relating to the transactions herein contemplated as the Bank may reasonably request.
- 7.5 <u>Required Initial Documents and Certificates</u>. On or before the issuance of the first Letter of Credit hereunder (the "<u>Initial Date</u>"), the Bank shall have received from the Borrower or, as applicable, each Guarantor the following, in each case in form, scope and substance satisfactory to the Bank:
 - (i) the SBA/Ex-Im Bank Joint Application, duly executed by the parties thereto, together with the \$100 application fee related thereto;
 - (ii) the Borrower Agreement duly executed by the parties thereto;
 - (iii) the Fast Track Borrower Supplement duly executed by the parties thereto;
 - (iv) this Agreement duly executed by the parties hereto, together with all schedules and exhibits thereto;
 - (v) the Note duly executed by the Borrower;
 - (vi) the International Security Agreement duly executed by the parties thereto;
 - (vii) the International Pledge Agreement duly executed by the parties thereto;
 - (viii) the Standby Letter of Credit Agreement duly executed by the parties thereto;
 - (ix) an officer's certificate of the Borrower signed in the name of the Borrower by its President, Chief Financial Officer, Vice President or Secretary, and dated as of the Closing Date to which are attached true and correct copies of the Organization Documents of the Borrower and corporate resolutions duly adopted by the Board of Directors of the Borrower authorizing the transactions contemplated by the International Loan Documents;

- (x) such certificates as may be appropriate to demonstrate the continued existence and good standing of the Borrower in its jurisdiction of organization;
- (xi) an officer's certificate of each Guarantor signed in the name of such Guarantor by its President, Chief Financial Officer, Vice President or Secretary, and dated as of the Closing Date to which are attached true and correct copies of the Organization Documents of such Guarantor and resolutions duly adopted by the Board of Directors or Managers of such Guarantor authorizing the transactions contemplated by the International Loan Documents;
- (xii) such certificates as may be appropriate to demonstrate the continued existence and good standing of each Guarantor in its jurisdiction of organization;
 - (xiii) a Weekly International Borrowing Base Certificate dated within five (5) calendar days preceding the Initial Date;
- (xiv) copies of UCC Financing Statements and all other requisite filing documents necessary to perfect the Liens granted pursuant to the International Security Documents and, if applicable, duly executed releases or assignments of Liens and UCC-3 financing statements in recordable form, each in form and substance satisfactory to the Bank, and filed with the appropriate filing offices, covering all of the Collateral subject thereto, as may be necessary to reflect that the Liens granted to the Bank are of the priority required by Section 9.1 hereof;
 - (xv) the facility fee required under Section 4.1;
- (xvi) the Borrower's monthly Export-Related Accounts Receivable aging schedule showing all of the Borrower's Export-Related Accounts Receivable as of the last day of the calendar month immediately preceding the Closing Date;
 - (xvii) the Borrower's financial statements described in Section 6.2;
 - (xviii) an Exceptions Approval Letter properly signed by Ex-Im Bank;
 - (xix) the Guaranty, duly executed and delivered by each Guarantor;
- (xx) a control agreement in the form of Exhibit J hereto for the Cash Collateral Account at Wells Fargo, duly executed by Bank, Wells Fargo and the Borrower;
 - (xxi) a copy of the Standing Instruction Agreement executed by Ahli United Bank;

- (xxii) a customary opinion dated as of the Closing Date of legal counsel to the Borrower and Guarantor as to such matters as the Bank may reasonably request;
- (xxiii) any amendments and/or consents required to the Domestic Loan Documents and the Travelers Agreement (as defined in the Domestic Credit Agreement) to allow the Liens granted pursuant to the International Security Documents;
- (xxiv) a confirmation of the pledge of each Guarantor's equity interests in the Durrat Joint Venture, executed by each other member of the Durrat Joint Venture; and
 - (xxv) any further documentation or evidence that the Bank or Ex-Im Bank may require.

In addition, as of the Initial Date, all legal matters incident to the transactions herein contemplated shall be reasonably satisfactory to the Bank.

- 7.6 Ex-Im Bank Acknowledgment, Etc. On or before the Initial Date, Ex-Im Bank shall have received from the Bank copies of all documents required pursuant to the Ex-Im Bank Guaranty and the Fast Track Agreement. The Bank shall have received from Ex-Im Bank a Loan Authorization Agreement executed by Ex-Im Bank, together with all other items referred to in Paragraph 3 of the Fast Track Agreement.
- 7.7 <u>Post-Closing Lien Search</u>. After the Closing Date, the Bank shall have received the results of all post-closing lien searches confirming that the Bank has obtained a perfected security interest in the Collateral of the priority required by <u>Section 9.1</u>.

ARTICLE VIII. AFFIRMATIVE COVENANTS.

The Borrower covenants and agrees that, so long as the Borrower requests issuance of Letters of Credit hereunder and until payment in full of all Obligations, the expiry of all Letters of Credit and the termination of this Agreement, the Borrower shall, and shall (except in the case of the covenants set forth in Sections 8.1 and 8.5) cause each Subsidiary to:

- 8.1 <u>Financial Statements and Information</u>. Deliver to the Bank:
- (a) within 90 days after the end of each Fiscal Year of the Borrower, the Annual Audited Financial Statements of the Borrower and its Subsidiaries for such Fiscal Year;
- (b) within 75 days after the first calendar day of each Fiscal Year, a budget for such Fiscal Year, which budget shall be prepared on a Fiscal Quarter basis and shall contain a projected, consolidated balance sheet, consolidated statement of earnings (broken out in reasonable detail by business segment) and a consolidated statement of cash flow of the Borrower and its Subsidiaries for such Fiscal Year. It is understood that

- (i) any projections or budget furnished to the Bank are subject to significant uncertainties and contingencies, which are beyond the control of the Borrower and its Subsidiaries, (ii) no assurance is given by the Borrower and its Subsidiaries that such projections will be realized, and (iii) the actual results may differ from such projections and such differences may be material;
- (c) within 45 days after the first three Fiscal Quarters of each Fiscal Year, the Quarterly Unaudited Financial Statements of the Borrower and its Subsidiaries;
 - (d) a Weekly International Borrowing Base Certificate as required by Section 2.2(a);
 - (e) the notices and information required to be delivered by Section 6.1(f) (or any comparable section) of the Domestic Credit Agreement;
- (f) with reasonable promptness after the sending or filing thereof, copies of all reports and registrations statements which the Borrower or any Subsidiary files with the Securities and Exchange Commission or any national securities exchange;
- (g) in connection with the Durrat Project and the Diyaar Project, within 45 days after the end of each calendar month, (i) copies of all standing instructions delivered to Ahli United Bank (to the extent not previously delivered), (ii) monthly statements of the Ahli United Bank Deposit Account, and (iii) copies of any and all invoices issued by (x) the Durrat Joint Venture to Durrat, (y) the Diyaar Joint Venture to Diyaar and (z) Great Lakes LLC to (I) the Durrat Joint Venture and (II) the Diyaar Joint Venture; and
 - (h) such additional financial or other information as the Bank may reasonably request.

Together with each delivery of the Annual Audited Financial Statements and the Quarterly Unaudited Financial Statements, the Borrower will deliver to the Bank a Compliance Certificate substantially in the form of Exhibit D hereto, showing (in reasonable detail and appropriate calculations and computations in form reasonably satisfactory to the Bank) compliance with the financial covenants set forth in Section 9.5, stating that there exists no Event of Default or Default or Default, or, if any such Event of Default or Default exists, stating the nature thereof, the period of existence thereof and what action the Borrower has taken or proposes to take with respect thereto. The Bank is authorized to deliver a copy of any information and financial statement delivered to it pursuant to this Agreement to Ex-Im Bank and any Governmental Authority having jurisdiction over the Bank.

- 8.2 <u>Books and Records</u>. Maintain, and cause its Subsidiaries to maintain, proper books of record and account in accordance with GAAP (including records relating to Collateral) which accurately reflect all of its business affairs and transactions in relation to its business and activities in all material respects.
- 8.3 <u>Insurance</u>. Maintain, and cause its Subsidiaries to maintain, insurance with financially sound, responsible and reputable companies in such types (including, without limitation, fire and casualty) and amounts and against such casualties, risks and contingencies as

is customarily carried by owners of similar businesses and Properties, and furnish to the Bank, together with each delivery of Annual Audited Financial Statements under Section 8.1(a), certificates of insurance evidencing the insurance carried. Without limiting the generality of the foregoing, the Borrower will comply with the insurance requirements set forth in each International Security Document to which the Borrower is a party. For purposes hereof, the Borrower's types and amounts of insurance set forth in Schedule 8.3 shall be acceptable to the Bank.

8.4 Inspection of Property and Records; Audits of Collateral; Etc.

- (a) Permit, and cause its Subsidiaries to permit, any Person designated by the Bank in writing to visit and inspect any of the Properties, books and financial records of the Borrower and its Subsidiaries and discuss its affairs and finances with its officers, all at such times during normal business hours as the Bank may reasonably request at a mutually agreeable time, and cause its officers and employees to give full cooperation and assistance in connection therewith. Without limiting the generality of the foregoing, the Borrower will permit, and cause its Subsidiaries to permit, any officer or employee of, or agent designated by, the Bank to have access to, examine and inspect the Collateral and to audit and make copies from the Borrower's and any Subsidiaries' records, files and books of account in order to verify such Collateral.
- (b) The Borrower will, from time to time, on a quarterly basis (other than the fourth Fiscal Quarter), permit the Bank, an independent certified public accountant or another appropriate entity acceptable to the Bank to audit the Borrower's books, records and procedures with respect to Inventory, Accounts Receivable and other Collateral. This inspection shall include a physical inspection of the Collateral in accordance with normal commercial lending practices, except that a book audit shall be made of the Borrower' Accounts Receivable and any intangible Collateral. Promptly after the conclusion of each such audit, the Borrower shall pay to the Bank its reasonable and customary out-of-pocket expenses associated with such audit.
- 8.5 <u>Notice of Certain Matters</u>. Promptly, but in any event within five (5) Business Days, notify the Bank in writing of the occurrence of any of the following:
 - (a) the Borrower or any Guarantor (i) applies for, consents to or suffers the appointment of, or the taking of possession by, a receiver, custodian, trustee, liquidator or similar fiduciary of itself or of all or a substantial part of its property or calls a meeting of its creditors, (ii) admits in writing its inability, or is generally unable, to pay its debts as they become due or ceases operations of its present business, (iii) makes a general assignment for the benefit of creditors, (iv) commences a voluntary case under any state or federal bankruptcy laws (as now or hereafter in effect), (v) is adjudicated as bankrupt or insolvent, (vi) files a petition seeking to take advantage of any other law providing for the relief of debtors, (vii) acquiesces to, or fails to have dismissed within thirty (30) days, any petition filed against it in any involuntary case under such bankruptcy laws, or (vii) takes any action for the purpose of effecting any of the foregoing;

- (b) any Lien in any of the Collateral, granted or intended by the International Loan Documents to be granted to Bank, ceases to be a valid, enforceable, perfected, first priority Lien (or a lesser priority if expressly permitted pursuant to Section 6 of the Loan Authorization Agreement) subject only to Permitted Liens;
- (c) the issuance of any levy, assessment, attachment, seizure or Lien, other than a Permitted Lien, against any of the Collateral which is not stayed or lifted within thirty (30) calendar days;
 - (d) any proceeding is commenced by or against the Borrower or any Guarantor for the liquidation of its assets or dissolution;
- (e) any litigation is filed against the Borrower or any Guarantor which has had or could reasonably be expected to have an Ex-Im Material Adverse Effect and such litigation is not withdrawn or dismissed within thirty (30) calendar days of the filing thereof;
 - (f) any Default or Event of Default under the International Loan Documents;
 - (g) any failure to comply with any terms of the Loan Authorization Agreement;
- (h) any material provision of this Agreement or any other International Loan Document for any reason ceases to be valid, binding and enforceable in accordance with its terms;
 - (i) any event which has had or could reasonably be expected to have an Ex-Im Material Adverse Effect;
 - (j) the aggregate outstanding amount of Disbursements exceeds the applicable Export-Related Borrowing Base;
 - (k) the existence of any condition requiring a mandatory prepayment or deposit of cash collateral pursuant to Section 3.1 hereof;
- (l) any failure to pay when due any amount payable to the Bank by the Borrower or any Subsidiary under any non-Ex-Im Bank guaranteed loan(s) extended by the Bank to the Borrower or any Subsidiary;
- (m) any actual breach, or threatened breach made in writing, in any material respect of any material provision of any Export Order included in the International Borrowing Base, by any party to any such contract; or
- (n) the Borrower has included in the International Borrowing Base any Accounts Receivable that no longer qualifies as Eligible Export-Related Accounts Receivable.

8.6 <u>Security and Further Assurances</u>. Each Loan Party shall, from time to time as the Bank may reasonably request, at the Borrower's own expense, promptly execute and deliver all such further instruments (including, without limiting the generality of the foregoing, additional security agreements, and financing statements) and do such other acts as the Bank may reasonably request for the purpose of protecting or perfecting any Lien created or granted or intended to be created or granted in the International Security Documents or in order to ensure that any such Lien is of the priority purported to be granted thereby and as required by <u>Section 9.1</u> hereof, or in order to police or protect any Collateral or otherwise to carry out more effectually the purposes and intent of the International Loan Documents.

8.7 Establishment of Cash Collateral Account. Establish and maintain a deposit account with Wells Fargo, in which payments related to Export-Related Accounts Receivable will be deposited directly by wire transfer (the "Cash Collateral Account"); provided, however, that payments from Durrat and Diyaar will be deposited in an account at Ahli United Bank (the "Ahli United Bank Deposit Account") with the portion of such payments belonging to the Borrower or Guarantor to be deposited in the Cash Collateral Account reasonably promptly, and in any event within two (2) Business Days (as defined in the Standing Instruction Agreement), from the date of the deposit in the Ahli United Bank Deposit Account and, provided, further that (i) payments related to the Export Orders identified on Schedule 8.7 and (ii) payments related to Export Orders that have a contractual value of less than \$1,000,000 and which are not included in the International Borrowing Base (collectively, the "Excluded Export-Related Accounts Receivable") shall not be required to be deposited into the Cash Collateral Account. The Cash Collateral Account, including the withdrawal and release of funds therefrom, shall be governed by the control agreement in the form of Exhibit J hereto. So long as no Event of Default has occurred and is continuing, the Bank will deliver orders to Wells Fargo providing for a daily transfer of all funds from the Cash Collateral Account to the Borrower's operating account at Bank of America, N.A. or such other account as the Borrower may direct in writing from time to time.

8.8 <u>Maintenance of Property</u>. Maintain in good repair, working order and condition, excepting ordinary wear and tear and damage due to casualty, all of its material properties which are used in the conduct of its business (whether owned or leased by such party) and make or cause to be made all appropriate repairs, renewals and replacements thereof, in each case consistent with sound business practices; provided that such obligation shall not apply to property that the Borrower or such Subsidiary reasonably determines in good faith that maintenance thereof is no longer economically desirable.

8.9 Corporate Existence; Payment of Taxes and Claims; Compliance With Laws.

- (a) Except as otherwise permitted by the Domestic Credit Agreement, maintain, and cause each Subsidiary to maintain, its existence;
- (b) pay, and cause to be paid, all taxes, assessments, governmental charges, claims for labor, and other obligations that have become due and payable and that by law have or may give rise to a Lien on such Person's Property other than Liens permitted by <u>Section 9.1</u>, <u>provided</u> that none of the Borrower or any Subsidiary shall be required to pay such tax, assessment, governmental charge, claim for labor or other obligations

- which (i) is being contested in good faith by appropriate proceedings which will prevent the forfeiture or sale of any Property or any material interference with the use thereof by such Person, and has been adequately reserved against in accordance with GAAP, and (ii) if not so paid, could not reasonably be expected to have a Material Adverse Effect; and
- (c) comply, and cause each Subsidiary to comply, with all Governmental Requirements except where failure to so comply could not reasonably be expected to have a Material Adverse Effect.
- 8.10 <u>Assignment of International Letter of Credit Proceeds</u>. Each letter of credit or bank guaranty issued in favor of the Borrower with respect to the sale of Items related to Export Orders to be financed under this Agreement and included in the International Borrowing Base, if any, (a) shall be in form and substance reasonably satisfactory to the Bank, and (b) shall be issued in the manner required by the Bank by financial institutions reasonably acceptable to the Bank. If required by the Bank, any such letter of credit or bank guaranty shall provide for an assignment of proceeds thereof, in form and substance reasonably satisfactory to the Bank.

8.11 Environmental Matters. At all times:

- (a) comply, and cause each Subsidiary to comply, with all applicable Governmental Requirements and Governmental Approvals under Environmental Laws except where failure to so comply could not reasonably be expected to have a Material Adverse Effect; and
- (b) promptly notify the Bank after becoming aware of any enforcement, cleanup, removal or other governmental or regulatory actions instituted or threatened in writing against Borrower or any Subsidiary pursuant to any applicable Environmental Laws which could reasonably be expected to result in liabilities of Borrower or its Subsidiaries in excess of \$3,000,000 (to the extent not covered by insurance or third party indemnification).
- 8.12 <u>Controlling Affiliates</u>. If any Person shall become a Controlling Affiliate after the Closing Date, then the Borrower (a) will cause such Controlling Affiliate if required under <u>Section 6(c)</u> of the Fast Track Lender Agreement, to execute and deliver to the Bank the Guaranty within thirty (30) days of becoming a Controlling Affiliate and (ii) to execute and deliver to the Bank the Security Agreement within thirty (30) days of becoming a Controlling Affiliate, if such Controlling Affiliate's assets will be included in the International Borrowing Base.
- 8.13 <u>Assembly of Export Order Summaries</u>. In the event the Borrower has furnished summaries of Export Orders to the Bank pursuant to <u>Section 2.2(a)</u>, at least once each calendar quarter, make available to the Bank a sampling of such Export Orders selected by the Bank representing at least 10% of the aggregate U.S. dollar volume of all Export Orders and 10% of the number of Export Orders supporting Letters of Credit issued during the preceding calendar quarter available for the Bank's review and inspection.

- 8.14 <u>Delivery of Domestic Loan Documents</u>. Deliver or cause to be delivered to the Bank promptly after the effectiveness thereof, any amendments, modifications, renewals and extensions of the Domestic Credit Agreement.
- 8.15 <u>Reaffirmation of Borrower</u>. Contemporaneously with the consummation of the Aldabra Transactions, the Successor Borrower shall execute and deliver to the Bank a Reaffirmation, Ratification and Assumption Agreement, substantially in the form attached hereto as <u>Exhibit I</u> and such other documents as may be reasonably requested by the Bank or Ex-Im Bank to assure themselves of the continuing effectiveness of the International Loan Documents being reaffirmed, ratified and assumed by the Successor Borrower.

ARTICLE IX. NEGATIVE COVENANTS.

So long as the Borrower may request issuance of Letters of Credit hereunder and until payment in full of all Obligations and termination of this Agreement, unless the Bank shall otherwise give its prior written consent in its sole and absolute discretion:

- 9.1 <u>Liens</u>. The Borrower will not, and will not permit any Subsidiary to create, incur, assume or permit to exist any Lien upon any of its Properties, or the proceeds thereof, whether now owned or hereafter acquired, except (a) Liens permitted pursuant to Section 6.2(h) (or such comparable section) of the Domestic Credit Agreement, <u>provided</u> that no such Liens (other than Permitted Liens) shall encumber the Collateral subject to the International Security Documents, and (b) Liens in favor of the Bank created under the International Loan Documents.
- 9.2 <u>Merger, Consolidation, Acquisitions, Sales of Assets, Etc.</u> Without the prior written consent of the Bank and Ex-Im Bank, the Borrower will not: (a) merge, consolidate or otherwise combine with any other Person, (b) acquire all or substantially all of the assets or capital stock of another Person; (c) sell, lease, transfer, assign or otherwise dispose of any of its assets, except for the sale of Inventory in the ordinary course of business and the disposition of obsolete equipment in the ordinary course of business; (d) make any material change in its organizational structure or identity, or (e) enter into any agreement to do any of the foregoing; except that the Borrower may consummate the Aldabra Transactions.
- 9.3 <u>Conduct of Business</u>. The Borrower shall not, and shall not permit any of its Subsidiaries to, engage in any business or activity other than (i) the businesses of dredging, aggregate mining and supply, towing services, marine construction, dredging reclamation activities and demolitions; and (ii) any businesses or activities reasonably related, complimentary or incidental thereto, or the commercial and industrial demolition business.
- 9.4 <u>Investments</u>. The Borrower will not, and will not permit any Subsidiary to make any Investments, except as permitted by <u>Section 6.2(b)</u> (or any comparable section) of the Domestic Credit Agreement or otherwise consented to by the Domestic Lenders under the Domestic Credit Agreement but not otherwise prohibited by <u>Section 9.2</u>. This <u>Section 9.4</u> shall not be construed to prohibit advances made to employees, officers and directors in the usual, customary and ordinary course of business.
 - 9.5 Financial Covenants.

(a) Capital Expenditures.

(i) The Borrower and its consolidated Subsidiaries shall not make or permit Capital Expenditures in an aggregate amount in excess of \$22,000,000 during any Fiscal Year (with respect to any such Fiscal Year, the "Base Capital Expenditure Amount"); provided, however, that the Base Capital Expenditure Amount for any Fiscal Year after Fiscal Year 2006 may be increased by (A) an amount equal to the excess, if any, of (i) the Base Capital Expenditure Amount for the immediately preceding Fiscal Year, over (ii) the actual amount of Capital Expenditures made by the Borrower and its Subsidiaries during such immediately preceding Fiscal Year; plus (B) the amount of Capital Expenditures permitted in the immediately succeeding Fiscal Year (provided that the Base Capital Expenditure Amount for such succeeding Fiscal Year shall be reduced by the amount of any increase pursuant to this Clause (B); and provided, further, that in no event shall the amount of Capital Expenditures made by the Borrower and its consolidated Subsidiaries in any Fiscal Year exceed \$26,000,000).

(b) Maximum Total Leverage.

(i) The Borrower and its consolidated Subsidiaries shall not permit the ratio (the "<u>Total Leverage Ratio</u>") of (i) the aggregate unpaid principal amount of Total Funded Debt as of the last day of any Fiscal Quarter ending during the periods described below to (ii) Adjusted Consolidated EBITDA for the four (4) consecutive Fiscal Quarter period ending as of such date, to exceed the corresponding ratio set forth below opposite such period:

<u>Period</u>	Ratio
Closing Date through and including September 30, 2006	5.60 to 1.00
October 1, 2006 through and including December 31, 2006	5.00 to 1.00
January 1, 2007 through and including December 31, 2007	4.75 to 1.00
January 1, 2008 through and including December 31, 2008	4.50 to 1.00
January 1, 2009 through and including December 31, 2009	4.00 to 1.00
January 1, 2010 and thereafter	3.50 to 1.00

(c) Maximum Senior Leverage.

(i) The Borrower and its consolidated Subsidiaries shall not permit the ratio (the "Senior Leverage Ratio") of (i) the aggregate unpaid principal amount of Senior Debt as of last day of any Fiscal Quarter ending during the periods described below to (ii) Adjusted Consolidated EBITDA for the four (4) consecutive Fiscal Quarter period ending as of such date, to exceed the corresponding ratio set forth below opposite such period:

<u>Period</u>	Ratio
January 1, 2006 through and including December 31, 2006	2.00 to 1.00
January 1, 2007 through and including December 31, 2007	1.75 to 1.00

January 1, 2008 through and including December 31, 2008	1.50 to 1.00
January 1, 2009 through and thereafter	1.25 to 1.00

(d) Interest Coverage Ratio.

(i) The Borrower and its consolidated Subsidiaries shall not permit the ratio (the "Interest Coverage Ratio") of (i) Adjusted Consolidated EBITDA for any four (4) consecutive Fiscal Quarter period ending as of the last day of any Fiscal Quarter ending during the periods described below to (ii) Interest Expense for such period ending as of such date, to be less than the corresponding ratio set forth below opposite such period:

Period Period	Ratio
January 1, 2006 through and including December 31, 2006	2.00 to 1.00
January 1, 2007 through and including December 31, 2008	2.25 to 1.00
January 1, 2009 through and thereafter	2.50 to 1.00

- 9.6 <u>Debt</u>. The Borrower will not, and will not permit any Subsidiary to create, incur, assume or otherwise become or remain liable with respect to any Debt, other than, without duplication, the following:
 - (a) Debt to the Bank; and
 - (b) Debt permitted by Section 6.2(i) (or any comparable section) of the Domestic Credit Agreement.
- 9.7 <u>Affiliate Transactions</u>. The Borrower shall not, nor shall it permit any Subsidiary to, enter into, or cause, suffer or permit to exist any arrangement or contract with any of its other Affiliates (as defined in the Domestic Credit Agreement) unless such arrangement or contract is fair and equitable to Borrower or such Subsidiary and is an arrangement or contract of the kind which would be entered into by a prudent Person in the position of Borrower or such Subsidiary with a Person that is not one of its Affiliates (as defined in the Domestic Credit Agreement); <u>provided</u>, <u>however</u>, that nothing in this <u>Section 9.7</u> shall prohibit (i) any transactions otherwise permitted hereby pursuant to any Management Agreement (as defined in the Domestic Credit Agreement) or any Tax Sharing Agreement (as defined in the Domestic Credit Agreement), (ii) any transaction among the Borrower and any Subsidiary Guarantor (as defined in the Domestic Credit Agreement) or (iii) the NASDI Restructuring (as defined in the Domestic Credit Agreement).
- 9.8 <u>Restricted Payments</u>. The Borrower will not declare, pay or make, or offer to pay or make any Restricted Payment (directly or indirectly through an Affiliate (as defined in the Domestic Credit Agreement)), except as permitted under Section 6.2(c) (or any comparable section) of the Domestic Credit Agreement.
- 9.9 <u>Suspension and Debarment, Etc</u>. Unless authorized by Ex-Im Bank, the Borrower will not knowingly enter into any transactions in connection with the Items with any Person who

is debarred, suspended, declared ineligible or voluntarily excluded from participation in procurement or nonprocurement transactions with any U.S. federal government department or agency pursuant to any of the Debarment Regulations. The Borrower will provide immediate written notice to the Bank if at any time it learns that the certification set forth in <u>Section 6.15</u> was erroneous when made or has become erroneous by reason of changed circumstances.

9.10 <u>Debt of Eligible Joint Ventures</u>. The Borrower shall not, nor shall it permit any Subsidiary, to consent to the incurrence of Debt by any Eligible Joint Venture.

ARTICLE X. EVENTS OF DEFAULT; REMEDIES.

- 10.1 Events of Default. If any of the following events shall occur, then the Bank may (a) by written notice to the Borrower, declare the Commitment of the Bank and the obligation of the Bank to issue or cause to be issued Letters of Credit to be terminated, whereupon the same shall forthwith terminate, and (b) by written notice to the Borrower declare all outstanding Obligations, to be forthwith due and payable, whereupon all outstanding Obligations shall become and be forthwith due and payable without presentment, demand, protest, or further notice of any kind (including, without limitation, notice of default, notice of intent to accelerate and notice of acceleration), all of which are hereby expressly waived by the Borrower; provided, however, that with respect to any Event of Default described in Sections 10.1(d), (e) and (f) hereof, the Commitment of the Bank and the obligation of the Bank to issue or cause to be issued Letters of Credit hereunder shall automatically be terminated and all outstanding Obligations shall automatically become immediately due and payable, without presentment, demand, protest, or any notice of any kind (including, without limitation, notice of default, notice of intent to accelerate and notice of acceleration), all of which are hereby expressly waived by the Borrower:
 - (a) <u>Failure to Make Payments When Due</u>. The Borrower does not pay, repay or prepay on the date when due (i) any Unpaid Drawings or (ii) any other Obligations and such failure shall continue unremedied for five (5) Business Days; or
 - (b) <u>Misrepresentation or Breach of Warranty.</u> Any representation or warranty made by any Loan Party herein, in any other International Loan Document, or in any amendment, waiver or modification thereof shall be incorrect, false or misleading in any material respect when made; or

(c) Violation of Covenants.

- (i) Any Loan Party shall fail duly and punctually to perform or observe any covenant or agreement binding on such Loan Party under Section 8.4, 8.5 (other than 8.5(k)), 8.9(a), 8.12, 8.15 or Article IX of this Agreement; or
- (ii) Any Loan Party shall fail duly and punctually to perform or observe any covenant or agreement binding on such Loan Party under Section 8.1, 8.2 and 8.5(k) of this Agreement, and such failure shall continue unremedied for ten (10) Business Days after an Authorized Officer of such Loan Party first has knowledge of such failure or such Loan Party receives written notice thereof from the Bank, whichever is earlier; or

- (iii) Any Loan Party shall fail duly and punctually to perform or observe any covenant or agreement binding on such Loan Party under this Agreement (other than as provided in subsection (a) above or in clause (i) or (ii) of this subsection (c)) or under any of the other International Loan Documents (other than the Guaranty), and such failure shall continue unremedied for thirty (30) days after an Authorized Officer first has knowledge of such failure or such Loan Party receives written notice thereof from the Bank, whichever is earlier; or
- (d) <u>Bankruptcy and Other Matters</u>. Any Loan Party (A) applies for, consents to or suffers the appointment of, or the taking of possession by, a receiver, a custodian, trustee, liquidator or similar fiduciary of itself or of all or a substantial part of its property or calls a meeting of its creditors, (B) admits in writing its ability, or is generally unable, to pay its debts as they become due or ceases operations of its present business, (C) makes a general assignment for the benefit of creditors, (D) commences a voluntary case under any state or federal bankruptcy laws (as now or hereafter in effect), (E) is adjudicated as bankrupt or insolvent, (F) files a petition seeking to take advantage of any other law providing for the relief of debtors, (G) acquiesces to, or fails to have dismissed within thirty (30) days, any petition filed against it in any involuntary case under such bankruptcy laws, or (H) takes any action for the purpose of effecting any of the foregoing; or

(e) Bankruptcy of Subsidiary.

- (i) Any Subsidiary of the Borrower (other than a Loan Party) shall generally not pay its debts as such debts become due, or shall admit in writing its inability to pay its debts generally, or shall make a general assignment for the benefit of creditors; or
- (ii) any proceeding shall be instituted by or against any Subsidiary of the Borrower (other than a Loan Party) seeking to adjudicate it a bankrupt or insolvent, or seeking liquidation (other than in a transaction permitted under Section 6.2(a)(i) or any comparable section of the Domestic Credit Agreement), winding up, reorganization, arrangement, adjustment, protection, relief or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors or seeking the entry of an order for relief or the appointment of a receiver, trustee or other similar official for it or for any substantial part of its property and, in the case of any such proceeding instituted against it (but not instituted by it), either such proceeding shall remain undismissed, undischarged, unvacated, unbonded or unstayed for a period of sixty (60) days, or any of the actions sought in such proceeding (including the entry of an order for relief against, or the appointment of a receiver, trustee, custodian or other similar official for, it or any substantial part of its property) shall occur; or
- (iii) any Subsidiary of the Borrower (other than a Loan Party) shall take any corporate action pursuant to a resolution or consent of its board of directors or shareholders to authorize any of the actions set forth in this <u>Section 10.1(e)</u>.

- (f) Dissolution. Any proceeding is commenced by or against any Loan Party for the liquidation of its assets or dissolution; or
- (g) Collateral and Liens.
- (i) Any Lien in any of the Collateral, granted or intended by the International Loan Documents to be granted to the Bank, ceases to be a valid, enforceable, perfected, first priority Lien (or lesser priority if expressly permitted pursuant to Section 6 of the Loan Authorization Agreement) subject only to Permitted Liens; or
- (ii) any levy, assessment, attachment, seizure or Lien, other than a Permitted Lien, is made against any of the Collateral which is not stayed or lifted within thirty (30) calendar days; or
- (h) <u>Cross-default to Export Orders</u>. The Borrower is in breach in any material respect of any material provision of any Export Order included in the International Borrowing Base, which has not been waived by the applicable party thereto; or

(i) Cross-default to Domestic Loan Documents.

- (i) After giving effect to any applicable grace or cure period, an "Event of Default" under and as defined in the Domestic Credit Agreement shall have occurred, unless the same shall have been waived by the Domestic Lenders in accordance with the terms of the Domestic Credit Agreement; or
- (ii) or, unless waived in writing by Ex-Im Bank, the Domestic Credit Agreement has been terminated or ceases to be in full force and effect for any reason; or
- (j) <u>Cross-default to Non-Ex-Im Bank Debt</u>. The Borrower or any Guarantor fails to pay within thirty (30) calendar days of the date when due any amount payable to the Bank under any loan or other extension of credit from the Bank to the Borrower or such Guarantor which is not guaranteed by Ex-Im Bank, unless the same shall have been waived by the Bank in accordance with the terms thereof; or

(k) Cross-acceleration to Other Debt.

- (i) The Borrower or any Guarantor shall fail to pay any Debt in an aggregate principal amount in excess of \$5,000,000 at its final scheduled maturity date; or
- (ii) any breach or event of default shall occur under any instrument, agreement or indenture pertaining to any Debt referred to in clause (i) above, if the effect thereof, after giving effect to any applicable grace or cure period, is to accelerate the maturity of such Debt or cause such Debt to be declared due and payable or required to be prepaid (other than by a regularly scheduled required

prepayment), repurchased or redeemed prior to the originally stated maturity thereof; or

- (l) <u>Undischarged Judgment</u>. Final judgment shall be rendered against any Loan Party or Subsidiary of the Borrower in an amount in excess of \$5,000,000 (to the extent not covered by insurance) and the same shall remain undischarged for a period of 30 days during which execution shall not be effectively stayed, released, bonded or vacated; or
- (m) <u>Cross-default to Borrower Agreement, Loan Authorization Agreement and Fast Track Borrower Supplement</u>. The Borrower fails to comply with any provision contained in the Borrower Agreement, the Loan Authorization Agreement or the Fast Track Borrower Supplement and such failure is not cured within thirty (30) calendar days after the occurrence of such failure; or
 - (n) Change of Control. A Change of Control (as defined in the Domestic Credit Agreement) shall occur; or
- (o) <u>Impairment of Ex-Im Bank Guaranty</u>. The Ex-Im Bank Guaranty shall cease to be in full force and effect or shall be declared null and void or the validity or enforceability thereof shall be contested, challenged or denied by Ex-Im Bank, the Borrower, any Governmental Authority, any Affiliate of the Borrower or any Subsidiary or any Guarantor; or
- (p) <u>Impairment of International Loan Documents, Export Orders</u>. Any material provision of this Agreement, the Note, any Export Order included in the International Borrowing Base, or any other International Loan Document shall cease for any reason (other than by reason of any action or inaction by the Bank) to be in full force and effect against any Loan Party (other than in accordance with its terms) or shall be declared null and void or the validity or enforceability hereof or thereof shall be contested, challenged or denied by the Borrower, any Governmental Authority, any Affiliate of the Borrower or any Subsidiary of such Borrower or any Guarantor; or
- (q) <u>Distributions in Violation of the Ex-Im Bank Guaranty</u>. The Borrower shall have in good faith requested and the Bank shall have made a Disbursement that unintentionally turns out to be in violation of the Ex-Im Bank Guaranty, and the Borrower shall not have repaid such Disbursement promptly upon learning of such violation;
- (r) <u>Material Litigation</u>. Any litigation is filed against the Borrower or any Guarantor which has had or could reasonably be expected to have an Ex-Im Material Adverse Effect and such litigation is not withdrawn or dismissed within thirty (30) calendar days of the filing thereof; or
- (s) <u>Guaranty</u>. Any breach or default occurs under the Guaranty, or the Guaranty is terminated or any obligation to perform thereunder is terminated, or any Guarantor attempts to revoke the Guaranty.

10.2 Other Remedies. In addition to and cumulative of any rights or remedies expressly provided for in this Article X, if any one or more Events of Default shall have occurred (other than an Exit Date Event of Default), the Bank may proceed to protect and enforce its rights hereunder by any appropriate proceedings and the Liens evidenced by the International Security Documents shall be subject to foreclosure in any manner provided for therein or provided for by law as the Bank may elect. The Bank may also proceed either by the specific performance of any covenant or agreement contained in this Agreement or the other International Loan Documents or otherwise existing under any law. The Bank shall not, however, be under any obligation to marshal any assets in favor of the Borrower or against or in payment of any or all obligations under any International Loan Document. Upon the occurrence of an Event of Default (other than an Exit Date Event of Default), it is hereby acknowledged and agreed Ex-Im Bank shall have the right to (a) direct the Bank to make demand hereunder and under the other International Loan Documents and (b) request that the Bank declare all outstanding Obligations, to be forthwith due and payable, and the Bank shall have no liability to the Borrower for complying with such directions or requirements.

10.3 <u>Collateral Account</u>. The Borrower hereby agrees that in the event of (a) the occurrence of the Maturity Date, (b) the termination of the Commitment or (c) upon the occurrence and during the continuance of an Event of Default (other than an Exit Date Event of Default), it shall, if requested by the Bank in writing, provide to the Bank Cover equal to 100% of the then aggregate amount of Letter of Credit Outstandings, which Cover shall be held by the Bank in a collateral account to be maintained by the Bank. The Borrower hereby agrees to execute and deliver to the Bank such security agreements, pledges or other documents as the Bank may, from time to time, require to perfect the pledge, lien and security interest in and to any such funds provided for in this <u>Section 10.3</u>. Upon the payment or expiry of all Letter of Credit Outstandings, all such Collateral shall be promptly released to the Borrower in due course at Borrower's cost.

10.4 Exit Date Event of Default. On the date that is thirty (30) days after the date that Wells Fargo notifies the Borrower that it is assigning its interests in the Domestic Credit Agreement to a Person other than Wells Fargo, an Affiliate of Wells Fargo or an Affiliate of the Bank and an Exit Date has occurred during such thirty day period, then an Exit Date Event of Default shall be deemed to have occurred and the Commitment of the Bank and the obligation of the Bank to issue or cause to be issued Letters of Credit hereunder shall (unless Ex-Im Bank shall have consented to the issuance of additional Letters of Credit) automatically be terminated, whereupon the same shall forthwith terminate and no additional Letters of Credit shall be issued without Ex-Im Bank's written consent.

Notwithstanding anything to the contrary, if the only Event of Default that has occurred and is continuing is an Exit Date Event of Default, then the Bank and the Secured Party (as defined in the International Security Agreement) shall not have the right to exercise any right or remedy (other than those rights and remedies set forth in this Section 10.4) that may only be exercised upon the occurrence and during the continuance of an Event of Default.

10.5 <u>Remedies Cumulative</u>. No remedy, right or power conferred upon the Bank is intended to be exclusive of any other remedy, right or power given hereunder or now or hereafter

existing at law, in equity, or otherwise, and all such remedies, rights and powers shall be cumulative.

ARTICLE XI. MISCELLANEOUS.

- 11.1 Waivers, Amendments. No failure or delay on the part of the Bank in exercising any power or right hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any such right or power, or any abandonment or discontinuance of steps to enforce such a right or power, preclude any other or further exercise thereof or the exercise of any other right or power. No course of dealing between the Borrower or any Subsidiary of such Borrower, any Guarantor and the Bank shall operate as a waiver of any right of the Bank or Ex-Im Bank. No amendment, modification or waiver of any provision of this Agreement, the Note or any other International Loan Document nor consent to any departure by the Borrower therefrom shall in any event be effective unless the same shall be (i) approved by Ex-Im Bank and (ii) in writing signed by all the parties thereto, and then such waiver or consent shall be effective only in the specific instance and for the purpose for which given. Except as otherwise provided by any International Loan Document or applicable law, no notice to or demand on the Borrower in any case shall entitle the Borrower to any other or further notice or demand in similar or other circumstances.
- 11.2 Reimbursement of Expenses. WHETHER OR NOT THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT SHALL BE CONSUMMATED, THE BORROWER AGREES TO REIMBURSE THE BANK FOR ITS REASONABLE OUT-OF-POCKET EXPENSES, INCLUDING THE REASONABLE FEES AND EXPENSES OF OUTSIDE COUNSEL TO THE BANK, IN CONNECTION WITH SUCH TRANSACTIONS, OR ANY OF THEM, OR OTHERWISE IN CONNECTION WITH THIS AGREEMENT OR ANY OTHER INTERNATIONAL LOAN DOCUMENT, INCLUDING (A) THE NEGOTIATION, PREPARATION, EXECUTION, ADMINISTRATION, MODIFICATION AND ENFORCEMENT OF THIS AGREEMENT OR ANY OTHER INTERNATIONAL LOAN DOCUMENT, AND (B) REASONABLE OUT-OF-POCKET COSTS AND EXPENSES OF THE BANK IN CONNECTION WITH DUE DILIGENCE, TRANSPORTATION AND DUPLICATION. THE BORROWER AGREES TO PAY ANY AND ALL STAMP, DOCUMENTARY AND OTHER SIMILAR TAXES WHICH MAY BE PAYABLE OR DETERMINED TO BE PAYABLE IN CONNECTION WITH THE EXECUTION AND DELIVERY OF THIS AGREEMENT, THE NOTE OR ANY OTHER INTERNATIONAL LOAN DOCUMENT, AND TO SAVE THE BANK HARMLESS FROM ANY AND ALL LIABILITIES WITH RESPECT TO OR RESULTING FROM ANY DELAY OR OMISSION TO PAY ANY SUCH TAXES. THE OBLIGATIONS OF THE BORROWER UNDER THIS SECTION 11.2 SHALL SURVIVE THE TERMINATION OF THIS AGREEMENT AND/OR THE PAYMENT OF THE OBLIGATIONS.
- 11.3 <u>Notices</u>. All notices and other communications provided for herein shall be in writing (including facsimile or cable communication) and shall be mailed, telecopied, cabled or delivered, addressed as follows:
 - (a) If to the Borrower or any Guarantor:

Great Lakes Dredge & Dock Corporation 2122 York Road Oak Brook, Illinois 60523 Attention: Ms. Deborah A. Wensel

Telephone No.: 630-574-2948 Telefax No.: 630-574-3007

(b) If to the Bank:

Wells Fargo HSBC Trade Bank, N.A. 1000 Louisiana Street, Fourth Floor Houston, Texas 77002 Attention: Mr. Scott Gildea

Telephone No.: (713) 319-1374 Telefax No.: (713) 739-1087

or to such other address as shall be designated by such party in a written notice to the other party. All such notices and communications shall, when mailed, telecopied, transmitted, cabled, or sent by reputable overnight courier become effective (i) five (5) days after deposited in the mail, (ii) when transmitted to the telecopier, or delivered to the cable company, or (iii) on the following Business Day after deposit with the overnight courier, except that notices and communications to the Bank shall not be effective until actually received by the Bank.

- 11.4 <u>Governing Law</u>. THIS AGREEMENT, THE OTHER INTERNATIONAL LOAN DOCUMENTS (UNLESS ANY SUCH INTERNATIONAL LOAN DOCUMENT EXPRESSLY PROVIDES OTHERWISE), AND THE LETTERS OF CREDIT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO ITS CONFLICTS OF LAW PRINCIPLES (OTHER THAN THE PROVISIONS OF 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW).
- 11.5 <u>Survival of Representations</u>, <u>Warranties and Covenants</u>. All representations, warranties and covenants contained herein or made in writing by the Borrower and the Guarantors in connection herewith shall survive the execution and delivery of this Agreement and the Note, and will bind and inure to the benefit of the respective successors and assigns of the parties hereto, whether so expressed or not. No investigation at any time made by or on behalf of the Bank shall diminish the Bank's right to rely thereon.
- 11.6 <u>Counterparts</u>. This Agreement may be executed in several counterparts, and by the parties hereto on separate counterparts, and each counterpart, when so executed and delivered, shall constitute an original instrument, and all such separate counterparts shall constitute but one and the same instrument.
- 11.7 <u>Separability</u>. Should any clause, sentence, paragraph or Section of this Agreement be judicially declared to be invalid, unenforceable or void, such decision shall not have the effect of invalidating or voiding the remainder of this Agreement, and the parties hereto agree that the part or parts of this Agreement so held to be invalid, unenforceable or void will be

deemed to have been stricken herefrom and the remainder will have the same force and effectiveness as if such part or parts had never been included herein.

- 11.8 <u>Limitation of Liability</u>. No claim may be made by the Borrower or the Affiliates, directors, officers, employees, attorneys, or agents of the Borrower against the Bank or the Affiliates, directors, officers, employees, attorneys, or agents of the Bank for any special, indirect, consequential or punitive damages in respect of any claim for breach of contract arising out of or related to the transactions contemplated by this Agreement, or any act, omission, or event occurring in connection herewith; and the Borrower hereby waives, releases, and agrees not to sue upon any claim for any such damages, whether or not accrued and whether or not known or suspected to exist in its favor.
- 11.9 Set-off. If an Event of Default has occurred and is continuing, the Borrower hereby gives and confirms to the Bank a right of set-off of all moneys, securities and other property of the Borrower (whether special, general or limited) and the proceeds thereof, now or hereafter delivered to remain with or in transit in any manner to the Bank, its correspondents or its agents from or for the Borrower, whether for safekeeping, custody, pledge, transmission, collection or otherwise or coming into possession of the Bank in any way, and also, any balance of any deposit accounts and credits of the Borrower with, and any and all claims of security for the payment of the Obligations, and the Borrower hereby authorize the Bank at any time or times, without prior notice, to apply such money, securities, other property, proceeds, balances, credits of claims, or any part of the foregoing, to such Obligations in such amounts as it may select, whether such liabilities be contingent, unmatured or otherwise, and whether any collateral security therefor is deemed adequate or not. The rights described herein shall be in addition to any collateral security described in any separate agreement executed by the Borrower.
- 11.10 <u>Sale or Assignment</u>. The Bank may assign its rights hereunder and under each other International Loan Document and the Liens granted pursuant to the International Security Documents only to Ex-Im Bank in accordance with the terms and conditions of the Ex-Im Bank Guaranty. Notwithstanding the foregoing, the Bank may assign, transfer, negotiate, sell or participate all or part of its interests and rights in the Letters of Credit to an Affiliate or Subsidiary of the Bank, provided that the Bank retains all obligations under the Ex-Im Bank Guaranty with respect to Ex-Im Bank and the International Loan Documents.
- 11.11 Interest. All agreements between the Borrower, any Guarantor and the Bank, whether now existing or hereafter arising and whether written or oral, are hereby expressly limited so that in no contingency or event whatsoever, whether by reason of demand being made on the Note or otherwise, shall the amount paid, or agreed to be paid, to the Bank for the use, forbearance, or detention of the money to be loaned under this Agreement or otherwise or for the payment or performance of any covenant or obligation contained herein or in any other International Loan Document exceed the Highest Lawful Rate. If, as a result of any circumstances whatsoever, fulfillment of any provision hereof or of any of such documents, at the time performance of such provision shall be due, shall involve transcending the limit of validity prescribed by applicable usury law, then, ipso facto, the obligation to be fulfilled shall be reduced to the limit of such validity, and if, from any such circumstance, the Bank shall ever receive interest or anything which might be deemed interest under applicable law which would exceed the Highest Lawful Rate, such amount which would be excessive interest shall be applied

to the reduction of the principal amount owing on account of the Obligations or the amounts owing on other obligations of the Borrower or any Guarantor to the Bank under any International Loan Document and not to the payment of interest, or if such excessive interest exceeds the unpaid principal balance of the Obligations and the amounts owing on other obligations of the Borrower and any Guarantor to the Bank under any International Loan Document, as the case may be, such excess shall be refunded to the Borrower or such Guarantor, as applicable. All sums paid or agreed to be paid to the Bank for the use, forbearance, or detention of the indebtedness of the Borrower and the Guarantors to the Bank shall, to the extent permitted by applicable law, be amortized, prorated, allocated, and spread throughout the full term of such indebtedness until payment in full of the principal thereof (including the period of any renewal or extension thereof) so that the interest on account of such indebtedness shall not exceed the Highest Lawful Rate. Notwithstanding anything to the contrary contained in this Agreement or the Note, it is understood and agreed that if at any time the rate of interest which accrues on the outstanding Obligations shall exceed the Highest Lawful Rate, the rate of interest which accrues on the outstanding Obligations in the rate of interest which accrues on the outstanding Obligations below the Highest Lawful Rate until the total amount of interest accrued on the outstanding Obligations equals the amount of interest which would have accrued if such interest rate had at all times been in effect. The terms and provisions of this Section 11.11 shall control and supersede every other provision of all agreements between the Borrower, the Guarantors and the Bank.

11.12 Indemnification. EACH OF THE LOAN PARTIES AGREES TO INDEMNIFY, DEFEND, AND SAVE HARMLESS THE BANK AND ITS OFFICERS, DIRECTORS, EMPLOYEES, AGENTS, AND ATTORNEYS, AND EACH OF THEM (THE "INDEMNIFIED PARTIES"), FROM AND AGAINST ALL CLAIMS, ACTIONS, SUITS, AND OTHER LEGAL PROCEEDINGS, DAMAGES, COSTS, INTEREST, CHARGES, REASONABLE COUNSEL FEES, AND OTHER EXPENSES AND PENALTIES WHICH ANY OF THE INDEMNIFIED PARTIES MAY SUSTAIN OR INCUR BY REASON OF OR ARISING OUT OF (I) THE ISSUANCE OF ANY LETTER OF CREDIT HEREUNDER, THE EXECUTION AND DELIVERY OF THIS AGREEMENT, THE NOTE AND THE OTHER INTERNATIONAL LOAN DOCUMENTS AND THE CONSUMMATION OF THE TRANSACTIONS CONTEMPLATED THEREBY AND THE EXERCISE AND ENFORCEMENT OF ANY OF THE BANK'S RIGHTS UNDER THIS AGREEMENT, THE NOTE AND THE OTHER INTERNATIONAL LOAN DOCUMENTS OR OTHERWISE, INCLUDING, WITHOUT LIMITATION, DAMAGES, COSTS, AND EXPENSES INCURRED BY ANY OF THE INDEMNIFIED PARTIES IN INVESTIGATING, PREPARING FOR, DEFENDING AGAINST, OR PROVIDING EVIDENCE, PRODUCING DOCUMENTS, OR TAKING ANY OTHER ACTION IN RESPECT OF ANY COMMENCED OR THREATENED LITIGATION UNDER ANY FEDERAL SECURITIES LAW OR ANY SIMILAR LAW OF ANY JURISDICTION OR AT COMMON LAW OR (II) ANY VIOLATIONS OF ANY LAW OR ANY ACTION TAKEN OR OMITTED TO BE TAKEN WHICH COULD RESULT IN A VIOLATION OF ANY ENVIRONMENTAL LAW, OR IMPOSE ANY LIABILITY ON THE INDEMNIFIED PARTIES FOR DAMAGE TO HEALTH OR THE ENVIRONMENT; PROVIDED HOWEVER THAT NO INDEMNIFIED PARTY SHALL BE ENTITLED TO THE BENEFITS OF THIS SECTION 11.12 TO THE EXTENT OF ANY BAD FAITH, GROSS NEGLIGENCE, OR WILLFUL MISCONDUCT BY AN

INDEMNIFIED PARTY. THIS AGREEMENT IS INTENDED TO PROTECT AND INDEMNIFY THE INDEMNIFIED PARTIES AGAINST ALL RISKS HEREBY ASSUMED BY THE LOAN PARTIES. THE OBLIGATIONS OF THE LOAN PARTIES UNDER THIS SECTION 11.12 SHALL SURVIVE ANY EXERCISE OF THE POWER OF SALE GRANTED IN ANY INTERNATIONAL SECURITY DOCUMENT TO WHICH ANY OF THE LOAN PARTIES ARE A PARTY, ANY FORECLOSURE OF THE LIENS CREATED BY ANY INTERNATIONAL SECURITY DOCUMENT TO WHICH ANY OF THE LOAN PARTIES ARE A PARTY, OR CONVEYANCE IN LIEU OF FORECLOSURE, THE REPAYMENT OF THE OBLIGATIONS, THE DISCHARGE AND RELEASE OF ANY PERSON UNDER ANY INTERNATIONAL LOAN DOCUMENT AND ANY TERMINATION OF THIS AGREEMENT.

- 11.13 <u>Payments Set Aside</u>. To the extent that the Borrower makes a payment or payments to the Bank or the Bank enforces any security interest or exercises its right of setoff, and such payment or payments or the proceeds of such enforcement or setoff or any part thereof are subsequently invalidated, declared to be fraudulent or preferential, set aside and/or required to be repaid to a trustee, receiver or any other Person under any Debtor Law or equitable cause, then, to the extent of such recovery, the obligation or part thereof originally intended to be satisfied, and all rights and remedies therefor, shall be revived and shall continue in full force and effect as if such payment had not been made or such enforcement or setoff had not occurred.
- 11.14 <u>Credit Agreement Controls</u>. If there are any conflicts or inconsistencies between this Agreement and any of the other International Loan Documents, the provisions of this Agreement shall prevail and control.
- 11.15 <u>Amendment of Financial Covenants</u>. Notwithstanding anything herein to the contrary, the parties hereto agree that any amendment to or waiver to the financial covenants set forth in <u>Section 6.3</u> (or any comparable section) of the Domestic Credit Agreement or any related term or definition used therein shall, subject to the prior written approval of Ex-Im Bank and Bank, automatically amend or waive the corresponding financial covenants set forth in <u>Section 9.5</u> of this Agreement and any related term or definition used therein in the same manner. Subject to the prior written approval of Ex-Im Bank, the Bank agrees to execute an amendment or waiver to this Agreement to document such automatic amendment or waiver of the financial covenants and related terms and conditions.
- 11.16 Waiver of Jury Trial. TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAWS, THE PARTIES HERETO HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHTS OR REMEDIES UNDER THIS AGREEMENT OR INTERNATIONAL LOAN DOCUMENTS, OR ARISING FROM ANY FINANCING RELATIONSHIP EXISTING IN CONNECTION WITH THIS AGREEMENT OR ANY INTERNATIONAL LOAN DOCUMENT AND AGREE THAT ANY ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.
- 11.17 <u>USA Patriot Act Notice</u>. The Bank hereby notifies Borrower that, pursuant to the requirements of the USA Patriot Act, and the Bank's policies and practices, the Bank is required to obtain, verify and record certain information and documentation that identifies Borrower,

which information includes the name and address of Borrower and such other information that will allow the Bank to identify Borrower in accordance with the USA Patriot Act. In addition, Borrower shall (i) ensure that no Person who owns a controlling interest in or otherwise controls Borrower or any Subsidiary of Borrower is or shall be listed on the Specially Designated Nationals and Blocked Person List or other similar lists maintained by the Office of Foreign Assets Control ("OFAC"), the Department of the Treasury or included in any Executive Orders, (ii) not use or permit the use of the proceeds of the Loan to violate any of the foreign asset control regulations of OFAC or any enabling statute or Executive Order relating thereto, and (iii) comply, and cause its Subsidiaries to comply, with all applicable laws.

- 11.18 <u>Time of the Essence</u>. Time is of the essence of the International Loan Documents.
- 11.19 <u>Final Agreement</u>. THIS WRITTEN AGREEMENT AND THE INTERNATIONAL LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.
- 11.20 <u>Release of Collateral and Guarantors</u>. Subject to the prior written approval of Ex-Im Bank, the Bank hereby agrees to release, or cause to be released, the Liens held by it under the International Loan Documents and the Guaranty upon payment in full in cash of all the Obligations (other than unasserted contingent and indemnification obligations), termination of the Commitment and reduction of the Letter of Credit Outstandings to zero (or the making of other arrangements satisfactory to the Bank).
- 11.21 <u>Additional Guarantors</u>. Each Person that becomes a Guarantor after the Closing Date shall execute and deliver a Joinder to Guaranty in substantially the form attached hereto as <u>Exhibit G</u> and shall thereafter be a party to this Agreement as a Guarantor.

[The Balance of This Page is Left Blank Intentionally]

IN WITNESS WHEREOF, the parties hereto, by their respective officers thereunto duly authorized, have executed this Agreement effective as of the date first written above.

BORROWER:

GREAT LAKES DREDGE & DOCK CORPORATION

By: /s/ Deborah A. Wensel

Deborah A. Wensel

Senior Vice President, Chief Financial Officer and

Treasurer

GUARANTOR:

GREAT LAKES DREDGE & DOCK COMPANY, LLC

By: /s/ Deborah A. Wensel

Deborah A. Wensel

Senior Vice President, Chief Financial Officer and

Treasurer

BANK:

WELLS FARGO HSBC TRADE BANK, N.A.

By: /s/ Scott Gildea

Scott Gildea Vice President

EXHIBIT "A"

CERTAIN DEFINITIONS

As used in the Agreement, the following words and terms shall have the respective meanings indicated opposite each of them:

"1933 Act" means the Securities Act of 1933, as amended from time to time.

"Accounts Receivable" shall mean all of the Loan Parties' now owned or hereafter acquired (a) "accounts" (as such term is defined in the UCC), other receivables, book debts and other forms of obligations, whether arising out of goods sold or services rendered or from any other transaction; (b) rights in, to and under all purchase orders or receipts for goods or services; (c) rights to any goods represented or purported to be represented by any of the foregoing (including unpaid sellers' rights of rescission, replevin, reclamation and stoppage in transit and rights to returned, reclaimed or repossessed goods); (d) moneys due or to become due to such Borrower under all purchase orders and contracts (which includes Export Orders) for the sale of goods or the performance of services or both by Borrower (whether or not yet earned by performance on the part of Borrower), including the proceeds of the foregoing; (e) any notes, drafts, letters of credit, insurance proceeds or other instruments, documents and writings evidencing or supporting the foregoing; and (f) all collateral security and guarantees of any kind given by any other Person with respect to any of the foregoing.

"Adjusted Consolidated EBITDA" means EBITDA plus adjustments, without duplication, required or permitted by Regulation S-X of the 1933 Act.

"Affiliate" shall mean (i) all persons, companies or other entities owning or otherwise controlling twenty percent (20%) or more of the voting share capital (or equivalent right of ownership) of the Borrower, or having the power to direct the Borrower's policies or management whether by contract or otherwise (each such person, company or other entity, a "Controlling Affiliate"), (ii) all companies or other entities in which a Controlling Affiliate owns or otherwise controls twenty percent (20%) or more of the voting share capital (or equivalent right of ownership) or has the power to direct the policies or management of, whether by contract or otherwise and (iii) all companies or other entities in which the Borrower owns or otherwise controls twenty percent (20%) or more of the voting share capital (or equivalent right of ownership) of or has the power to direct the policies or management of, whether by contract or otherwise.

"Agreement" shall mean this International Letter of Credit Agreement, as the same may be amended, restated, modified, supplemented or otherwise changed from time to time.

"Ahli United Bank Deposit Account" shall have the meaning set forth in Section 8.7.

"Aldabra Transactions" means the mergers and other transactions contemplated by the Aldabra Transaction Documents.

"Aldabra Transaction Documents" means that certain Agreement and Plan of Merger dated as of June 20, 2006 (the "Merger Agreement") by and among GLDD Acquisitions Corp., a Delaware corporation, Aldabra Acquisition Corporation, a Delaware corporation, Aldabra Merger Sub, L.L.C., a Delaware limited liability company, Madison Dearborn Capital Partners IV, L.P., a Delaware limited partnership, solely in its capacity as Company Representative, and Terrapin Partners LLC, solely in its capacity as Buyer Representative, and such other documents, instruments and agreements executed and/or delivered pursuant to the Merger Agreement, all as the same may be amended, restated, supplemented or otherwise modified from time to time.

"Annual Audited Financial Statements" shall mean, as to any Person, the year-end consolidated financial statements of such Person (consolidated balance sheet, consolidated statement of earnings, and consolidated cash flow statements), all prepared in conformity with GAAP and audited by Deloitte & Touche LLP or other independent certified public accountants reasonably satisfactory to the Bank and Ex-Im Bank. Annual Financial Statements shall also include unaudited consolidating financial statements certified on behalf of such Person by an Authorized Officer of such Person who is the chief financial officer, treasurer, assistant treasurer or controller of the Borrower, as fairly presenting in all material respects the consolidating financial condition of such Person and its Subsidiaries.

"Annual Facility Fee Percentage" shall mean 1.00%.

- "Application" shall mean an application, in such form as the Bank or Wells Fargo may specify from time to time, requesting the Bank cause a Letter of Credit to be issued.
- "Approved Currency" means (a) Bahrain Dinars, (b) so long as the Borrower enters into Rate Protection Agreements if required by Ex-Im Bank and satisfactory Ex-Im Bank and Bank, (i) Qatari Riyals, (ii) Kuwati Dinars, (iii) UAE Dirhams, (iv) Euros, (v) British Pound Sterling, and (vi) Canadian Dollars and (c) subject to the prior written approval of the Bank and Ex-Im Bank, any other lawful currency other than Dollars which is freely transferable and convertible into Dollars.
- "<u>Authorized Officer</u>" means, with respect to any Loan Party, the President, Chief Executive Officer, Chief Financial Officer, Treasurer, any Vice President or Secretary of such Loan Party whose signatures and incumbency shall have been certified to the Bank pursuant to <u>Section 7.5</u>.
 - "Bahrain Dinar" means mean the lawful currency of the Kingdom of Bahrain.
 - "Base Capital Expenditure Amount" have the meaning specified in Section 9.5(a)(ii).
- "Borrower" shall mean the Original Borrower and from and after the consummation of the Aldabra Transactions, the Successor Borrower, and any successors and assigns.
- "Borrower Agreement" shall mean the Borrower Agreement dated as of the Closing Date executed by the Borrower and the Bank, as amended, restated, supplemented or otherwise modified from time to time.

"Business Day" means a day of the year on which banks are not required or authorized to close in any of the Kingdom of Bahrain, New York City, San Francisco or Chicago.

"Buyer" shall mean an entity which has entered into one or more Export Orders with any Loan Party or an Eligible Joint Venture or who is an obligor on Export-Related Accounts Receivable.

"Capital Expenditures" means, for any period, the aggregate amount of all expenditures of the Borrower and its Subsidiaries for fixed or capital assets made or incurred during such period (whether or not paid in cash and including that portion of Capitalized Leases which is capitalized on the consolidated balance sheet of the Borrower and its Subsidiaries) which, in accordance with GAAP, would be classified as capital expenditures; provided, however, that, for any such period, such aggregate amount shall be reduced by the sum of (in each case to the extent the following would otherwise be required to be capitalized on the consolidated balance sheet of the Borrower and its Subsidiaries) (a) proceeds received from the sale of fixed or capital assets which have been applied, within one year of receipt thereof, to the purchase or cost of design, installation, construction, repair or improvement of fixed or capital assets used or useful in the business of the Borrower and its Subsidiaries; (b) insurance or requisition proceeds or condemnation awards received in connection with the damage, destruction, requisition or condemnation of fixed or capital assets which have been applied, within one year of receipt thereof, to the purchase or cost of design, installation, construction, repair or improvement of fixed or capital assets used or useful in the business of the Borrower and its Subsidiaries; (c) proceeds of indemnity claims made by Holdings or the Borrower pursuant to the Acquisition Agreement (as defined in the Domestic Credit Agreement) to the extent used in connection with the purchase or cost of design, installation, construction, repair or improvement of fixed or capital assets used or useful in the business of the Borrower and its Subsidiaries in connection with such indemnity claim; (d) with regard to equipment purchased simultaneously with the trade-in of existing equipment of the Borrower or its Subsidiaries, the amount of the credit extended for such trade-in; (e) expenditures for vessels identified to be sold by the Borrower or any of its Subsidiaries and then leased-back (on an operating lease basis) by the Borrower or any of its Subsidiaries to the extent that such sale occurs within 180 days of the completion of such vessel and the Borrower has notified the Domestic Lenders in writing in reasonable detail of the timing, facts and circumstances of such sale and lease-back to the extent that the aggregate amount of such expenditures does not exceed \$15,000,000 at any time (provided that if any vessel so identified is not sold, the amount of the expenditures made that relate to such vessel shall be added back to the amount of Capital Expenditures for the Fiscal Quarter in which such expenditure was made), and (f) the repurchase of the vessels set forth on Schedule 6.3(a) to the Domestic Credit Agreement for the amounts set forth on Schedule 6.3(a) to the Domestic Credit Agreement. Anything herein to the contrary notwithstanding, Capital Expenditures shall not include expenditures made as consideration or purchase price for Permitted Business Acquisitions (as defined in the Domestic Credit Agreement).

"Capital Good" shall mean a capital good (e.g., manufacturing equipment, licensing agreements) that will establish or expand foreign production capacity of an exportable good.

"<u>Capitalized Lease</u>" means, with respect to any Person, any lease of any property by that Person as lessee, the obligation for Rentals with respect to which is required to be accounted for as a capital lease on the balance sheet of such person in accordance with GAAP.

"<u>Capitalized Rentals</u>" means, as of the date of any determination, the amount at which the aggregate Rentals due and to become due under all Capitalized Leases under which the Borrower or any of its Subsidiaries is a lessee would be reflected as a liability on a consolidated balance sheet of the Borrower and its Subsidiaries.

"<u>Capital Stock</u>" means any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person (other than a corporation) including, without limitation, membership interests and partnership interests or units, and any and all warrants, rights or options to purchase any of the foregoing.

"Cash Collateral Account" shall have the meaning set forth in Section 8.7.

"Closing Date" shall mean September 29, 2006.

"Code" shall mean the Internal Revenue Code of 1986, as amended, as now or hereafter in effect, together with all regulations, rulings and interpretations thereof or thereunder issued by the Internal Revenue Service.

"Collateral" shall mean all Collateral as defined in the International Security Agreement and the International Pledge Agreement.

"Commitment" shall have the meaning set forth in Section 2.1(a).

"Compliance Certificate" shall mean a certificate executed by an Authorized Officer of the Borrower who is the chief financial officer, treasurer, assistant treasurer or controller of the Borrower, substantially in the form of Exhibit D, with such changes as acceptable to the Bank.

"Controlling Affiliate" shall have the meaning set forth in the definition of "Affiliate".

"Country Limitation Schedule" shall have the meaning set forth in the Borrower Agreement.

"Cover" means (a) a letter of credit with terms and from a bank acceptable to the Bank in its sole discretion or (b) immediately available funds to be held by the Bank in a collateral account maintained by the Bank at its principal office and collaterally assigned to the Bank as security for the Obligations using documentation reasonably satisfactory to Bank in the amount required by any applicable provision hereof. Such letter of credit or amount of immediately available funds shall be retained by Bank (and with respect to the amount of immediately available funds, in such collateral account) until such time as the applicable Letter of Credits shall have expired and reimbursement obligations, if any, with respect thereto shall have been fully satisfied; provided, however, that at such time if a Default or Event of Default has occurred and is continuing, Bank shall not be required to release such letter of credit or amount in any Cover until such Default or Event of Default shall have been cured or waived.

"Credit Accommodation" shall mean, collectively, Disbursements and issuances of Letter of Credit.

"Debt" means and includes, with respect to any Person, (i) indebtedness for borrowed money, (ii) obligations evidenced by bonds (including, without limitation, license, bid, performance, lien or payment bonds), debentures, notes or other similar instruments, (iii) obligations which have been incurred in connection with the acquisition of property or services (including, without limitation, obligations to pay the deferred purchase price of property or services), excluding trade payables and accrued expenses incurred in the ordinary course of business, (iv) obligations secured by any Lien or other charge upon property or assets owned by such Person, even though such Person has not assumed or become liable for the payment of such obligations, (v) obligations created or arising under any conditional sale or other title retention agreement with respect to property acquired by such Person, notwithstanding the fact that the rights and remedies of the seller, lender or lessor under such agreement in the event of default are limited to repossession or sale of property, (vi) the principal amount of Capitalized Rentals under any Capitalized Lease, (vii) reimbursement obligations with respect to letters of credit, and (viii) obligations under direct or indirect guaranties in respect of, and obligations (contingent or otherwise) to purchase or otherwise acquire, or otherwise to assure a creditor against loss in respect of, indebtedness or obligations of others of the kinds referred to in clauses (i) through (vii) above. For the avoidance of doubt, notwithstanding FAS 150, the Capital Stock of Holdings issued in connection with the Initial Capital Contribution (as defined in the Domestic Credit Agreement), and all other Capital Stock issued by Holdings thereafter having substantially the same terms, shall not constitute Debt (including for the purpose of calculation the covenants in Section 9.5) so long as such Capital Stock does not require any cash payments or dividends thereon or require any mandatory redemption or repurc

"<u>Debtor Laws</u>" shall mean all applicable liquidation, conservatorship, bankruptcy, moratorium, arrangement, receivership, insolvency, reorganization or other similar laws affecting creditors' rights generally, or general equitable principles, regardless of whether enforcement is sought in a proceeding at law or in equity.

- "Default" means an event which, with the lapse of time or the giving of notice, or both, would be an Event of Default.
- "Default Rate" shall mean the lower of (a) the Highest Lawful Rate, or (b) a rate per annum equal to the Prime Rate plus two percent (2%).
- "Determination Date" shall have the meaning set forth in Section 2.2(b).
- "Disbursement" shall mean an advance to fund a Drawing under a Letter of Credit issued or caused to be issued by the Bank for the account of Borrower.
- "Diyaar" shall mean Diyaar Al Muharraq Company W.L.L.
- "Diyaar Joint Venture" means a joint venture to created between Great Lakes LLC and Nass Contracting Company WLL for the purpose of carrying out the Diyaar Project.

- "Diyaar Project" shall mean the performance of dredging, reclamation and shore protection works off the coast of Al Muharraq in the Kingdom of Bahrain.
- "Diyaar Project Letter of Credit" shall mean a Letter of Credit that (i) is issued in connection with the Diyaar Project and (ii) shall have a face amount of approximately \$15 million.
 - "Dollars" and "\$" shall mean lawful currency of the United States of America.
- "<u>Domestic Credit Agreement</u>" shall mean the Credit Agreement dated as of December 22, 2003, among Holdings, the Borrower, the other Loan Parties from time to time party thereto, the financial institutions from time to time party thereto and Bank of America, N.A., as Administrative Agent and an Issuing Lender, as the same may be amended, restated, supplemented, modified, refunded, refinanced or replaced, in whole or in part, from time to time.
- "<u>Domestic Lenders</u>" shall mean Bank of America, N.A., as administrative agent, and other agents, arrangers and lenders now or hereafter party to the Domestic Credit Agreement.
- "<u>Domestic Loan Documents</u>" shall mean the Domestic Credit Agreement and the other "Loan Documents" as defined in the Domestic Credit Agreement, and all amendments, restatements, supplements, modifications, renewals, extensions, increases and rearrangements of, and substitutions for, any of the foregoing.
 - "Drawings" shall have the meaning set forth in Section 2.5(b).
 - "Durrat" shall mean Durrat Khaleej Al Bahrain Company.
- "<u>Durrat Joint Venture</u>" means the Great Lakes Nass Joint Venture created pursuant to the Joint Venture Agreement dated December 19, 2004 between Great Lakes LLC (as successor to Great Lakes Dredge & Dock Company) and Nass Contracting Company WLL.
 - "<u>Durrat Project</u>" shall mean the performance of reclamation and dredging and shore protection works at Durrat Al Bahrain.
- "EBITDA" means, with respect to any period, as determined in accordance with GAAP, the sum of the amounts for such period of Net Income, (a) <u>plus</u>, without duplication and to the extent reflected as a charge in the consolidated statement of such Net Income for such period: (i) depreciation, depletion and amortization expense, (ii) federal, state, local and foreign income taxes, (iii) Interest Expense, (iv) transaction fees and expenses incurred in connection with the Transactions (as defined in the Domestic Credit Agreement) to the extent not exceeding in the aggregate \$17,000,000, (v) non-cash charges and losses, (vi) any amounts included in the calculation of Net Income for amortization or non-cash charges for the write-off or impairment of goodwill, intangibles or other purchase accounting adjustments related to the accounting for the Transactions or other acquisitions under GAAP (including Financial Accounting Standards No. 141 and 142), (vii) fees and expenses incurred in connection with the Bonding Agreement (as defined in the Domestic Credit Agreement) and the Equipment Financing Debt (as defined in the Domestic Credit Agreement), (viii) management fees paid pursuant to any Management Agreement (as defined in the Domestic Credit Agreement) to the extent permitted to be paid

under the Domestic Credit Agreement, and (ix) Net Income attributable to the minority equity interest in NASDI (as defined in the Domestic Credit Agreement) that is not owned by the Borrower to the extent the Net Income in respect of such minority equity interest is received by the Borrower, (b) minus, without duplication, (i) non-cash gains and (ii) Net Income attributable to the minority equity interest in NASDI that is not owned, directly or indirectly, by the Borrower to the extent the Net Income in respect of such minority equity interest is distributed to the holder or holders of such minority equity interest, and (c) plus, without duplication, cash dividends received by the Borrower or any Subsidiary from Amboy Aggregates, a New Jersey joint venture, and any other equity joint ventures.

"<u>Eligible Export-Related Accounts Receivable</u>" shall mean Export-Related Accounts Receivable of the Loan Parties meeting all of the following criteria as of the date of any determination of Eligible Export-Related Accounts Receivable:

- (a) the account debtor's obligation to pay the account receivable is not conditional upon such account debtor's approval or the account receivable is not subject to any repurchase obligation or return right, other than warranty obligations in the ordinary course of business;
- (b) the account receivable shall arise from the performance by one or more of the Borrower or any Subsidiary of services which have been fully and satisfactorily performed, or from the absolute sale by one or more of the Borrower or any Subsidiary of goods (i) in which one or more of the Borrower or any of its Subsidiaries had sole and complete ownership and (ii) which have been shipped and delivered to the account debtor, which shall evidence which such obligee has possession of shipping and delivery receipts;
- (c) the account receivable shall arise in the ordinary course of business, and no notice of bankruptcy or insolvency of the account debtor (unless such account debtor's obligations are secured by a letter of credit or bond), nor any notice of such account debtor's inability to pay its debts generally as they become due, has been received by the Borrower; and
 - (d) for which the account debtor is contractually obligated to pay.

In addition, pursuant to the requirements of Ex-Im Bank, in no event shall Eligible Export-Related Accounts Receivable include any Export-Related Accounts Receivable:

- (i) that does not arise from the sale of Items in the ordinary course of the Borrower's business;
- (ii) that is not subject to a valid, perfected first priority Lien in favor of the Bank;
- (iii) as to which any covenant, representation or warranty contained in the International Loan Documents with respect to such Account Receivable has been breached;

- (iv) that is not owned by a Loan Party or is subject to any right, claim or interest of another Person other than the Lien in favor of the Bank;
- (v) with respect to which an invoice has not been sent;
- (vi) that arises from the sale of defense articles or defense services;
- (vii) that is due and payable from an account debtor located in a Prohibited Country;
- (viii) that does not comply with the requirements of the Country Limitation Schedule;
- (ix) that is due and payable more than 90 days from the earlier of the shipment date or the date of the invoice;
- (x) that is not paid within 60 calendar days from its original due date, unless it is insured through Ex-Im Bank export credit insurance for comprehensive commercial and political risk, or through Ex-Im Bank approved private insurers for comparable coverage, in which case it is not paid within 90 calendar days from its due date;
- (xi) of an account debtor for whom twenty-five percent (25%) or more of the Accounts Receivable of such account debtor do not satisfy the requirements of subclauses (ix) and (x) above;
- (xii) except with respect to Eligible Joint Ventures, that arises from a sale of goods to or performance of services for any employee of Borrower, stockholders of Borrower, Subsidiaries of Borrower, a Person with a controlling interest in Borrower or a Person which shares common controlling ownership with Borrower;
- (xiii) that is backed by a letter of credit unless the Items covered by the subject letter of credit have been shipped or the covered services provided;
 - (xiv) that Bank or Ex-Im Bank, in the exercise of reasonable judgment, deems uncollectible for any reason;
- (xv) that is due and payable in a currency other than U.S. dollars or an Approved Currency, except as may be approved in writing by Bank and Ex-Im Bank;
 - (xvi) that is due and payable from a military account debtor, except as may be approved in writing by Ex-Im Bank;
 - (xvii) that does not comply with the terms of sale set forth in Section 7 of the Loan Authorization Agreement;

(xviii) that is due and payable from an account debtor who (a) applies for, suffers, or consents to the appointment of, or the taking of possession by, a receiver, custodian, trustee or liquidator of itself or of all or a substantial part of its property or calls a meeting of its creditors, (b) admits in writing its inability, or is generally unable, to pay its debts as they become due or ceases operations of its present business, (c) makes a general assignment for the benefit of creditors, (d) commences a voluntary case under any state or federal bankruptcy laws (as now or hereafter in effect), (e) is adjudicated as bankrupt or insolvent, (f) files a petition seeking to take advantage of any other law providing for the relief of debtors, (g) acquiesces to, or fails to have dismissed, any petition which is filed against it in any involuntary case under such bankruptcy laws, or (h) takes any action for the purpose of effecting any of the foregoing;

- (xix) that arises from a bill-and-hold, guaranteed sale, sale-and-return, sale on approval, consignment or any other repurchase or return basis or is evidenced by chattel paper;
- (xx) for which the Items giving rise to such Account Receivable have not been shipped to the account debtor or when the Items are services, such services have not been performed or when the Export Order specifies a timing for invoicing the Items other than shipment or performance and the Items have not been invoiced in accordance with such terms of the Export Order, or the Accounts Receivable otherwise do not represent a final sale;
- (xxi) that is subject to any offset, deduction, defense, dispute, or counterclaim or the account debtor is also a creditor or supplier of the Borrower or the Account Receivable is contingent in any respect or for any reason;
- (xxii) for which the Borrower has made any agreement with the account debtor for any deduction therefrom, except for discounts or allowances made in the ordinary course of business for prompt payment, all of which discounts or allowances are reflected in the calculation of the face value of each respective invoice related thereto;
 - (xxiii) for which any of the Items giving rise to such Account Receivable have been returned, rejected or repossessed;
 - (xxiv) that is included as an eligible receivable under any other credit facility to which Borrower is a party;
- (xxv) any of the Items giving rise to such Accounts Receivable are Capital Goods, unless the transaction is in accordance with <u>Section 2.14</u> of the Borrower Agreement;
- (xxvi) that is due and payable from an account debtor that is, or is located in, the United States; provided, however, that this subsection (xxvi) shall not preclude an Export-Related Accounts Receivable arising from the sale of Items to foreign contractors or subcontractors providing services to a United States

Embassy or the United States Military located overseas from being deemed an Eligible Export-Related Accounts Receivable;

(xxvii) that arise from the sale of Items that do not meet the U.S. Content requirements in accordance with <u>Section 2.01(b)(ii)</u> of the Borrower Agreement;

(xxviii) that arise from the sale of any Items to be used in the construction, alteration, operation or maintenance of nuclear power, enrichment, reprocessing, research or heavy water production facilities;

(xxix) that, acting reasonably and in good faith and after consultation with the Borrower, the Bank has deemed such Accounts Receivable ineligible because of uncertainty about the creditworthiness of the account debtor or because the Bank otherwise reasonably considers the collateral value thereof to be impaired or its ability to realize such value to be insecure. Amounts payable to any account debtor shall be netted out of Eligible Export-Related Accounts Receivable owing by such account debtor. In the event of any dispute under the foregoing criteria, about whether an Account Receivable is or has ceased to be an Eligible Export-Related Account Receivable, the reasonable decision of the Bank, made in good faith, shall be conclusive and binding, absent manifest error; or

(xxx) that are Excluded Export-Related Accounts Receivable.

In addition, "Eligible Export-Related Accounts Receivable" shall also mean any account receivable of the Loan Parties that is approved in writing by the Bank and Ex-Im Bank. The Bank and Ex-Im Bank have specifically approved as Eligible Export-Related Accounts Receivable those receivables due from the Eligible Joint Ventures arising from billings to Durrat and Diyaar, provided that payments by Durrat and Diyaar to the Eligible Joint Ventures are remitted to the Cash Collateral Account in accordance with the requirements of Section 8.7. As used in this definition, "Letter of Credit" shall mean an irrevocable letter of credit subject to the UCP 500, payable in the United States or at the issuing bank and issued for the benefit of the Borrower on behalf of a Buyer in connection with the purchase of Items. The value of each Eligible Export-Related Accounts Receivable shall be, at the date of determination thereof, the aggregate face amount of such Eligible Export-Related Accounts Receivable less taxes, discounts, credits, allowances and Retainages, except to the extent otherwise permitted by the Bank in writing.

"<u>Eligible Joint Venture</u>" means, collectively, (i) the Durrat Joint Venture, (ii) the Diyaar Joint Venture, provided the Borrower delivers standing instructions to Ahli United Bank with respect to payments made to the Diyaar Joint Venture and a consent to the pledge of Great Lakes equity interests in the Diyaar Joint Venture, each in form and substance satisfactory to the Bank and (iii) any other joint venture which is approved in writing from time to time by the Ex-Im Bank and the Bank; provided, however, with respect to each joint venture set forth in clause (i), (ii) and (iii) above, the equity interests of such joint venture owned by a Borrower, a Guarantor or a Subsidiary of the Borrower or a Guarantor shall be pledged to the Bank as security for the obligations pursuant to an International Pledge Agreement.

"Environmental Laws" means all international, foreign, federal, state or local laws, statutes, common law duties, rules, regulations, ordinances and codes, together with all administrative or judicial orders, licenses, authorizations and permits of, and agreements with, any Governmental Authorities, in each case relating to environmental, health, safety and land use and natural resource matters now or hereafter in effect; including, without limitation, the Comprehensive Environmental Response, Compensation and Liability Act of 1980 ("CERCLA"), the Clean Air Act, the Federal Water Pollution Control Act of 1972, the Solid Waste Disposal Act, as amended, the Resource Conservation and Recovery Act, the Toxic Substances Control Act, and the Emergency Planning and Community Right-to-Know Act, and the Occupational Safety and Health Act, and any analogous state or local laws.

"ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended from time to time, and all rules, regulations, rulings and interpretations adopted by the Internal Revenue Service or the U.S. Department of Labor thereunder.

"Event of Default" shall mean any of the events specified in Section 10.1, provided that there has been satisfied any requirement in connection with such event for the giving of notice, or the lapse of time, or the happening of any further condition, event or act.

"Exceptions Approval Letter" means a letter from Ex-Im Bank approving any and all applicable exceptions from the rules and regulations of Ex-Im Bank governing its Working Capital Guaranty Program.

"Excluded Export-Related Accounts Receivable" shall have the meaning set forth in Section 8.7.

"Ex-Im Bank" shall mean the Export-Import Bank of the United States.

"<u>Ex-Im Bank Guaranty</u>" shall mean that certain Master Guaranty Agreement No. MN-MGA-05-001 dated as of November 1, 2005, between Ex-Im Bank and the Bank, as the same may be amended, modified or supplemented from time to time.

"<u>Ex-Im Material Adverse Effect</u>" shall mean a material adverse effect on (a) the business, assets, operations, prospects or financial or other condition of Borrower or any Guarantor, (b) the Borrower's ability to pay or perform the Obligations in accordance with the terms thereof, (c) the Collateral or Bank's Liens on the Collateral or the priority of such Lien, or (d) the Bank's rights and remedies under the International Loan Documents.

"Existing Letters of Credit" shall mean the letters of credit issued under the Domestic Credit Agreement and listed on Schedule 1.1.

"Exit Date" shall mean the date that none of Wells Fargo nor Bank nor an Affiliate thereof are participants in the Domestic Credit Agreement (i.e., no longer Domestic Lenders) unless prior to such date Ex-Im Bank has waived in writing the requirement that (i) the Borrower maintain a domestic revolving credit facility with a revolving commitment in an aggregate amount of at least \$5,000,000 and which facility is not guaranteed by Ex-Im Bank or any other governmental program or (ii) Wells Fargo, Bank, or an Affiliate thereof are participants in the Domestic Credit Agreement.

"Exit Date Event of Default" shall mean the occurrence of the event specified in Section 10.4, provided that there has been satisfied any requirement in connection with such event for the giving of notice, or the lapse of time, or the happening of any further condition, event or act.

"Export Order" shall mean a documented purchase order or contract evidencing a Buyer's agreement to purchase the Items from a Loan Party for export from the United States, which documentation shall include written information that is necessary to confirm such purchase order or contract, including identification of the Items, the name of the Buyer, the country of destination, contact information for the Buyer and the total amount of the purchase order or contract.

"Export-Related Accounts Receivable" shall mean those Accounts Receivable arising from the sale of Items which are due and payable to the Borrower or a Guarantor in the United States, except for Excluded Export-Related Accounts Receivable which shall not be required to be due and payable in the United States and as otherwise approved of by Ex-Im Bank and Bank.

"<u>Fast Track Agreement</u>" shall mean the Fast Track Lender Agreement, Number MN-FTLA-05-001 dated as of December 27, 2005, between Ex-Im Bank and the Bank, as the same may be amended, restated, modified or supplemented from time to time.

"<u>Fast Track Borrower Supplement</u>" shall mean the Fast Track Borrower Agreement Supplement executed by the Borrower as of September 29, 2006, as the same may be amended, restated, modified or supplemented from time to time.

"Final Disbursement Date" shall mean the date set forth in paragraph 10 of the Loan Authorization Agreement.

"Fiscal Quarter" means any quarter of any Fiscal Year.

"Fiscal Year" means the Fiscal Year of the Borrower consisting of a period of twelve consecutive months ending on December 31.

"GAAP" means generally accepted accounting principles set forth in the rules, regulations, statements, opinions and pronouncements of the American Institute of Certified Public Accountants and of the Financial Accounting Standards Board (or agencies with similar functions of comparable stature and authority within the accounting profession), except as provided in the definitions of "Debt" and "Interest Expense" in respect of the treatment of Capital Stock of Holdings pursuant to Statement of Financial Accounting Standards No. 150 ("FAS 150"), which, subject to Section 1.3, are applicable to the circumstances as of the date of determination.

"Governmental Approval" means any authorization, consent, approval, license or exemption of, registration or filing with, or report or notice to, any Governmental Authority.

"Governmental Authority," shall mean any foreign governmental authority, the United States of America, any State of the United States of America and any political subdivision of any of the foregoing, and any agency, department, commission, board, bureau, court or other tribunal

having jurisdiction over the Bank, any Guarantor or the Borrower, or any of their respective assets or Property.

"Governmental Requirement" means any law, statute, code, ordinance, order, rule, regulation, judgment, decree, injunction, writ, edict, franchise, permit, certificate, license, award, authorization or other direction, guideline, or requirement of any Governmental Authority.

"Great Lakes LLC" means Great Lakes Dredge & Dock Company, LLC, a Delaware limited liability company, and any successors and assigns.

"Guaranties" by any Person shall mean all obligations (other than endorsements in the ordinary course of business of negotiable instruments for deposit or collection) of such Person guaranteeing or in effect guaranteeing any Debt, dividend or other obligation, of any other Person (the "Primary Obligor") in any manner, including, without limitation, all obligations incurred through an agreement, contingent or otherwise, by such Person: (i) to purchase such Debt or obligation or any property or assets constituting security therefor, (ii) to advance or supply funds (x) for the purchase or payment of such Debt or obligation or (y) to maintain working capital or other balance sheet condition or otherwise to advance or make available funds for the purchase or payment of such Debt or obligation, or (iii) to lease property or to purchase Securities or other property or services primarily for the purpose of assuring the owner of such Debt or obligation of the ability of the Primary Obligor to make payment of the Debt or obligation, or (iv) otherwise to assure the owner of the Debt or obligation of the Primary Obligor against loss in respect thereof. For the purposes of all computations made under this Agreement, a Guaranty in respect of any Debt for borrowed money shall be deemed to be Debt equal to the principal amount of such Debt for borrowed money which has been guaranteed (or the aggregate amount of such Debt which is guaranteed under such Guaranty, whichever is less), and a Guaranty in respect of any other obligation or liability or any dividend shall be deemed to be Debt equal to the maximum aggregate amount of such obligation, liability or dividend so guaranteed. Guaranties shall not include reimbursement obligations with respect to letters of credit but shall include guaranties of reimbursement obligations with respect to such letters of credit.

"Guaranty" shall mean each Guaranty executed by the Guarantors, substantially in the form of Exhibit F.

"Guarantors" means Great Lakes LLC, and each Subsidiary of the Borrower which executes and delivers a Guaranty and an International Security Agreement.

"<u>Highest Lawful Rate</u>" shall mean, with respect to the Bank, the maximum nonusurious interest rate that at any time or from time to time may be contracted for, taken, reserved, charged, or received with respect to the Obligations or on other amounts due to the Bank pursuant to this Agreement or any other International Loan Document, under laws applicable to the Bank which are presently in effect, or, to the extent allowed by law, under such applicable laws which may hereafter be in effect and which allow a higher maximum nonusurious interest rate than applicable laws now allow.

- "Holdings" means GLDD Acquisitions Corp., a Delaware corporation, and any successors and assigns.
- "Indemnified Parties" shall have the meaning set forth in Section 11.12.
- "Initial Date" shall have the meaning set forth in Section 7.5.
- "Interest Coverage Ratio" have the meaning specified in Section 9.5(d).

"Interest Expense" means, for any Fiscal Quarter, the aggregate consolidated interest expense (net of interest income) of the Borrower and its consolidated Subsidiaries for such Fiscal Quarter, as determined in accordance with GAAP, including (i) Commitment Fees (as defined in the Domestic Credit Agreement) paid or payable during such Fiscal Quarter, (ii) all other fees paid or payable with respect to the issuance or maintenance of any Guaranty or contingent Debt (including Letters of Credit (as defined in the Domestic Credit Agreement) but excluding fees paid under the Bonding Agreement (as defined in the Domestic Credit Agreement)), which, in accordance with GAAP, would be included as interest expense, (iii) net costs or benefits under any Rate Protection Agreement (excluding (A) any gain or loss recognized under GAAP resulting from the mark to market valuation of any Rate Protection Agreement and (B) the costs of any commodity hedging transaction or foreign currency hedging transaction) and (iv) the portion of any payments made in respect of Capitalized Rentals of the Borrower and its consolidated Subsidiaries allocable to interest expense, but excluding any amortization of costs and expenses incurred in connection with, and relating to, this Agreement or other financings permitted by this Agreement. For the avoidance of doubt, "Interest Expense" shall not include any non-cash dividends or other non-cash payments in respect of any Capital Stock of Holdings that is not included in the definition of "Debt" pursuant to the last sentence of such definition.

"<u>International Borrowing Base</u>" shall mean an amount equal to 90% of the Eligible Export-Related Accounts Receivable of the Loan Parties which is supported by a firm Export Order to which no party is in breach in any material respect of any material provision thereof.

"International Loan Documents" shall mean this Agreement, the Note, the Guaranty, the Standby Letter of Credit Agreement, all International Security Documents, and all instruments, certificates and agreements now or hereafter executed or delivered to the Bank pursuant to any of the foregoing and the transactions connected therewith, and all amendments, restatements, supplements, modifications, renewals, extensions, increases and rearrangements of, and substitutions for, any of the foregoing; provided that "International Loan Documents" shall not include the Domestic Loan Documents, the Ex-Im Bank Guaranty, the Fast Track Agreement, the Fast Track Borrower Agreement Supplement, the Loan Authorization Agreement and the Borrower Agreement.

"<u>International Security Agreements</u>" shall mean the International Security Agreements at any time executed and delivered by the Borrower and any Guarantor for the benefit of the Bank, and any and all amendments, modifications, renewals and extensions thereof.

"International Security Documents" shall mean this Agreement, the International Security Agreements, the International Pledge Agreements and any guaranties, as each may be amended or modified from time to time.

"International Pledge Agreements" shall mean the International Pledge Agreements at any time executed and delivered by the Borrower and any Guarantor for the benefit of the Bank, and any and all amendments, modifications, renewals and extensions thereof.

"Inventory" shall have the meaning set forth in the Borrower Agreement.

"Investments" means, as applied to any Person, any purchase or other acquisition by that Person of Securities or Debt, or of a beneficial interest in Securities or Debt, of any other Person, any loan, advance (other than deposits with financial institutions available for withdrawal on demand, prepaid expenses, advances to employees, officers and directors and similar items made or incurred in the ordinary course of business), capital contribution by that Person to any other Person, and all other items that are or would be classified as investments on a balance sheet prepared in accordance with GAAP. The amount of any Investment shall be the original principal or capital amount thereof less all returns of principal or equity thereon.

"Items" shall have the meaning set forth in the Borrower Agreement.

"Letter of Credit" and "Letters of Credit" shall have the meanings provided in Section 2.3(a).

"Letter of Credit Fee" shall have the meaning specified in Section 4.2.

"<u>Letter of Credit Obligations</u>" shall mean all undrawn amounts of outstanding obligations incurred by the Bank, whether direct or indirect, contingent or otherwise, due or not due, in connection with the issuance or guarantee by the Bank of Letters of Credit.

"<u>Letter of Credit Outstandings</u>" shall mean, at any time, the sum of, without duplication, (a) the aggregate Stated Amount of all outstanding Letters of Credit and (b) the amount of all Unpaid Drawings in respect of all Letters of Credit.

"Letter of Credit Request" shall have the meaning specified in Section 2.4(a).

"Lien" means any mortgage, deed of trust, pledge, hypothecation, assignment, deposit arrangement, security interest, encumbrance for the payment of money, lien (statutory or other), preference, priority or other security agreement or preferential arrangement of any kind or nature whatsoever, including, without limitation, any conditional sale or other title retention agreement, the interest of a lessor under a capital lease, any financial lease having substantially the same economic effect as any of the foregoing and the filing of any financing statement (other than a financing statement filed by a "true" lessor pursuant to Section 9-408 of the Code or other comparable law of any jurisdiction) naming the owner of the asset to which such Lien relates as debtor under the Code or other comparable law of any jurisdiction but excluding unauthorized financing statements filed under the UCC (or any similar law) with respect to which no security interest exists in the assets described in such financing statements.

"Loan Authorization Agreement" shall mean the duly-executed Fast Track Loan Authorization Agreement, setting forth certain terms and conditions of the credit facility, a copy of which is attached as Annex A to the Borrower Agreement.

"Loan Facility Anniversary Date" shall mean each one (1) year anniversary of September 29, 2006.

"Loan Party" means the Borrower and each Guarantor.

"Material Adverse Effect" means (a) a material adverse effect upon (i) the condition (financial or otherwise), operating results, assets, liabilities, business or operations of the Borrower and its Subsidiaries, taken as a whole, or (ii) the validity or enforceability of any of the International Loan Documents, the Liens granted to the Bank pursuant to the International Loan Documents, or the rights and remedies of the Bank thereunder.

"Maturity Date" shall mean December 31, 2008.

"<u>Net Income</u>" means, for any period, the aggregate of all amounts (exclusive of all amounts in respect of any extraordinary or non-recurring gain or loss) which, in accordance with GAAP, would be included as net income on a consolidated statement of income of the Borrower and its Subsidiaries for such period.

"Note" shall mean the International Revolving Note executed by the Borrower, as of September 29, 2006, and payable to the Bank, in the original principal amount of Twenty Million and No/100 Dollars (\$20,000,000.00) and each renewal, increase, extension, amendment, replacement, modification or other rearrangement thereof.

"Obligations" shall mean all advances to, and debts, liabilities, obligations, covenants and duties of, any, some or all of the Loan Parties arising under this Agreement, the Note and the other International Loan Documents whether now or hereafter arising, of any nature or amount whatsoever, and including any and all interest and fees that accrue after the commencement of proceedings under Debtor Laws, regardless of whether the same are allowed claims in such proceeding.

"OFAC" has the meaning specified in Section 11.17.

"Organization Document" means, (a) with respect to any corporation, the certificate or articles of incorporation and the bylaws (or equivalent or comparable constitutive documents with respect to any non-U.S. jurisdiction); (b) with respect to any limited liability company, the certificate or articles of formation or organization and operating agreement; and (c) with respect to any partnership, joint venture, trust or other form of business entity, the partnership, joint venture or other applicable agreement of formation or organization and any agreement, instrument, filing or notice with respect thereto filed in connection with its formation or organization with the applicable Governmental Authority in the jurisdiction of its formation or organization and, if applicable, any certificate or articles of formation or organization of such entity.

"Original Borrower" shall mean Great Lakes Dredge & Dock Corporation, a Delaware corporation.

"Payment Office" means 1000 Louisiana, 4th Floor, Houston, Texas 77002, Attention: Scott Gildea, or such other office or offices as the Bank may from time to time notify the Borrower.

"Permitted Liens" shall mean (a) Liens for taxes, assessments or other governmental charges or levies not delinquent, or, being contested in good faith and by appropriate proceedings and with respect to which proper reserves have been taken by the Borrower; provided, that, the Lien shall have no effect on the priority of the Liens in favor of the Bank or the value of the assets in which the Bank has such a Lien and a stay of enforcement of any such Lien shall be in effect; (b) deposits or pledges securing obligations under worker's compensation, unemployment insurance, social security or public liability laws or similar legislation; (c) deposits or pledges securing bids, tenders, contracts (other than contracts for the payment of money), leases, statutory obligations, surety and appeal bonds and other obligations of like nature arising in the ordinary course of the Borrower's business; (d) judgment Liens that have been stayed or bonded; (e) mechanics', workers', materialmen's or other like Liens arising in the ordinary course of the Borrower's business with respect to obligations which are not due; (f) Liens placed upon fixed assets hereafter acquired to secure a portion of the purchase price thereof, provided, that, any such Lien shall not encumber any other property of Borrower; (g) security interests being terminated concurrently with the execution of the International Loan Documents; and (h) Liens disclosed in Section 6.D. of the Loan Authorization Agreement, provided that, except as otherwise permitted by Ex-Im Bank in writing, such Liens in Section 6.D. shall be subordinate to the Liens in favor of the Bank on the Collateral.

"Person" shall mean an individual, partnership, joint venture, corporation, joint stock company, bank, trust, unincorporated organization and/or a government or any department or agency thereof.

"Plan" shall mean any plan subject to Title IV of ERISA or Section 412 of the Code and maintained at any time since January 1, 1986 for employees of the Borrower or any Subsidiary of the Borrower thereof or of any member of a "controlled group of corporations" or "trade or business," as such terms are defined in Section 414(b) or (c) of the Code, of which the Borrower or any Subsidiary of the Borrower thereof is a member, or any plan subject to Title IV of ERISA or Section 412 of the Code to which the Borrower or any Subsidiary of the Borrower thereof is required to contribute, or has been required to contribute at any time since January 1, 1986, on behalf of its employees.

"Prime Rate" means the rate of interest from time to time announced publicly by Wells Fargo, in San Francisco, California, as its prime rate. Such rate is set by Wells Fargo as a general reference rate of interest, taking into account such factors as Wells Fargo may deem appropriate, it being understood that many of Wells Fargo's commercial or other loans are priced in relation to such rate, that it is not necessarily the lowest or best rate actually charged to any customer and that Wells Fargo may make various commercial or other loans at rates of interest having no relationship to such rate. In addition, such rate is evidenced by the recording thereof after its announcement in such internal publication or publications as Wells Fargo may designate, and each change in the Prime Rate will be effective on the day the change is announced within Wells Fargo; provided, however, such rate shall be a rate of interest generally applied by Wells

Fargo to other loan transactions to the extent such transactions include rates based in whole or in part on the Prime Rate.

"Prohibited Country" shall mean any country in which Ex-Im Bank coverage is not available for commercial reasons or in which Ex-Im Bank is legally prohibited from doing business, as designated in the Country Limitation Schedule.

"Property" or "Properties" means any interest in any kind of property or assets, whether real, personal or mixed, and whether tangible or intangible.

"Quarterly Unaudited Financial Statements" shall mean the unaudited quarterly consolidated financial statements of a Person, which statements shall include a consolidated and consolidating balance sheet as of the end of such Fiscal Quarter and a consolidated and consolidating statement of earnings and consolidated statement of cash flows for such Fiscal Quarter, and for the Fiscal Year to date, subject to normal year-end adjustments, and certified on behalf of such Person by an Authorized Officer of such Person who is the chief financial officer, treasurer, assistant treasurer or controller of the Borrower

"Rate Protection Agreement" means any interest rate hedging transaction, commodity hedging transaction, foreign currency hedging transaction or similar arrangement.

"Rentals" means and includes as of the date of any determination thereof all fixed rents (including as such all payments which the lessee is obligated to make to the lessor on termination of the lease or surrender of the property and including all payments on Capital Leases) payable by the Borrower or a Subsidiary, as lessee or sublessee under a lease of real or personal property, but shall be exclusive of any amounts required to be paid by the Borrower or a Subsidiary (whether designated as rents or additional rents) on account of maintenance, repairs, insurance, taxes and similar charges. Fixed rents under any so-called "percentage leases" shall be computed not only on the basis of the minimum rents, if any, required to be paid by the lessee but also on the basis of any additional rents whether based on sales volume or gross revenues or otherwise. With respect to leases providing for period of free rent or discounted rent, Rentals means the amount of the actual cash payments required under the lease, even though accounting convention may require that the rents be accrued on an amortized basis over the term of the lease.

"Restricted Payments" means (i) any dividend or other distribution on account of any shares of any class of Capital Stock of the Borrower (including, without limitation, any class of preferred stock) now or hereafter outstanding (except a dividend payable solely in shares or any warrants, options or other rights with respect thereto or rights to acquire shares, of common stock of the Borrower), including, without limitation, all payments which are from time to time due and owing by the Borrower pursuant to or with respect to the Acquisition Agreement (as defined in the Domestic Credit Agreement) (other than for indemnification or expense reimbursement pursuant to the terms thereof), (ii) any redemption, retirement, repurchase, sinking fund or similar payment, purchase or other acquisition for value of any shares of any class of Capital Stock of the Borrower now or hereafter outstanding or any warrants, options or other rights with respect thereto, (iii) any voluntary or mandatory redemption, repurchase, retirement, sinking fund payment or other payment of principal with respect to the Note Indenture Obligations (as

defined in the Domestic Credit Agreement), or any voluntary payment or other prepayment of interest with respect to the Note Indenture Obligations, (iv) any payment made to redeem, purchase, repurchase or retire, or to obtain the surrender of any outstanding warrants, options or other rights to acquire shares of any class of Capital Stock of the Borrower, (v) any voluntary prepayment, redemption, or repurchase or other voluntary payment of principal with respect to the Bonding Agreement, or any voluntary payment or other prepayment of interest with respect thereto, or (vi) the setting aside of funds for any of the foregoing.

"<u>Retainage</u>" shall mean that portion of the purchase price of an Export Order that a Buyer is not obligated to pay until the end of a specified period of time following the satisfactory performance under such Export Order.

"Security" has the meaning specified in Section 2(1) of the 1933 Act.

"Senior Debt" means Total Funded Debt consisting of (i) the outstanding principal balance of the Obligations (as defined in the Domestic Credit Agreement) (other than the aggregate undrawn face amount of Letters of Credit (as defined in the Domestic Credit Agreement)), (ii) Capitalized Rentals and (iii) and all other Total Funded Debt owing by the Borrower or any of its Subsidiaries which is secured in whole or in part by a Lien on any property of the Borrower or any of its Subsidiaries. "Senior Debt" shall be calculated net of cash and Cash Equivalents (as defined in the Domestic Credit Agreement) held by the Borrower and its Subsidiaries in the United States.

"Senior Leverage Ratio" have the meaning specified in Section 9.5(c).

"Standby Letter of Credit Agreement" shall mean the Standby Letter of Credit Agreement, executed by the Borrower substantially in the form of Exhibit G.

"<u>Stated Amount</u>" shall mean, as to each Letter of Credit, at any time, the maximum amount then available to be drawn thereunder (without regard to whether any conditions to drawing could then be met).

"Subsidiary" of any Person means any corporation, partnership, limited liability company or other association or entity of which more than fifty percent (50%) of the Voting Stock of such entity is at any time, directly or indirectly, owned by such Person. Unless the context is to the contrary, any reference to a Subsidiary herein shall mean a Subsidiary of the Borrower.

"Successor Borrower" shall mean Great Lakes Dredge & Dock Corporation (formerly named Great Lakes Dredge & Dock Holdings Corp.), a Delaware corporation and successor by merger to the Original Borrower.

"Total Funded Debt" of any Person means as of any date of determination, all Debt of the Borrower and its consolidated Subsidiaries which, in accordance with GAAP, should be included as liabilities in the consolidated balance sheet of the Borrower and its Subsidiaries at such time (excluding, however, the undrawn face amount of all letters of credit and all Capitalized Rentals due within one year from the date of determination hereunder). "Total Funded Debt," when used with respect to the Borrower, shall mean the aggregate amount of all such Total Funded Debt of the Borrower and its Subsidiaries determined on a consolidated basis in accordance with GAAP

(eliminating intercompany items). "Total Funded Debt" shall be calculated net of cash and Cash Equivalents (as defined in the Domestic Credit Agreement) held by the Borrower and its Subsidiaries in the United States.

- "Total Leverage Ratio" have the meaning specified in Section 9.5(b).
- "UCC" shall mean the Uniform Commercial Code as in effect from time to time in the State of New York.
- "Unpaid Drawing" shall have the meaning specified in Section 2.5(a).
- "<u>USA Patriot Act</u>" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001 (Title III of Pub. L. 107-56, signed into law October 26, 2001).
- "<u>U.S. Content</u>" shall have the meaning set forth in the Borrower Agreement. The parties hereto hereby agree that in the event any dispute arises as to the U.S. Content of any good or service, the decision of Ex-Im Bank shall be final and conclusive.
- "<u>Voting Stock</u>" means Securities or other equity interests of any class or classes of a corporation, partnership, limited liability company or other association or entity the holders of which are ordinarily, in the absence of contingencies, entitled to elect a majority of the board of directors, managers, general partners, managing members or Persons performing similar functions.
 - "Warranty Letter of Credit" shall have the meaning set forth in the Borrower Agreement.
 - "Weekly International Borrowing Base Certificate" shall have the meaning set forth in Section 2.2(a).
 - "Wells Fargo" shall mean Wells Fargo Bank, N.A.

EXHIBIT "B"

INTERNATIONAL REVOLVING NOTE

See attached.

Attachment to Exhibit B

INTERNATIONAL REVOLVING NOTE

\$20,000,000.00 September 29, 2006

FOR VALUE RECEIVED, the undersigned, GREAT LAKES DREDGE & DOCK CORPORATION, a Delaware corporation (the "Borrower"), HEREBY PROMISES TO PAY to the order of WELLS FARGO HSBC TRADE BANK, N.A. (the "Bank"), on or before December 31, 2008 (the "Maturity Date"), the principal sum of Twenty Million and 00/100 Dollars (\$20,000,000.00), or, if less, the aggregate principal amount of the Obligations outstanding on the Maturity Date, in accordance with the terms and provisions of that certain Letter of Credit Agreement dated as of September 29, 2006 by and between the Borrower and the Bank (as amended from time to time, the "Agreement"; capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to such terms in the Agreement).

The outstanding principal balance of this Note shall be due and payable as provided in the Agreement. The Borrower promises to pay interest on the outstanding principal amount of Obligations from and including the date of any Unpaid Drawing evidenced by this Note to but not including the Maturity Date at the rate or rates, and shall be due and payable on the dates, set forth in the Agreement. Interest shall accrue and be payable at the Default Rate when provided for in the Agreement.

Payments of principal and interest, and all amounts due with respect to costs and expenses, shall be made in lawful money of the United States of America in immediately available funds, without deduction, set-off or counterclaim to the Bank in accordance with Section 3.3 of the Agreement, on the dates on which such payments shall become due pursuant to the terms and provisions set forth in the Agreement.

If any payment of principal or interest on this Note shall become due on a Saturday, Sunday, or public holiday on which the Bank is not open for business, such payment shall be made on the next succeeding Business Day and such extension of time shall in such case be included in computing interest in connection with such payment.

All payments and prepayments made hereon shall be recorded in the holder's records and such records shall be controlling, absent manifest error.

In addition to all principal and accrued interest on this Note, the Borrower agrees to pay (a) all reasonable out-of-pocket costs and expenses incurred by any owners and holders of this Note in collecting this Note through any probate, reorganization, bankruptcy or any other proceeding and (b) reasonable attorneys' fees when and if this Note is placed in the hands of an attorney for collection after default.

All agreements between the Borrower, any Guarantor and the Bank, whether now existing or hereafter arising and whether written or oral, are hereby expressly limited so that in no contingency or event whatsoever, whether by reason of demand being made on this Note or

otherwise, shall the amount paid, or agreed to be paid, to the Bank for the use, forbearance, or detention of the money to be loaned under the Agreement and evidenced by this Note or otherwise or for the payment or performance of any covenant or obligation contained in the Agreement, this Note or in any other International Loan Document exceed the Highest Lawful Rate. If, as a result of any circumstances whatsoever, fulfillment of any provision hereof or of any of such documents, at the time performance of such provision shall be due, shall involve transcending the limit of validity prescribed by applicable usury law, then, ipso facto, the obligation to be fulfilled shall be reduced to the limit of such validity, and if, from any such circumstance, the Bank shall ever receive interest or anything which might be deemed interest under applicable law which would exceed the Highest Lawful Rate, such amount which would be excessive interest shall be applied to the reduction of the principal amount owing on account of the Obligations or the amounts owing on other obligations of the Borrower and any Guarantor to the Bank under any International Loan Document and not to the payment of interest, or if such excessive interest exceeds the unpaid principal balance of the Obligations and the amounts owing on other obligations of the Borrower and any Guarantor the Bank under any International Loan Documents, as the case may be, such excess shall be refunded to the Borrower or such Guarantor, as applicable. All sums paid or agreed to be paid to the Bank for the use, forbearance, or detention of the indebtedness of the Borrower and the Guarantors to the Bank shall, to the extent permitted by applicable law, be amortized, prorated, allocated, and spread throughout the full term of such indebtedness until payment in full of the principal thereof (including the period of any renewal or extension thereof) so that the interest on account of such indebtedness shall not exceed the Highest Lawful Rate. Notwithstanding anything to the contrary contained in the Agreement or this Note, it is understood and agreed that if at any time the rate of interest which accrues on the outstanding Obligations shall exceed the Highest Lawful Rate, the rate of interest which accrues on the outstanding Obligations shall be limited to the Highest Lawful Rate, but any subsequent reductions in the rate of interest which accrues on the outstanding Obligations shall not reduce the rate of interest which accrues on the outstanding Obligations below the Highest Lawful Rate until the total amount of interest accrued on the outstanding Obligations equals the amount of interest which would have accrued if such interest rate had at all times been in effect. The terms and provisions of this paragraph shall control and supersede every other provision of all agreements between the Borrower, the Guarantors and the Bank.

This Note is entitled to the benefits of the Agreement, which Agreement, among other things, contains provisions for acceleration of the maturity hereof upon the happening of certain stated events, for prepayments on account of principal hereof prior to the maturity hereof upon the terms and conditions and with the effect therein specified, and provisions to the effect that no provision of the Agreement or this Note shall require the payment or permit the collection of interest in excess of the Highest Lawful Rate. The obligations of the Borrower hereunder are secured by the International Security Documents. It is contemplated that by reason of prepayments or repayments hereon prior to the Maturity Date, there may be times when no indebtedness is owing hereunder prior to such date, but notwithstanding such occurrences, this Note shall remain valid and shall be in full force and effect as to reimbursement obligations in respect of Letters of Credit issued pursuant to the Agreement subsequent to each such issuance.

Except as otherwise specifically provided for in the Agreement, the Borrower and any and all endorsers, guarantors and sureties severally waive grace, demand, presentment for

payment, notice of dishonor or default, protest, notice of protest, notice of intent to accelerate, notice of acceleration and diligence in collecting and bringing of suit against any party hereto, and agree to all renewals, extensions or partial payments hereon, with or without notice, before or after maturity.

THIS NOTE, THE OTHER INTERNATIONAL LOAN DOCUMENTS (UNLESS ANY SUCH INTERNATIONAL LOAN DOCUMENT EXPRESSLY PROVIDES OTHERWISE) SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO ITS CONFLICTS OF LAW PRINCIPLES (OTHER THAN PROVISIONS OF 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW).

THIS NOTE AND THE OTHER INTERNATIONAL LOAN DOCUMENTS REPRESENT THE FINAL AGREEMENT AMONG THE PARTIES AS TO THE SUBJECT MATTER HEREOF AND THEREOF AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES; AND THERE ARE NO UNWRITTEN ORAL AGREEMENTS AMONG THE PARTIES.

IN WITNESS WHEREOF, the Borrower has caused this Note to be executed and delivered by its officer thereunto duly authorized effective as	of the date
first above written.	

GREAT LAKES DREDGE & DOCK CORPORATION

By:					
	Name:				
	Title:				

EXHIBIT "C"

WEEKLY INTERNATIONAL BORROWING BASE CERTIFICATE

See attached.

Attachment to Exhibit C

BORROWING BASE CERTIFICATE FOR EX-IM BANK GUARANTEED LINE

Scott Gildea, Wells Fargo HSBC Trade Bank, Houston

Tel: 713-319-1374 Fax: 713-739-1086 gildeas@wellsfargo.com

To: Brad Young, Wells Fargo HSBC Trade Bank, San Francisco

Tel: 415.396.6166 Fax: 415.975.6558 youngbk@wellsfargo.com

Enter Date Below

Column C

As of Date ("Reporting Date"):

XXXX

0

Borrower: Great Lakes Dredge & Dock Corporation

In accordance with the International Letter of Credit Agreement dated as of September 29, 2006 (the "Credit Agreement"), set forth below is the calculation of the Borrowing Base and Availability as of the date shown above (the "Reporting Date"). All terms used in this certificate have the meanings given to them in the Credit Agreement. Unless otherwise indicated, all amounts are as of the Reporting Date.

On the fourth week of each calendar month, Borrower is to submit with this Borrowing Base Certificate (i) a monthly Export-Related Accounts Receivable aging report and (ii) a monthly listing of Export Orders per tab below in addition to copies of 10% of the number of invoices as well as export orders, which must exceed 10% of the total dollar balance of the corresponding export-related A/R and export orders. These copies should be faxed or e-mailed in PDF format to Brad Young at 415.975.6558 or youngbk@wellsfargo.com.

Column A Column B Eligible Export-Related Accounts Receivable: arise from the export of goods or services from the United States and which must be due and collectible in the United States, except as provided in the Credit Agreement for receivables due from the Eligible Joint Ventures. The Bank and Ex-Im Bank have specifically approved as Eligible Export-Related Accounts Receivables those receivables due from the Eligible Joint Ventures arising from billings to Durrat Khaleej Al Bahrain Company ("Durrat Khaleej") and Diyaar Al Muharraq Company WLL ("Diyaar Al Muharraq"), provided that payments

by Durrat Khaleej and Diyaar Al Muharraq to the Eligible Joint Ventures are remitted within one (1) Business Day to the Cash Collateral Account. The foreign content (i.e. local costs) of these export receivables should be shown as ineligible under 2.xvii. Note: foreign A/R, which are not generated from the export of U.S. goods or services, are not eligible for financing under the Credit Agreement.

A. Beginning Gross Export A/R from prior week.
a. Add Export Sales

- Less Export Payments received (enter positive figure; formula subtracts)
- Less Credit memos or other adjustments (enter positive figure; formula subtracts)

Total Control Balance: A + a.-b.-c.

G/L Balance

A/R Aging Balance

For information purposes:

AA. Projected U.S. content percentage for the Durrat Khaleej project.

AB. Projected U.S. content percentage for the Diyaar Al Muharraq project.

- 1. Export A/R Beginning Balance (Use lower of Control, G/L or A/R Aging balance). Export A/R are net of any and all rights of offset or counterclaim, including
 - 1.a Included in the total on line 1 are receivables due from Eligible Joint Ventures arising from billings on the Durrat Khaleej project.
 - 1.b Included in the total on line 1 are receivables due from Eligible Joint Ventures arising from billings on the Diyaar Al Muharraq project.
- 2. Ineligible Export-Related Accounts Receivable
 - (i) that does not arise from the sale of Items in the ordinary course of the Borrower's business;
 - (ii) that is not subject to a valid, perfected first priority Lien in favor of the Lender or which are subject to any lien, security interest or claim in favor of any Person other than the Lender:
 - (iii) as to which any covenant, representation or warranty contained in the Loan Documents with respect to such Account Receivable has been breached;
 - (iv) that is not owned by the Borrower or is subject to any right, claim or interest of another Person other than the Liens in favor of the Lender;
 - (v) with respect to which an invoice has not been sent;
 - (vi) that arises from the sale of defense articles or defense services;
 - (vii) that is due and payable from a Buyer located in a country with which Ex-Im Bank is prohibited from doing business as designated in the Country Limitation Schedule ("CLS") which can be found on Ex-Im's website at www.exim.gov under "Apply". Those countries with an "X" for financing under one year are ineligible;
 - (viii) that does not comply with the requirements of the Country Limitation Schedule ("CLS"). Some countries which are open for short-term Ex-Im support have restrictions outlined in the footnotes which may make these receivables and corresponding inventory ineligible;
 - (ix) that is due and payable more than ninety (90) calendar days from invoice date;

 - (x) that is more than sixty (60) calendar days past its original due date; (xi) that account where the amount owed under (ix) or (x) above exceeds 25% of the total amount owed, then the entire account is ineligible; (xii) as to which any covenant, representation or warranty contained in the International Loan Documents with respect to such Accounts receivable has been breached:
 - (xiii) that arises from a sale of goods to or performance of services for an employee of the Borrower, a stockholder of the Borrower, a subsidiary of the Borrower, a Person with a controlling interest in the Borrower or a Person which shares common controlling ownership with the Borrower (does not apply to receivables due from the JV arising from billings for the Durrat Khaleej and Diyaar Al Muharraq projects);
 - (xiv) that is backed by a letter of credit unless the Items covered by the subject letter of credit have been shipped;

 - (xv) that the Lender or Ex-Im Bank, in its reasonable judgment, deems uncollectible for any reason; (xvi) that is due and payable in a currency other than Dollars or an Approved Currency, except as may be approved in writing by Ex-Im Bank (receivables in Bahrain Dinars have been approved by Ex-Im Bank);
 - (xvii) that is due and payable from a military Buyer, except as may be approved in writing by Ex-Im Bank; (xviii) that does not comply with the terms of sale set forth in Section 7 of the Loan Authorization Agreement;

 - (xix) that is due and payable from a Buyer who (A) applies for, suffers, or consents to the appointment of, or the taking of possession by, a receiver, custodian, trustee or liquidator of itself or of all or a substantial part of its property or calls a meeting of its creditors, (B) admits in writing its inability, or is generally unable, to pay its debts as they become due or ceases operations of its present business, (C) makes a general assignment for the benefit of creditors, (D) commences a voluntary case under any state or federal bankruptcy laws (as now or hereafter in effect), (E) is adjudicated as bankrupt or insolvent, (F) files a petition seeking to take advantage of any other law providing for the relief of debtors, (G) acquiesces to, or fails to have dismissed, any petition which is filed against it in any involuntary case under such bankruptcy laws, or

 - (H) takes any action for the purpose of effecting any of the foregoing; (xx) that arises from a bill and hold, guaranteed sale, sale and return, sale on approval, consignment or any other repurchase or return basis or is evidenced by chattel paper;
 - (xxi) for which the Items giving rise to such Account Receivable have not been shipped to the Buyer or when the Items are services, such services have not been performed (retainage), or when the Export Order specifies a timing for invoicing the Items other than shipment or performance and the Items have not been invoiced in accordance with such terms of the Export Order, or the Accounts Receivable otherwise do not represent a final sale;

(xxii) that is subject to any offset, deduction, defense, dispute, or counterclaim or the Buyer is also a creditor or supplier of the Borrower or the Account Receivable is contingent in any respect or for any reason; (xxiii) for which the Borrower has made any agreement with the Buyer for any deduction therefrom, except for discounts or allowances made in the ordinary course of business for prompt payment, all of which discounts or allowances are reflected in the calculation of the face value of each respective invoice related thereto; (xxiv) for which any of the Items giving rise to such Account Receivable have been returned, rejected or repossessed; (xxv) to the extent it includes any finance charges, service charges, taxes, discounts, credits, allowances and Retainages; (xxvi) that arises from the sale of Items that do not meet the U.S. Content requirements in accordance with Section 2.01(b)(ii) of the Borrower Agreement; (xxvii) that arise from the sale of Items containing any Foreign Content not incorporated into such Items in the US; (xxxiii) that arise from the sale of any Items to be used in the construction, alteration, operation or maintenance of nuclear power, enrichment, reprocessing, research or heavy water production facilities; or (xxix)that is included as an eligible receivable under any other credit facility to which Borrower is a party; (xxx) that is due and payable from a Buyer that is, or is located in, the United States; provided however, that this subsection shall not preclude an Export-Related Accounts Receivable arising from the sale of Items to foreign contractors or subcontractors providing services to a United States Embassy or the United States Military located overseas from being deemed an Eligible Export-Related Accounts Receivable; (xxxi) that, acting reasonably and in good faith and after consultation with the Borrower, the Bank has deemed such Accounts ineligible because of uncertainty about the creditworthiness of the account debtor or because the Bank otherwise reasonably considers the collateral value thereof to be impaired or its ability to realize such value to be insecure. Accounts payable to any account debtor shall be netted out of Eligible Export-Related Accounts Receivable owing by such account debtor. In the event of any dispute under the foregoing criteria, about whether an Account Receivable is or has ceased to be an Eligible Export-Related Account Receivable, the reasonable decision of the Bank, made in good faith, shall be conclusive and binding, absent manifest error; (xxx) that are Excluded Export-Related Accounts Receivable. 2. (xxxii.) Total A/R ineligibles \$ 0 3. Total Eligible Export-Related Accounts Receivable (line 1 less line 2xxxii.) Standby Letters of Credit: Except with the written consent of the Bank, and other than with respect to the Diyaar Project Letter of Credit, the maturity date of standby letters of credit shall not exceed 12 months from the date of issuance. The Diyaar project Letter of Credit shall have a maturity date up the lesser of 27 months or the term of the international letter of credit facility. The standby letters of credit may be used for bid, advance payment, or performance purposes. However, the standbys may not be used for warranty purposes. If a standby L/C is issued for a combination of performance and warranty purposes, the L/C may be booked under this line during the performance period, but not for the warranty period. Standby L/Cs may only be issued for projects where the export-related A/R will be eligible for financing under the Ex-Im Bank Working Capital Guarantee program. XXXX B. Standby L/C Details 4. At issuance, Amount of First Standby. 4.a Beneficiary of Standby and Country 4 h Date Issued 4.c Expiration Date 4.d L/C Reference Number Assigned by the Bank 4.e. Purpose of standby5. At issuance, Amount of Second Standby 5.a Beneficiary of Standby and Country 5.b. Date Issued 5.c Expiration Date 5.d L/C Reference Number Assigned by the Bank 5.e. Purpose of standby6. At Issuance, Amount of Third Standby 6.a Beneficiary of Standby and Country 6.b. Date Issued 6.c Expiration Date 6.d L/C Reference Number Assigned by the Bank 6.e. Purpose of standby At issuance, Amount of Fourth Standby 7.a Beneficiary of Standby and Country 7.b. Date Issued 7.c Expiration Date 7.d L/C Reference Number Assigned by the Bank 7.e. Purpose of standby 8. Total of Standbys 0.00 Amount C. Borrowing Base (a) 90% of Eligible Export-Related Accounts Receivable (from line 3) 0 10. Outstanding and Operative Standby Letters of Credit as of the Reporting Date 11. 25% Collateral Required Against Standby Letters of Credit 0 12. Borrowing Base Less 25% Collateral (line 9.a less line 11) is collateral availability to support new standby letters of credit 0 13. A negative amount in line 12 indicates the additional collateral needed in the form of export A/R or cash to cover the overadvance. XXXX \$ 20,000,000

9. Borrowing Base:

14. Maximum Amount of Line

15. Maximum amount of Standby L/Cs which can be issued provided sufficient collateral (line 20 - line 21)

16. Maximum new Standby L/Cs allowed based on positive collateral availability on line 12 (four times line 12)

17. Final Availability is the minimum of line 15 or line 16

The Borrower represents and warrants that this Borrowing Base Certificate is a true and correct statement regarding the status of the matters set forth herein. The Borrower further represents and warrants that no Default or Event of Default has occurred and is continuing, and that the Eligible Export-Related Accounts Receivable have been generated from the sale of Items which have been exported. The Borrower acknowledges that any Letters of Credit issued on for the account of the Borrower under the Credit Agreement will be based upon the Lender's reliance on the information contained herein.

Great Lakes Dredge & Dock Corporation

Date Signed & Signed by:

Name and Title of Signer:

Submitted:

20,000,000

0

\$

Country

Bahrain

Note: Ex-Im Bank prohibits or restricts its support of short-term transactions for various countries shown in its Country Limitation Schedule ("CLS"). The most current CLS can be viewed on Ex-Im Bank's website at www.exim.gov under "Apply".

Since Puerto Rico is a U.S. Territory and not a foreign country, Puerto Rican accounts cannot be treated as export- related accounts receivable.

EXHIBIT "D"

COMPLIANCE CERTIFICATE

(International)

See attached.

Attachment to Exhibit D

COMPLIANCE CERTIFICATE

(International)

I,, hereby do certify on behalf of Great Lakes Dredge & Dock Corporation, a Delaware corporation (the " <u>Borrower</u> "), in my
capacity solely as an officer of the Borrower and not in my individual capacity, pursuant to the International Letter of Credit Agreement dated as of September 29
2006 (the "Letter of Credit Agreement") by and among the Borrower, the other Loan Parties party thereto and Wells Fargo HSBC Trade Bank, N.A., as follows:
1. I am the duly elected, qualified and acting of the Borrower.
2. No Default or Event of Default has occurred and is continuing under the Letter of Credit Agreement on the date hereof.
[If an Default or an Event of Default Exists, the Borrower must state the nature thereof, the period of existence thereof and what action the Borrower has aken or proposes to take with respect thereto.]
3. This certificate is the "Compliance Certificate" required to be delivered pursuant to <u>Section 8.1</u> of the Letter of Credit Agreement [for the Fiscal Quarter ending
4. Set forth on Schedules A through D hereto are the calculations which provide the basis for the certification in paragraph 3 above.
All terms used and not otherwise defined herein shall have the meanings ascribed to such terms in the Letter of Credit Agreement.

IN WITNESS WHEREOF, the undersigned has executed this Compliance Certificate on behalf of	the Borrower as of the day of, 20
GREAT	LAKES DREDGE & DOCK CORPORATION
By: Name: Title:	
2	

 $\underline{\text{Schedule } A}$

[Capital Expenditures]

Schedule B

Total Leverage

Schedule C

Senior Leverage

Schedule D

Interest Coverage Ratio

EXHIBIT "E"

LETTER OF CREDIT REQUEST

(International)

See attached.

Attachment to Exhibit E

LETTER OF CREDIT REQUEST

(International)

		20	
		20	

Wells Fargo HSBC Trade Bank, N.A. 1000 Louisiana Street, Fourth Floor Houston, Texas 77002 Attention: Mr. Scott Gildea

Gentlemen:

Reference is made to that certain International Letter of Credit Agreement dated as of September ___, 2006, (the "<u>Agreement</u>") executed by and among GREAT LAKES DREDGE & DOCK CORPORATION, a Delaware corporation (the "<u>Borrower</u>"), the other Loan Parties party thereto and WELLS FARGO HSBC TRADE BANK, N.A. (the "<u>Bank</u>"). Capitalized terms which are used but not defined herein shall have the respective meanings assigned to such terms in the Agreement.

The Borrower hereby requests the issuance of a Letter of Credit under the Agreement, and in that connection sets forth on the Application all necessary information relating to such Letter of Credit ("Proposed Letter of Credit") as required by Section 2.3 of the Agreement.

The Borrower hereby certifies that after giving effect to the issuance of the Proposed Letter of Credit the sum of the Stated Amount of such Letter of Credit plus all Letter of Credit Outstandings will not exceed the Commitment.

The Borrower hereby further certifies that:

- (a) on the date hereof all applicable conditions to the issuance of the Proposed Letter of Credit set forth in <u>Section 2.3</u> and <u>Article VII</u> of the Agreement have been satisfied and the Proposed Letter of Credit complies with the terms of the Agreement;
- (b) as of the date hereof and as a result of the issuance of the Proposed Letter of Credit there does not and will not exist any Default or Event of Default; and
- (c) the representations and warranties contained in <u>Article VI</u> of the Agreement and in each other International Loan Document are true and correct in all material respects on and as of the date hereof and shall be true and correct in all material respects upon the issuance of the Proposed Letter of Credit, with the same force and effect as though made on and as of the date hereof and thereof, except for any representation or warranty which is specified as being made as of an earlier date, in which case such representation or warranty shall only speak as to such earlier date.

Upon the issuance of the Proposed Letter of Credit, the Borrower will be deemed to have recertified the foregoing on such issuance date.					
EXECUTED AND DELIVERED this day of, 20					
	GREAT LAKES DREDGE & DOCK CORPORATION				
	Ву:				
	Name:				
	Title:				

EXHIBIT "F"

GUARANTY

(International)

See attached.

Attachment to Exhibit F

GUARANTY

This Guaranty (herein referred to as the "Guaranty") is dated as of September 29, 2006, and is executed by GREAT LAKES DREDGE & DOCK COMPANY, LLC, a Delaware limited liability company (the "Guarantor"), in order to induce WELLS FARGO HSBC TRADE BANK, N.A., a national banking association ("Trade Bank"), and WELLS FARGO BANK, NATIONAL ASSOCIATION, a national banking association ("Wells Fargo") (Trade Bank and Wells Fargo are singly, a "Bank", and collectively, the "Banks"), to provide financial accommodations to GREAT LAKES DREDGE & DOCK CORPORATION, a Delaware corporation (the "Borrower").

1. **GUARANTY**; **DEFINITIONS**. In consideration of any credit or other financial accommodation heretofore, now or hereafter extended or made to the Borrower by the Banks and for other valuable consideration, the undersigned Guarantor and any debtor-in-possession or trustee in bankruptcy, heir, executor or other successor which succeeds to the interest of the Guarantor hereby unconditionally guarantees the full and prompt payment to the Banks, or order, on demand in lawful money of the United States of America and in immediately available funds, of any and all Obligations of the Borrower to the Banks, provided, however the Guarantor acknowledges and agrees (a) the Obligations guaranteed hereunder shall include obligations and liabilities arising out of the Standby Letter of Credit Agreement and all instruments, certificates and agreements executed in connection therewith owed by any Loan Party to Wells Fargo or any affiliate thereof pursuant to the terms thereto and (b) all references herein to "Obligations" herein shall include such Obligations described in clause (a) hereof for all purposes hereunder. As used in this Guaranty, the term "Obligations" means all Obligations as defined in that certain International Letter of Credit Agreement dated as of September 29, 2006, by and among the Borrower, the other Loan Parties, including the Guarantor, from time to time party thereto and Trade Bank (as amended, restated, supplemented or otherwise modified from time to time, the "Letter of Credit Agreement"); terms used herein and not otherwise defined herein shall have the meanings set forth in the Letter of Credit Agreement). Notwithstanding anything in this Guaranty to the contrary, the obligations of the Guarantor under this Guaranty shall be limited to a maximum aggregate amount equal to the largest amount that would not render the Guarantor's obligations hereunder subject to avoidance as a fraudulent transfer or fraudulent conveyance under Section 548 of Title 11 of the United States Code or any applicable provisions of comparable state law (collectively, the "Fraudulent Transfer Laws"), in each case after giving effect to all other liabilities of the Guarantor, contingent or otherwise, that are relevant under the Fraudulent Transfer Laws and after giving effect as assets to the value (as determined under the applicable provisions of the Fraudulent Transfer Laws) of any rights to subrogation, reimbursement or contribution of the Guarantor pursuant to (y) applicable law, or (z) any agreement providing for rights of subrogation, reimbursement or contribution in favor of the Guarantor, or for an equitable allocation among the Guarantor, the Borrower, and any other person or entity liable for the Obligations.

2. <u>SUCCESSIVE TRANSACTIONS</u>; <u>REVOCATIONS</u>; <u>OBLIGATION UNDER OTHER GUARANTIES</u>. This Guaranty shall be an absolute, continuing, irrevocable, and unconditional guaranty of payment and performance and not a guaranty of collection, and

the Guarantor shall remain liable on its obligations hereunder until the indefeasible payment and performance in full of the Obligations and the termination of all commitments by the Trade Bank to extend credit to or for the benefit of the Borrower under the Letter of Credit Agreement. All rights, powers and remedies under this Guaranty shall apply to all past, present and future Obligations of the Borrower to the Banks, including those arising under successive transactions which shall either continue the Obligations, increase or decrease the Obligations, or from time to time create new Obligations after all or any prior part of the Obligations has been satisfied, and notwithstanding the dissolution, liquidation or bankruptcy of the Borrower or the Guarantor or any other event or proceeding affecting the Borrower or the Guarantor. No set-off, counterclaim, recoupment, reduction, or diminution of any obligation, or any defense of any kind or nature (other than payment or performance) which the Borrower may have against either Bank, or any other party, or which the Guarantor may have against the Borrower, either Bank, or any other party, shall be available to, or shall be asserted by, the Guarantor against either Bank, or any subsequent holder of the Obligations or any part thereof or against payment of the Obligations or any part thereof. If the Guarantor becomes liable for any indebtedness owing by the Borrower to the Banks by endorsement or otherwise, other than under this Guaranty, such liability shall not be in any manner impaired or affected hereby, and the rights of the Banks hereunder shall be cumulative of any and all other rights that the Banks may ever have against the Guarantor. The exercise by either Bank of any right or remedy hereunder or under any other instrument, or at law or in equity, shall not preclude the concurrent or subsequent exercise of any other right or remedy.

3. SEPARATE ACTIONS; WAIVER OF STATUTE OF LIMITATIONS; REINSTATEMENT OF LIABILITY. The obligations hereunder are independent of the obligations of the Borrower, and a separate action or actions may be brought and prosecuted against the Guarantor whether action is brought against the Borrower or any other person, or whether the Borrower or any other person is joined in any such action or actions. A separate action or actions may be bought and prosecuted against the Guarantor by the either Bank; the Banks do not need to join each other in any such action. The Guarantor acknowledges that there are no conditions precedent to the effectiveness of this Guaranty, and this Guaranty is in full force and effect and is binding on the Guarantor as of the date first written above, regardless of whether either Bank obtains collateral or any guaranties from others or takes any other action contemplated by the Guarantor. The Guarantor waives, to the extent permitted under applicable law, the benefit of any statute of limitations affecting the Guarantor's liability hereunder or the enforcement thereof, and the Guarantor agrees that any payment of any Obligations or other act which shall toll any statute of limitations applicable thereto shall similarly operate to toll such statute of limitations applicable to the Guarantor's liability hereunder. The liability of the Guarantor hereunder shall be reinstated and revived and the rights of the Banks shall continue if and to the extent for any reason any amount at any time paid on account of any Obligations guaranteed hereby is rescinded or must otherwise be restored by either Bank, whether as a result of any proceeding in bankruptcy or reorganization or otherwise, all as though such amount had not been paid. The determination as to whether any amount so paid must be rescinded or restored shall be made by each Bank in its sole discretion; provided, however, that if a Bank chooses to contest any such matter at the request of the Guarantor, the Guarantor agrees to indemnify and hold th

- 4. <u>AUTHORIZATIONS TO THE BANKS</u>. The Guarantor authorizes each Bank either before or after revocation hereof, without notice to or demand on the Guarantor, and without affecting the Guarantor's liability hereunder, from time to time to: (a) alter, compromise, renew, extend, accelerate or otherwise change the time for payment of, or otherwise change the terms of the Obligations or any portion thereof, including any increase in the principal amount of the Obligations or available credit facilities, any increase or decrease of the rate of interest thereon, or provide additional types of credit facilities and services; (b) take and hold security for the payment of this Guaranty or the Obligations or any portion thereof, and exchange, enforce, waive, subordinate or release any such security; (c) to accept and hold any endorsement or guaranty of payment of the Obligations or any part thereof, and to discharge, release or substitute any one or more of the endorsers or any other guarantors of the Obligations, or any portion thereof, or any other party thereto, and to enforce or refrain from enforcing, or compromise or modify, the terms of any obligations of any such endorser, guarantor, person or entity; (d) apply such security and direct the order or manner of sale thereof, including, without limitation, a non-judicial sale permitted by the terms of the International Loan Documents, as such Bank in its discretion may determine; (e) apply payments received by such Bank from the Borrower or any other person or entity to any Obligations owed to such Bank, in such order as such Bank shall determine in its sole discretion, whether or not such Obligations are covered by this Guaranty, and the Guarantor hereby waives any provision of law regarding application of payments which specifies otherwise. Each Bank may without notice assign this Guaranty in whole or in part.
- 5. <u>COVENANTS</u>. The Guarantor covenants and agrees that, as long as the Obligations or any part thereof are outstanding or the Trade Bank has any commitment to the Borrower, the Guarantor will timely furnish promptly to the Trade Bank such information concerning Guarantor as is required under the terms of the Letter of Credit Agreement.
- 6. **REPRESENTATIONS AND WARRANTIES.** The Guarantor represents and warrants to the Banks that: (a) this Guaranty is executed at the Borrower's request; (b) neither Bank has made any representation to the Guarantor as to the creditworthiness of the Borrower or any other person or entity liable for the Obligations; (c) the Guarantor has established adequate means of obtaining from the Borrower on a continuing basis financial and other information pertaining to the Borrower's financial condition; (d) the Guarantor has the power and authority to execute, deliver and perform its obligations under this Guaranty, and this Guaranty constitutes the legal, valid and binding obligation of the Guarantor, enforceable against the Guarantor in accordance with its terms, except as limited by Debtor Laws; (e) the execution, delivery, and performance by the Guarantor of this Guaranty will not violate any law or any order of any court, governmental authority or arbitrator and will not conflict with, result in a breach of, or constitute a default under, or result in the imposition of any lien (other than Liens created or permitted under the International Loan Documents) upon any assets of the Guarantor pursuant to the provisions of any indenture, mortgage, deed of trust, security agreement, franchise, permit, license, or other instrument or agreement to which the Guarantor or its properties is bound; (f) the Guarantor has read and understands the terms and conditions of the International Loan Documents; (g) the value of the consideration received and to be received by the Guarantor as a result of the Banks making extensions of credit to the Borrower and the Guarantor executing and delivering this Guaranty is reasonably worth at least as much as the liability and obligation of the Guarantor hereunder, and such liability and obligation and such extensions of credit may reasonably be expected to benefit the Guarantor directly or indirectly; (h) the fair value of the

assets of the Guarantor (on a going concern basis) is, after giving effect to this Guaranty, greater than the total amount of liabilities (including contingent and unliquidated liabilities) of the Guarantor after giving effect to this Guaranty, and that, after giving effect to this Guaranty, the Guarantor is able to pay all of its liabilities as they mature in the ordinary course and it does not have unreasonably small capital with which to carry on its business. The Guarantor agrees to keep adequately informed from such means of any facts, events or circumstances which might in any way affect the Guarantor's risks hereunder, and the Guarantor further agrees that the Banks shall not have any obligation to disclose to the Guarantor any information or material about the Borrower or any other person or entity liable for the Obligations which is acquired by the Banks in any manner.

7. THE GUARANTOR'S WAIVERS.

- 7.1. The Guarantor waives any right to require the Banks to: (a) proceed against the Borrower or any other person; (b) marshal assets or proceed against or exhaust any security held from the Borrower or any other person; (c) give notice of the terms, time and place of any public or private sale of personal property security held from the Borrower or any other person, or otherwise comply with the provisions of Section 9-611 of the Uniform Commercial Code as enacted in the State of New York (the "<u>UCC</u>"); (d) take any action or pursue any other remedy in the Banks' power; or (e) make any presentment or demand for performance, or give any notice of non-performance, protest, notice of protest or notice of dishonor in connection with any obligations or evidences of indebtedness held by the Banks as security for or which constitute in whole or in part the Obligations guaranteed hereunder, or in connection with the creation of new or additional Obligations.
- 7.2. The Guarantor irrevocably and unconditionally waives, to the extent permitted under applicable law, all suretyship defenses and any defense to its obligations hereunder based upon or arising by reason of: (a) any disability or other defense of the Borrower or any other person; (b) the cessation or limitation from any cause whatsoever, other than indefeasible payment in full, of the Obligations of the Borrower or any other person; (c) any lack of authority of any officer, director, partner, agent or any other person acting or purporting to act on behalf of the Borrower which is a corporation, partnership or other type of entity, or any defect in the formation of the Borrower; (d) the application by the Borrower of the proceeds of any Obligations for purposes other than the purposes represented by the Borrower to, or intended or understood by, the Banks or the Guarantor; (e) any act or omission by the Banks which directly or indirectly results in or aids the discharge of the Borrower or any other person or entity liable for the Obligations or any portion of the Obligations by operation of law or otherwise, or which in any way impairs or suspends any rights or remedies of the Banks against the Borrower or any other person or entity liable for the Obligations; (f) any impairment of the value of any interest in any security for the Obligations or any portion thereof, including, without limitation, the failure to obtain or maintain perfection or recordation of any interest in any such security; (g) any modification of the Obligations, in any form whatsoever, including any modification made after revocation hereof to any Obligations incurred prior to such

revocation, renewal, extension, acceleration or other change in the time for payment of, or other change in the terms of, the Obligations or any portion thereof, including increase in the principal amount of the Obligations or available credit facilities, any increase or decrease of the rate of interest thereon or provide additional types of credit facilities or services; (h) the invalidity or unenforceability of any of the Obligations; or (i) it is especially and expressly agreed that if the indebtedness of the Borrower now or at any time hereafter exceeds the amount permitted by law, or if the Borrower's obligation to pay interest, attorneys' fees, costs or expenses or any other sums ceases to exist by operation of law, or if the Borrower is not liable because the act of creating the obligation is ultra vires, or the officers creating same acted without authority, and for these reasons the Obligations to the Banks which the Guarantor agrees to pay cannot be enforced against the Borrower, such fact shall in no manner affect the Guarantor's liability hereunder, notwithstanding the fact that the Borrower is not liable or such Obligations, but the Guarantor shall be liable hereunder to the same extent as the Guarantor would have been if the Obligations of the Borrower had been fully enforceable against said the Borrower. The Guarantor hereby waives notice of intent to accelerate the Obligations and notice of acceleration of the Obligations. Until all Obligations shall have been indefeasibly paid in full and the termination of all commitments of the Trade Bank to extend credit to or for the benefit of the Borrower under the Letter of Credit Agreement, the Guarantor shall have no right of subrogation, and the Guarantor waives any right to enforce any remedy which either Bank now has or may hereafter have against the Borrower or any other person, and waives any benefit of, or any right to participate in, any security now or hereafter held by the Banks. The Guarantor further waives all rights and defenses the Guarantor may have arising out of (A) any election of remedies by either Bank, even though that election of remedies, such as a non-judicial foreclosure with respect to any security for any portion of the Obligations, destroys the Guarantor's rights of subrogation or the Guarantor's rights to proceed against the Borrower or any other person or entity for reimbursement, or (B) any loss of rights the Guarantor may suffer by reason of any rights, power or remedies of the Borrower in connection with any anti-deficiency laws or any other laws limiting, qualifying or discharging the Borrower or any of the Obligations. Until all Obligations shall have been indefeasibly paid in full and the termination of all commitments by the Trade Bank to extend credit to or for the benefit of the Borrower under the Letter of Credit Agreement, the Guarantor further waives any right to enforce any remedy which either Bank now has or may hereafter have against the Borrower or any other person, and waives any benefit of, or any right to participate in, any security now or hereafter held by either Bank.

8. THE BANKS' RIGHTS WITH RESPECT TO THE GUARANTOR'S PROPERTY IN THE BANKS' POSSESSION. If an Event of Default has occurred and is continuing, the Guarantor hereby gives and confirms to each Bank a right of set-off of all moneys, securities and other property of the Guarantor (whether special, general or limited) and the proceeds thereof, now or hereafter delivered to remain with or in transit in any manner to the Banks, its correspondents or its agents from or for the Guarantor, whether for safekeeping, custody, pledge, transmission, collection or otherwise or coming into possession of the Banks in any way, and also, any balance of any deposit accounts and credits of the Guarantor with, and any and all claims of security for the payment of the Obligations, and the Guarantor hereby authorize each Bank at any time or times, without prior notice, to apply such money, securities,

other property, proceeds, balances, credits of claims, or any part of the foregoing, to such Obligations in such amounts as it may select, whether such liabilities be contingent, unmatured or otherwise, and whether any collateral security therefor is deemed adequate or not. The rights described herein shall be in addition to any collateral security described in any separate agreement executed by the Guarantor.

- 9. **REMEDIES; NO WAIVER.** All rights, powers and remedies of the Banks hereunder are cumulative. No delay, failure or discontinuance of either Bank in exercising any right, power or remedy hereunder shall affect or operate as a waiver of such right, power or remedy; nor shall any single or partial exercise of any such right, power or remedy preclude, waive or otherwise affect any other or future exercise thereof or the exercise of any other right, power or remedy. Any waiver, permit, consent or approval of any kind by either Bank either any breach of this Guaranty, or any such waiver of any provision or condition thereof, must be in writing, must be signed by the Banks and shall be in effect only to the extent set forth in writing.
- 10. **COSTS, EXPENSES AND ATTORNEYS' FEES.** The Guarantor shall pay to each Bank promptly upon demand the full amount of all reasonable out-of-pocket payments, advances, charges, costs and expenses, including attorneys' fees, expended or incurred by such Bank in connection with (a) the enforcement of this Guaranty, including, without limitation, any of the foregoing incurred in connection with any bankruptcy proceeding relating to the Borrower, the Guarantor or any other person or entity liable for the Obligations, (b) the collection of any of the Obligations, and (c) realizing upon or protecting any security for this Guaranty or for payment of any Obligations.
- 11. **ADDITIONAL GUARANTORS.** Upon the execution and delivery by any Person of a Joinder to Guaranty in the form of Exhibit G to the Letter of Credit Agreement, such Person (i) shall be and become a Guarantor hereunder and (ii) a party to the Letter of Credit Agreement and each reference in this Guaranty and the other International Loan Documents to "Guarantor" shall also mean a reference to such Person. The execution and delivery of such Joinder to Guaranty shall not require the consent of any other Guarantor hereunder, and the rights and obligations of each Guarantor hereunder shall remain in full force and effect notwithstanding the addition of any new Guarantor as a party to this Guaranty.
- 12. <u>SUCCESSORS, ASSIGNS</u>. This Guaranty shall be binding upon and inure to the benefit of the successors and assigns of the parties (including any debtor-in-possession or trustee in bankruptcy); provided, that the Guarantor may not assign or transfer any of its interest, rights or obligations hereunder without the Banks' prior written consent. The Guarantor acknowledges that each Bank has the right to sell, assign, transfer, negotiate or grant participation in all or in any part of, or any interest in, any Obligations of the Borrower to such Bank and any obligations with respect thereto, including this Guaranty. In connection therewith, each Bank may disclose all documents and information which such Bank now has or hereafter acquires relating to the Guarantor and/or this Guaranty, whether furnished by the Borrower, the Guarantor or otherwise. The Guarantor further agrees that either Bank may disclose such documents and information to the Borrower.
- 13. **AMENDMENT.** This Guaranty is absolute and unconditional and shall not be changed or affected by any representation, oral agreement, act or thing whatsoever, except as

herein provided. This Guaranty is intended by the Guarantor to be the final, complete and exclusive expression of the agreement between the Guarantor and the Banks. This Guaranty may be amended or modified only in writing signed by the Banks and the Guarantor.

- 14. <u>UNDERSTANDING WITH RESPECT TO WAIVERS; SEVERABILITY OF PROVISIONS</u>. The Guarantor warrants and agrees that each of the waivers set forth herein is made by the Guarantor's full knowledge of its significance and consequences, and that under the circumstances, the waivers are reasonable and not contrary to public policy or law. If any waiver or other provision of this Guaranty shall be held to be prohibited by or invalid under applicable public policy or law, such waiver or other provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such waiver or other provision or any remaining provisions of this Guaranty.
- 15. **COUNTERPARTS.** This Guaranty may be executed in any number of counterparts, each of which shall constitute an original, but all of which when taken together shall constitute one and the same Guaranty. A telecopy or electronic copy of any such executed counterpart shall be deemed valid as an original.
- 16. **GOVERNING LAW.** This Guaranty shall be governed by and construed in accordance with the laws of the State of New York without regard to its conflicts of law principles (other than the provisions of 5-1401 and 5-1402 of the New York General Obligations Law). **THE PARTIES WAIVE ANY RIGHT TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING BASED ON OR PERTAINING TO THIS GUARANTY.**
- 17. <u>PLACE OF PERFORMANCE</u>. PAYMENTS OF ALL SUMS OF MONEY AND THE PERFORMANCE OF ALL OF THE COVENANTS AND AGREEMENTS HEREUNDER SHALL BE PAYABLE AND DUE AT THE OFFICE OF THE TRADE BANK SET FORTH IN THE LETTER OF CREDIT AGREEMENT.
- 18. **COMPLIANCE WITH APPLICABLE USURY LAWS.** It is the intention of the Banks and the Guarantor to contract in strict compliance with applicable usury law from time to time in effect. In furtherance therefore, such persons stipulate and agree that none of the terms and provisions contained in this Guaranty or any other International Loan Document shall ever be construed to create a contract to pay for the use, forbearance or detention of money interest in excess of the maximum interest permitted to be charged by applicable law from time to time in effect. Neither the Guarantor nor any of its subsidiaries or affiliates nor any present or future guarantors, endorsers, or other parties hereafter becoming liable for payment of the Obligations shall ever be liable for unearned interest thereon or shall ever be required to pay interest thereon in excess of maximum interest that may be lawfully charged under applicable law from time to time in effect, and that provisions of this Section shall control over all other provision of this Guaranty or any other International Loan Document which may be in apparent conflict herewith. In the event that the Banks or any other holder of the Obligations shall collect moneys which are deemed to constitute interest which would otherwise increase the effective interest on the Obligations or any part thereof to an amount in excess of that permitted to be charged by applicable law then in effect, all such sums deemed to constitute interest in excess of such legal limit shall be immediately returned to the Guarantor or the other payor thereof upon such determination. In determining whether or not the interest paid or payable, under any specific

contingency, exceeds the maximum amount permitted under applicable law, the Banks and the Guarantor (and any other payors thereof) shall to the greatest extent permitted under applicable law, (a) characterize any non-principal payment as an expense, fee or premium rather than as interest, (b) exclude voluntary prepayments and the effects thereof, and (c) amortize, prorate, allocate, and spread the total amount of interest throughout the maximum available period of time in accordance with the amounts owing by the Guarantor from time to time and the maximum legal rates of interest from time to time in effect under applicable law in order to lawfully charge the maximum amount of interest permitted under applicable law. The term "applicable law" as used herein shall mean the law of the State of New York or the laws of the United States, whichever laws allow the greater rate of interest, as such laws now exist or may be changed or amended or come into effect in the future.

- 19. <u>FINAL AGREEMENT</u>. THIS WRITTEN AGREEMENT REPRESENTS THE FINAL AGREEMENT BETWEEN THE PARTIES AND MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OF THE PARTIES. THERE ARE NO UNWRITTEN ORAL AGREEMENTS BETWEEN THE PARTIES.
- 20. NOTICES. ALL NOTICES, REQUESTS AND OTHER COMMUNICATIONS TO THE GUARANTOR OR THE TRADE BANK HEREUNDER SHALL BE GIVEN IN WRITING (INCLUDING BANK WIRE, TELEX OR SIMILAR WRITING AND SHALL BE EFFECTIVE IN ACCORDANCE WITH THE PROVISIONS OF THE LETTER OF CREDIT AGREEMENT, AND SHALL BE GIVEN FOR (A) THE GUARANTOR AT THE "ADDRESSES FOR NOTICES" SPECIFIED ON THE SIGNATURE PAGE HEREOF AND (B) THE TRADE BANK AT THE ADDRESS PROVIDED IN THE LETTER OF CREDIT AGREEMENT.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the undersigned has executed this Guaranty as of the date first written above.

GREAT LAKES DREDGE & DOCK COMPANY, LLC

By:

Deborah A. Wensel Senior Vice President, Chief Financial Officer and Treasurer

Address for Notices:

2122 York Road

Oak Brook, Illinois 60523

Attention: Ms. Deborah A. Wensel Telephone No.: 630-574-2948 Telefax No.: 630-574-3007

EXHIBIT "G"

JOINDER TO GUARANTY

(International)

See attached.

Attachment to Exhibit G

JOINDER TO GUARANTY

THIS JOINDER TO GUARANTY, dated as of ________, 20___ (this "<u>Joinder</u>"), is to the Guaranty, dated as of September 29, 2006 (as amended, supplemented, amended, restated or otherwise modified from time to time, the "<u>Guaranty</u>"), among the Guarantor (such capitalized term, and other terms used in this Joinder, to have the meanings set forth in the Guaranty) from time to time party thereto, in favor the Banks.

WITNESSETH:

WHEREAS, pursuant to the provisions of <u>Section 11</u> of the Guaranty and as required by the Letter of Credit Agreement, the undersigned is becoming a Guarantor under the Guaranty;

WHEREAS, the undersigned Guarantor desires to become a "Guarantor" under the Guaranty in order to induce the Banks to continue to extend credit under the Letter of Credit Agreement dated as of September 29, 2006 (as amended, restated, supplemented or otherwise modified from time to time); and

NOW, THEREFORE, in consideration of the premises, and for other consideration (the receipt and sufficiency of which is hereby acknowledged), the undersigned agrees, for the benefit of the Banks, as follows.

SECTION 1. <u>Party to Guaranty, etc.</u> In accordance with the terms of the Guaranty, by its signature below the undersigned hereby irrevocably agrees to become a Guarantor under the Guaranty with the same force and effect as if it were an original signatory thereto and the undersigned Guarantor hereby (a) agrees to be bound by and comply with all of the terms and provisions of the Guaranty applicable to it as a Guarantor and (b) represents and warrants that the representations and warranties made by it as a Guarantor thereunder are true and correct as of the date hereof, unless stated to relate solely to an earlier date, in which case, such representations and warranties shall be true and correct in all material respects as of such earlier date. In furtherance of the foregoing, each reference to a "Guarantor" in the Guaranty and the other International Loan Documents shall be deemed to include the undersigned Guarantor.

SECTION 2. <u>Representations</u>. The undersigned Guarantor hereby represents and warrants that this Joinder has been duly authorized, executed and delivered by it and that this Joinder and the Guaranty constitute the legal, valid and binding obligation of the undersigned Guarantor, enforceable against it in accordance with its terms, except as limited by Debtor Laws.

SECTION 3. Full Force of Guaranty. Except as expressly supplemented hereby, the Guaranty shall remain in full force and effect in accordance with its terms.

SECTION 4. <u>Severability</u>. In the event any one or more of the provisions contained in this Supplement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein and in the Guaranty shall not in any way be affected or impaired.

Joinder to Guaranty - 1

SECTION 5. Governing Law. THIS JOINDER SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK WITHOUT REGARD TO ITS CONFLICTS OF LAW PRINCIPLES (OTHER THAN THE PROVISIONS OF 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW). **THE PARTIES WAIVE ANY RIGHT TO TRIAL BY JURY IN ANY ACTION OR PROCEEDING BASED ON OR PERTAINING TO THIS GUARANTY.** THIS JOINDER AND THE OTHER INTERNATIONAL LOAN DOCUMENTS CONSTITUTE THE ENTIRE UNDERSTANDING AMONG THE PARTIES HERETO WITH RESPECT TO THE SUBJECT MATTER HEREOF AND SUPERSEDE ANY PRIOR AGREEMENTS, WRITTEN OR ORAL, WITH RESPECT THERETO.

SECTION 6. <u>Counterparts</u>. This Joinder may be executed by the parties hereto in several counterparts, each of which shall be deemed to be an original (including electronic and telecopy copies) and all of which shall constitute together but one and the same agreement.

Remainder of page intentionally left blank. Signature page to follow.

Joinder to Guaranty - 2

	[NAME OF GUARANTOR]	
	By: Name:	
ACCEPTED AND AGREED:		
WELLS FARGO HSBC TRADE BANK, N.A.		
By: Name: Citle:		
WELLS FARGO BANK, NATIONAL ASSOCIATION		
Name:		

IN WITNESS WHEREOF, the undersigned Guarantor has caused this Joinder to be duly executed and delivered by an authorized officer or representative as of the date first above written.

Joinder to Guaranty - 3

EXHIBIT "H"

STANDBY LETTER OF CREDIT AGREEMENT

(International)

See attached.

Attachment to Exhibit H

STANDBY LETTER OF CREDIT AGREEMENT

To: WELLS FARGO BANK, NATIONAL ASSOCIATION

Applicant hereby requests that you, Wells Fargo Bank, National Association ("Wells Fargo"), issue in your name one or more standby letters of credit pursuant to Applications for the issuance of such Credits and the terms and conditions of this Agreement. Each Credit will be issued at Applicant's request and for its account, and, unless otherwise specifically provided in any Loan Document, at your option. Applicant agrees that the terms and conditions in this Agreement shall apply to each Application and the Credit issued pursuant to each Application, and to transactions under each Application, each Credit and this Agreement. The Applicant and Wells Fargo further agree that, unless otherwise agreed to in writing by the Applicant and Wells Fargo, all Credits issued pursuant to this Agreement shall be deemed to be "Letters of Credit" issued under and as defined in the Credit Agreement.

SECTION 1. **DEFINITIONS**. As used in this Agreement, the following terms shall have the meanings set forth after each term: "Agreement" means this Standby Letter of Credit Agreement as it may be revised or amended from time to time. "Applicant" means collectively each person and/or entity signing this Agreement as Applicant. "Application" means your printed form titled "Application For Standby Letter of Credit" or any other form acceptable to you on which Applicant applies for the issuance by you of a Credit and/or an application for amendment of a Credit or any combination of such applications, as the context may require. "Beneficiary" means the person or entity named on an Application as the beneficiary or any transferee of such beneficiary. "Collateral" means the Property, together with the proceeds of such Property, securing any or all of Applicant's obligations and liabilities at any time existing under or in connection with any L/C Document and/or any Loan Document. "Commission Fee" means the fee, computed at the commission fee rate specified by you or specified in any Loan Document, charged by you at the time or times specified by you on the amount of each Credit and on the amount of each increase in a Credit for the time period each Credit is outstanding, "Credit" means an instrument or document titled "Irrevocable Standby Letter of Credit" or "Standby Letter of Credit", or any instrument or document whatever it is titled or whether or not it is titled functioning as a standby letter of credit, issued under or pursuant to an Application, and all renewals, extensions and amendments of such instrument or document. "Credit Agreement" means that certain International Letter of Credit Agreement ___, 2006 by and among Great Lakes Dredge & Dock Company, LLC, a Delaware limited liability company, the Applicant, the dated as of September __ other Loan Parties from time to time party thereto, and Wells Fargo HSBC Trade Bank, N.A. as the Bank, as the same may be amended, restated, supplemented or otherwise modified from time to time. "Demand" means any sight draft, electronic or telegraphic transmission or other written demand drawn or made, or purported to be drawn or made, under or in connection with any Credit. "Document" means any instrument, statement, certificate or other document referred to in or related to any Credit or required by any Credit to be presented with any Demand. "Dollars" means the lawful currency at any time for the payment of public or private debts in the United States of America. "Event of Default" means any of the events set forth in Section 13 of this Agreement. "Expiration Date" means the date any Credit expires. "Guarantor" means any person or entity guaranteeing the payment and/or performance of any or all of Applicant's obligations under or in connection with any L/C Document and/or any Loan Document. "Holding Company" means any company or other entity directly or indirectly controlling you. "L/C Document" means this Agreement, each Application, each Credit, and each Demand. "Loan Document" means each and any promissory note, loan agreement (including, but not limited to, the Credit Agreement), security agreement, pledge agreement, guarantee or other agreement or document executed in connection with, or relating to, any extension of credit under which any Credit is issued. "Maximum Rate" means the maximum amount of interest (as defined by applicable laws), if any, permitted to be paid, taken, reserved, received, collected or charged under applicable laws, as the same may be amended or modified from time to time. "Negotiation Fee" means the fee, computed at the negotiation fee rate specified by you or specified in any Loan Document, charged by you on the amount of each Demand paid by you or any other bank specified by you when each Demand is paid. "Payment Office" means the office specified by you or specified in any Loan Document as the office where reimbursements and other payments under or in connection with any L/C Document are to be made by Applicant. "Prime Rate" means the rate of interest most recently announced within Wells Fargo at its principal office as its Prime Rate, with the understanding that the Prime Rate is one of Wells Fargo's base rates and serves as the basis upon which effective rates of interest are calculated for those loans making reference thereto, and is evidenced by the recording thereof after its announcement in such internal publication or publications as Wells Fargo may designate. "Property" means all forms of property, whether tangible or intangible, real, personal or mixed. "Rate of Exchange" means Wells Fargo's then

current selling rate of exchange in San Francisco, California for sales of the currency of payment of any Demand, or of any fees or expenses or other amounts payable under this Agreement, for cable transfer to the country of which such currency is the legal tender. "<u>UCP</u>" means the Uniform Customs and Practice for Documentary Credits, an International Chamber of Commerce publication, or any substitution therefor or replacement thereof. "<u>Unpaid and Undrawn Balance</u>" means at any time the entire amount which has not been paid by you under all the Credits issued for Applicant's account, including, without limitation, the amount of each Demand on which you have not yet effected payment as well as the amount undrawn under all such Credits. "<u>Wells Fargo & Company</u>" means Wells Fargo Bank, National Association, a national banking association.

SECTION 2. **HONORING DEMANDS AND DOCUMENTS**. You may receive, accept and honor, as complying with the terms of any Credit, any Demand and any Documents accompanying such Demand; provided, however, that such Demand and accompanying Documents appear on their face to comply substantially with the provisions of such Credit and are, or appear on their face to be, signed or issued by (a) a person or entity authorized under such Credit to draw, sign or issue such Demand and accompanying Documents, or (b) an administrator, executor, trustee in bankruptcy, debtor in possession, assignee for the benefit of creditors, liquidator, receiver or other legal representative or successor in interest by operation of law of any such person or entity.

SECTION 3. **REIMBURSEMENT FOR PAYMENT OF DEMANDS**. Applicant shall reimburse you for all amounts paid by you on each Demand, including, without limitation, all such amounts paid by you to any paying, negotiating or other bank. If in connection with the issuance of any Credit, you agree to pay any other bank the amount of any payment or negotiation made by such other bank under such Credit upon your receipt of a cable, telex or other written telecommunication advising you of such payment or negotiation, or authorizes any other bank to debit your account for the amount of such payment or negotiation, Applicant agrees to reimburse you for all such amounts paid by you, or debited to your account with such other bank, even if any Demand or Document specified in such Credit fails to arrive in whole or in part or if, upon the arrival of any such Demand or Document, the terms of such Credit have not been complied with or such Demand or Document does not conform to the requirements of such Credit or is not otherwise in order.

SECTION 4. **FEES AND EXPENSES**. Applicant agrees to pay to you (a) all Commission Fees, Negotiation Fees, cable fees, amendment fees, non-usance fees, and cancellation fees of, and all out-of-pocket expenses incurred by, you under or in connection with any L/C Document, and (b) all fees and charges of banks or other entities other than you under or in connection with any L/C Document if any Application (i) does not indicate who will pay such fees and charges, (ii) indicates that such fees and charges are to be paid by Applicant, or (iii) indicates that such fees and charges are to be paid by the Beneficiary and the Beneficiary does not, for any reason whatsoever, pay such fees or charges. There shall be no refund of any portion of any Commission Fee in the event any Credit is used, reduced, amended, modified or terminated before its Expiration Date.

SECTION 5. **DEFAULT INTEREST**. Unless otherwise specified in any Loan Document, or on an Application and agreed to by you, all amounts to be reimbursed by Applicant to you, and all fees and expenses to be paid by Applicant to you, and all other amounts due from Applicant to you under or in connection with any L/C Documents, will bear interest (to the extent permitted by law), payable on demand, from the date you paid the amounts to be reimbursed or the date such fees, expenses and other amounts were due until such amounts are paid in full, at a rate per annum (computed on the basis of a 360-day year, actual days elapsed) which is the lesser of (a) two percent (2%) above the Prime Rate in effect from time to time, or (b) the Maximum Rate.

SECTION 6. TIME AND METHOD OF REIMBURSEMENT AND PAYMENT. Unless otherwise specified in this Section, in any Loan Document, or on an Application and agreed to by you, all amounts to be reimbursed by Applicant to you, all fees and expenses to be paid by Applicant to you, and all interest and other amounts due to you from Applicant under or in connection with any L/C Documents will be reimbursed or paid at the Payment Office in Dollars in immediately available funds without setoff or counterclaim (i) on demand or, (ii) at your option by your debiting any of Applicant's accounts with you, with each such debit being made without presentment, protest, demand for reimbursement or payment, notice of dishonor or any other notice whatsoever, all of which are hereby expressly waived by Applicant. Each such debit will be made at the time each Demand is paid by you or, if earlier, at the time each amount is paid by you to any paying, negotiating or other bank, or at the time each fee and expense is to be paid or any interest or other amount is due under or in connection with any L/C Documents. If any Demand or any fee, expense, interest or other amount payable under or in connection with any L/C Documents is payable in a currency

other than Dollars, Applicant agrees to reimburse you for all amounts paid by you on such Demand, and/or to pay you all such fees, expenses, interest and other amounts, in one of the three following ways, as determined by you in your sole discretion in each case: (a) at such place as you shall direct, in such other currency; or (b) at the Payment Office in the Dollar equivalent of the amount of such other currency calculated at the Rate of Exchange on the date determined by you in your sole discretion; or (c) at the Payment Office in the Dollar equivalent, as determined by you (which determination shall be deemed correct absent manifest error), of such fees, expenses, interest or other amounts or of the actual cost to you of paying such Demand. Applicant assumes all political, economic and other risks of disruptions or interruptions in any currency exchange.

SECTION 7. AGREEMENTS OF APPLICANT. Applicant agrees that (a) unless otherwise specifically provided in any Loan Document, you shall not be obligated at any time to issue any Credit for Applicant's account; (b) unless otherwise specifically provided in any Loan Document, if any Credit is issued by you for Applicant's account, you shall not be obligated to issue any further Credit for Applicant's account or to make other extensions of credit to Applicant or in any other manner to extend any financial consideration to Applicant; (c) you have not given Applicant any legal or other advice with regard to any L/C Document or Loan Document; (d) if you at any time discuss with Applicant the wording for any Credit, any such discussion will not constitute legal or other advice by you or any representation or warranty by you that any wording or Credit will satisfy Applicant's needs; (e) Applicant is responsible for the wording of each Credit, including, without limitation, any drawing conditions, and will not rely on you in any way in connection with the wording of any Credit or the structuring of any transaction related to any Credit; (f) Applicant, and not you, is responsible for entering into the contracts relating to the Credits between Applicant and the Beneficiaries and for causing Credits to be issued; (g) you may, as you deem appropriate, modify or alter and use in any Credit the terminology contained on the Application for such Credit; (h) unless the Application for a Credit specifies whether the Documents to be presented with a Demand under such Credit must be sent to you in one parcel or in two parcels or may be sent to you in any number of parcels, you may, if you so desire, make such determination and specify in the Credit whether such Documents must be sent in one parcel or two parcels or may be sent in any number of parcels; (i) you shall not be deemed Applicant's agent or the agent of any Beneficiary or any other user of any Credit, and neither Applicant, nor any Beneficiary nor any other user of any Credit shall be deemed your agent; (j) Applicant will promptly examine all Documents and each Credit if and when they are delivered to Applicant and, in the event of any claim of noncompliance of any Documents or any Credit with Applicant's instructions or any Application, or in the event of any other irregularity, Applicant will promptly notify you in writing of such noncompliance or irregularity, Applicant being conclusively deemed to have waived any such claim of noncompliance or irregularity unless such notice is promptly given; (k) all directions and correspondence relating to any L/C Document are to be sent at Applicant's risk; (l) if any Credit has a provision concerning the automatic extension of its Expiration Date, you may, at your sole option, give notice of nonrenewal of such Credit and if Applicant does not at any time want such Credit to be renewed Applicant will so notify you at least fifteen (15) calendar days before you are to notify the Beneficiary of such Credit or any advising bank of such nonrenewal pursuant to the terms of such Credit; (m) Applicant will not seek to obtain, apply for, or acquiesce in any temporary or permanent restraining order, preliminary or permanent injunction, permanent injunction or any other pretrial or permanent injunctive or similar relief, restraining, prohibiting or enjoining you, any of your correspondents or any advising, confirming, negotiating, paying or other bank from paying or negotiating any Demand or honoring any other obligation under or in connection with any Credit; and (n) except for Applicant's obligations specifically affected by those actions or failures to act referred to in <u>subsections (ii)</u> and <u>(vii)</u> of this <u>Section 7(n)</u> which you have performed or approved or accepted, Applicant's obligations under or in connection with each L/C Document and Loan Document shall be absolute, unconditional and irrevocable, and shall be performed strictly in accordance with the terms of each such L/C Document and Loan Document under all circumstances whatsoever, including, without limitation, the following circumstances, the circumstances listed in Section 12(b) through (u) of this Agreement, and any other event or circumstance similar to such circumstances: (i) any lack of validity or enforceability of any L/C Document, any Loan Document, any Document or any agreement relating to any of the foregoing; (ii) any amendment of or waiver relating to, or any consent to or departure from, any L/C Document, any Loan Document or any Document; (iii) any release or substitution at any time of any Property held as Collateral; (iv) your failure to deliver to Applicant any Document you have received with a drawing under a Credit because doing so would, or is likely to, violate any law, rule or regulation of any government authority; (v) the existence of any claim, set-off, defense or other right which Applicant may have at any time against you or any Beneficiary (or any person or entity for whom any Beneficiary may be acting) or any other person or entity, whether under or in connection with any L/C Document, any Loan Document, any Document or any Property referred to in or related to any of the foregoing or under or in connection with any unrelated transaction; (vi) any breach of contract or other dispute between or among any two or more of you, Applicant, any Beneficiary, any transferee of any Beneficiary, any person or entity for whom any Beneficiary or any transferee of any Beneficiary may be acting, or any other person or entity; or (vii) any delay, extension of time, renewal, compromise or other indulgence

granted or agreed to by you with or without notice to Applicant, or Applicant's approval, in respect of any of Applicant's indebtedness or other obligations to you under or in connection with any L/C Document or any Loan Document.

SECTION 8. **COMPLIANCE WITH LAWS AND REGULATIONS**. Applicant represents and warrants to you that no Application, Credit or transaction under any Application and/or Credit will contravene any law or regulation of the government of the United States or any state thereof. Applicant agrees (a) to comply with all federal, state and foreign exchange regulations and other government laws and regulations now or hereafter applicable to any L/C Document, to any payments under or in connection with any L/C Document, or to the import, export, shipping or financing of the Property referred to in or shipped under or in connection with any Credit, and (b) to reimburse you for such amounts as you may be required to expend as a result of such laws or regulations, or any change therein or in the interpretation thereof by any court or administrative or government authority charged with the administration of such laws or regulations.

SECTION 9. TAXES, RESERVES AND CAPITAL ADEQUACY REQUIREMENTS. In addition to, and notwithstanding any other provision of any L/C Document or any Loan Document, in the event that any law, treaty, rule, regulation, guideline, request, order, directive or determination (whether or not having the force of law) of or from any government authority, including, without limitation, any court, central bank or government regulatory authority, or any change therein or in the interpretation or application thereof, (a) does or shall subject you to any tax of any kind whatsoever with respect to the L/C Documents, or change the basis of taxation of payments to you of any amount payable thereunder (except for changes in the rate of tax on your net income); (b) does or shall impose, modify or hold applicable any reserve, special deposit, assessment, compulsory loan, Federal Deposit Insurance Corporation insurance or similar requirement against assets held by, deposits or other liabilities in or for the account of, advances or loans by, other credit extended by or any other acquisition of funds by, any of your offices; (c) does or shall impose, modify or hold applicable any capital adequacy requirements (whether or not having the force of law); or (d) does or shall impose on you any other condition; and the result of any of the foregoing is (i) to increase the cost to you of issuing or maintaining any Credit or of performing any transaction under any L/C Document, (ii) to reduce any amount receivable by you under any L/C Document, or (iii) to reduce the rate of return on your capital or the capital of the Holding Company to a level below that which you or the Holding Company could have achieved but for any imposition, modification or application of any capital adequacy requirement (taking into consideration your policy and the policy of the Holding Company, as the case may be, with respect to capital adequacy), and any such increase or reduction is material (as determined by you or the Holding Company, as the case may be, in your or the Holding Company's sole discretion); then, in any such case, Applicant agrees to pay to you or the Holding Company, as the case may be, such amount or amounts as may be necessary to compensate you or the Holding Company for (A) any such additional cost, (B) any reduction in the amount received by you under any L/C Document, or (C) to the extent allocable (as determined by you or the Holding Company, as the case may be, in your or the Holding Company's sole discretion) to any L/C Document, any reduction in the rate of return on your capital or the capital of the Holding Company.

SECTION 10. **COLLATERAL**. As security for any or all of Applicant's obligations and liabilities to you at any time existing under or in connection with any L/C Document or any Loan Document, Applicant has granted to you under the Loan Documents a security interest in and to the collateral specified in the Loan Documents until such time as all Applicant's obligations and liabilities to you at any time existing under or in connection with each L/C Document and each Loan Document have been fully paid and discharged, all as security for such obligations and liabilities. Applicant agrees that the receipt by you of the Collateral or by you or any of your agents or correspondents at any time of any kind of security, including, without limitation, cash, shall not be deemed a waiver of any of your rights or powers under this Agreement. Applicant agrees to sign and deliver to you on demand, all such security agreements, financing statements and other documents as you shall at any time request which are necessary or desirable (in your sole opinion) to grant to you an effective and perfected security interest in and to any or all of the Collateral. Applicant agrees to pay all filing and recording fees related to the perfection of any security interest granted to you in accordance with this Section. Applicant hereby agrees that any or all of the Collateral may be held and disposed of by you as provided in this Agreement and the Loan Documents. Upon any transfer, sale, delivery, surrender or endorsement of any Property which is or was part of the Collateral, Applicant will indemnify and hold you and your agents and correspondents harmless from and against each and every claim, demand, action or suit which may arise against you or any of your agents or correspondents by reason of such transfer, sale, delivery, surrender or endorsement.

SECTION 11. INDEMNIFICATION. Except to the extent caused by your lack of good faith, and notwithstanding

any other provision of this Agreement, Applicant agrees to reimburse and indemnify you for (a) all amounts paid by you to any Beneficiary under or in connection with any guarantee or similar undertaking issued by such Beneficiary to a third party at Applicant's request, whether such request is communicated directly by Applicant or through you to such Beneficiary; and (b) all damages, losses, liabilities, actions, claims, suits, penalties, judgments, obligations, costs or expenses, of any kind whatsoever and howsoever caused, including, without limitation, attorneys' fees and interest, paid, suffered or incurred by, or imposed upon, you directly or indirectly arising out of or in connection with (i) any L/C Document, any Loan Document, any Document or any Property referred to in or related to any Credit; (ii) Applicant's failure to comply with any of its obligations under this Agreement; (iii) the issuance of any Credit; (iv) the transfer of any Credit; (v) any guarantee or similar undertaking, or any transactions thereunder, issued by any Beneficiary to a third party at Applicant's request, whether such request is communicated directly by Applicant or through you to such Beneficiary; (vi) any communication made by you, on Applicant's instructions, to any Beneficiary requesting that such Beneficiary issue a guarantee or similar undertaking to a third party or the issuance of any such guarantee or similar undertaking; (vii) the collection of any amounts Applicant owes to you under or in connection with any L/C Document or any Loan Document; (viii) the foreclosure against, or other enforcement of, any Collateral; (ix) the protection, exercise or enforcement of your rights and remedies under or in connection with any L/C Document or any Loan Document; (x) any court decrees or orders, including, without limitation, temporary or permanent restraining orders, preliminary or permanent injunctions, or any other pretrial or permanent injunctive or similar relief, restraining, prohibiting or enjoining or seeking to restrain, prohibit or enjoin you, any of your correspondents or any advising, confirming, negotiating, paying or other bank from paying or negotiating any Demand or honoring any other obligation under or in connection with any Credit; or (xi) any Credit being governed by laws or rules other than the UCP in effect on the date such Credit is issued. The indemnity provided in this Section will survive the termination of this Agreement and the expiration or cancellation of any or all the Credits.

SECTION 12. LIMITATION OF LIABILITY. Notwithstanding any other provision of this Agreement, neither you nor any of your agents or correspondents will have any liability to Applicant for any action, neglect or omission, if done in good faith, under or in connection with any L/C Document, Loan Document or Credit, including, without limitation, the issuance or any amendment of any Credit, the failure to issue or amend any Credit, or the honoring or dishonoring of any Demand under any Credit, and such good faith action, neglect or omission will bind Applicant. Notwithstanding any other provision of any L/C Document, in no event shall you or your officers or directors be liable or responsible, regardless of whether any claim is based on contract or tort, for (a) any special, consequential, indirect or incidental damages, including, without limitation, lost profits, arising out of or in connection with the issuance of any Credit or any action taken or not taken by you in connection with any L/C Document, any Loan Document, or any Document or Property referred to in or related to any Credit; (b) the honoring of any Demand in accordance with any order or directive of any court or government or regulatory body or entity requiring such honor despite any temporary restraining order, restraining order, preliminary injunction, permanent injunction or any type of pretrial or permanent injunctive relief or any similar relief, however named, restraining, prohibiting or enjoining such honor; (c) the dishonoring of any Demand in accordance with any legal or other restriction in force at the time and in the place of presentment or payment; (d) verifying the existence or reasonableness of any act or condition referenced, or any statement made, in connection with any drawing or presentment under any Credit; (e) the use which may be made of any Credit; (f) the validity of any purported transfer of any Credit or the identity of any purported transferee of any Beneficiary; (g) any acts or omissions of any Beneficiary or any other user of any Credit; (h) the form, validity, sufficiency, correctness, genuineness or legal effect of any Demand or any Document, or of any signatures or endorsements on any Demand or Document, even if any Demand or any Document should in fact prove to be in any or all respects invalid, insufficient, fraudulent or forged; (i) payment by you of any Demand when the Demand and any accompanying Documents appear on their face to comply substantially with the terms of the Credit to which they relate or dishonor by you of any Demand when the Demand and any accompanying Documents do not strictly comply on their face with the terms of the Credit to which they relate; (j) the failure of any Demand or Document to bear any reference or adequate reference to the Credit to which it relates; (k) the failure of any Document to accompany any Demand; (1) the failure of any person or entity to note the amount of any Demand on the Credit to which it relates or on any Document; (m) the failure of any person or entity to surrender or take up any Credit; (n) the failure of any Beneficiary to comply with the terms of any Credit or to meet the obligations of such Beneficiary to Applicant; (o) the failure of any person or entity to send or forward Documents if and as required by the terms of any Credit; (p) any errors, inaccuracies, omissions, interruptions or delays in transmission or delivery of any messages, directions or correspondence by mail, cable, telegraph, wireless or otherwise, whether or not they are in cipher; (q) any notice of nonrenewal of a Credit sent by you not being received on time or at any time by the Beneficiary of such Credit; (r) any inaccuracies in the translation of any messages, directions or correspondence; (s) any Beneficiary's use of the proceeds of any Demand; (t) any Beneficiary's failure to repay to you or Applicant the proceeds of any Demand if the terms of any Credit require such repayment; or (u) any act, error, neglect, default, negligence, gross negligence, omission,

willful misconduct, lack of good faith, insolvency or failure in business of any of your agents or correspondents or of any advising, confirming, negotiating, paying or other bank. The occurrence of any one or more of the contingencies referred to in the preceding sentence shall not affect, impair or prevent the vesting of your rights or powers under this Agreement or any Loan Document or Applicant's obligation to make reimbursement or payment to you under this Agreement or any Loan Document. The provisions of this Section will survive the termination of this Agreement and any Loan Documents and the expiration or cancellation of any or all the Credits.

SECTION 13. EVENTS OF DEFAULT. Each of the following shall constitute an Event of Default under this Agreement: (a) Applicant's or any Guarantor's failure to pay any principal, interest, fee or other amount when due under or in connection with any L/C Document or any Loan Document; (b) the occurrence and continuance of any default or defined event of default under any Loan Document or any other agreement, document or instrument signed or made by Applicant or any Guarantor in your favor; (c) Applicant's or any Guarantor's failure to perform or observe any term, covenant or agreement contained in this Agreement or any Loan Document (other than those referred to in subsections (a) and (b) of this Section, or the breach of any other obligation owed by Applicant or any Guarantor to you, and any such failure or breach shall be impossible to remedy or shall remain unremedied for thirty (30) calendar days after such failure or breach occurs; (d) any representation, warranty or certification made or furnished by Applicant or any Guarantor under or in connection with any L/C Document, any Loan Document or any Collateral, or as an inducement to you to enter into any L/C Document or Loan Document or accept any Collateral, shall be materially false, incorrect or incomplete when made; (e) any material provision of this Agreement or any Loan Document shall at any time for any reason cease to be valid and binding on Applicant or any Guarantor or shall be declared to be null and void, or the validity or enforceability thereof shall be contested by Applicant, any Guarantor or any government agency or authority, or Applicant or any Guarantor shall deny that it has any or further liability or obligation under this Agreement or any Loan Document; (f) any guarantee of, or any security covering, any of Applicant's indebtedness to you arising under or in connection with any L/C Document or any Loan Document fails to be in full force and effect at any time; (g) Applicant or any Guarantor suspends the transaction of its usual business or is expelled or suspended from any exchange; (h) Applicant or any Guarantor dies or is incapacitated; (i) Applicant or any Guarantor dissolves or liquidates; (j) Applicant or any Guarantor is not generally paying its debts as they become due; (k) Applicant or any Guarantor becomes insolvent, however such insolvency may be evidenced, or Applicant or any Guarantor makes any general assignment for the benefit of creditors; (1) a petition is filed by or against Applicant or any Guarantor seeking Applicant's or such Guarantor's liquidation or reorganization under the Bankruptcy Reform Act, Title 11 of the United States Code, as amended or recodified from time to time, or a similar action is brought by or against Applicant or any Guarantor under any federal, state or foreign law; (m) a proceeding is instituted by or against Applicant or any Guarantor for any relief under any bankruptcy, insolvency or other law relating to the relief of debtors, reorganization, readjustment or extension of indebtedness or composition with creditors; (n) a custodian or a receiver is appointed for, or a writ or order of attachment, execution or garnishment is issued, levied or made against, any substantial part of Applicant's or any Guarantor's Property or assets; (o) an application is made by any of Applicant's or any Guarantor's judgment creditors for an order directing you to pay over money or to deliver other of Applicant's or such Guarantor's Property; or (p) any government authority or any court takes possession of any substantial part of Applicant's or any Guarantor's Property or assets or assumes control over Applicant's or any Guarantor's affairs.

SECTION 14. **REMEDIES**. Upon the occurrence and continuance of any Event of Default all amounts paid by you on any Demand which have not previously been repaid to you, together with all interest on such amounts, and the Unpaid and Undrawn Balance, if any, shall automatically be owing by Applicant to you and shall be due and payable by Applicant on demand without presentment or any other notice of any kind, including, without limitation, notice of nonperformance, notice of protest, protest, notice of dishonor, notice of intention to accelerate, or notice of acceleration, all of which are expressly waived by Applicant. Upon payment of the Unpaid and Undrawn Balance to you Applicant shall have no further legal or equitable interest therein, and you will not be required to segregate on your books or records the Unpaid and Undrawn Balance paid by Applicant. After you receive the Unpaid and Undrawn Balance, you agree to pay to Applicant, upon termination of all of your liability under all the Credits and Demands, a sum equal to the amount which has not been drawn under all the Credits less all amounts due and owing to you from Applicant under or in connection with the L/C Documents and the Loan Documents. Further, upon the occurrence and continuance of any Event of Default, you may immediately foreclose against any and all Collateral, without demand for payment, advertisement or notice to Applicant, all of which are hereby expressly waived, in such parcel or parcels, at such times and upon such terms and conditions as you may deem proper, and you may apply the net proceeds of each foreclosure, together with any sums due from you to Applicant, to the payment of any and all obligations and liabilities due from Applicant to you under or in connection with the L/C Documents and the Loan Documents, all without prejudice to your rights against Applicant with respect to any and all such obligations and

liabilities which may be or remain unpaid. All your rights and remedies existing under the L/C Documents and the Loan Documents are in addition to, and not exclusive of, any rights or remedies otherwise available to you under applicable law. In addition to any rights now or hereafter granted under applicable law, and not by way of limitation of any such rights, upon the occurrence and continuance of any Event of Default, Applicant hereby authorizes you at any time or from time to time, without notice to Applicant or to any other person (any such notice being hereby expressly waived by Applicant) and to the extent permitted by law, to appropriate and to apply the Collateral and any and all Applicant's other deposits (general or special, including, without limitation, indebtedness evidenced by certificates of deposit) with you, whether matured or unmatured, and any other indebtedness at any time held or owing by you to or for Applicant's credit or its account, against and on account of Applicant's obligations and liabilities to you under or in connection with any of the L/C Documents or the Loan Documents, irrespective of whether or not you shall have made any demand for payment of any or all such obligations and liabilities or declared any or all such obligations and liabilities to be due and payable, and although any or all such obligations and liabilities contingent or unmatured.

SECTION 15. **WAIVERS**. No delay, extension of time, renewal, compromise or other indulgence which may occur or be granted by you under any L/C Document or any Loan Document shall impair your rights or powers under this Agreement or any Application. You shall not be deemed to have waived any of your rights under this Agreement or any Application unless such waiver is in writing signed by your authorized representative. No such waiver, unless expressly provided therein, shall be effective as to any transactions which occur subsequent to the date of such waiver or as to the continuance of any Event of Default after such waiver. No amendment or modification of this Agreement shall be effective unless it is in writing signed by Applicant's and your authorized representative(s).

SECTION 16. **AMENDMENTS AND MODIFICATIONS TO CREDITS**. At Applicant's verbal or written request, or with Applicant's verbal or written consent, and without extinguishing or otherwise affecting Applicant's obligations under this Agreement or any Loan Document, you may with respect to any Credit, in writing or by any other action, but you will not be obligated to, (a) increase the amount of such Credit, (b) extend the time for, and amend or modify the terms and conditions governing, the making and honoring of any Demand or Document or any other terms and conditions of such Credit, or (c) waive the failure of any Demand or Document to comply with the terms of such Credit, and any Collateral pledged or granted to you in connection with such Credit will secure Applicant's obligations to you with respect to such Credit as amended, modified or waived. No amendment to, or modification of, the terms of any Credit will become effective if the Beneficiary of such Credit or any confirming bank objects to such amendment or modification. If any Credit is amended or modified in accordance with this Section, Applicant shall be bound by, and obligated under, the provisions of this Agreement with respect to such Credit as so amended or modified, and any action taken by you or any advising, confirming, negotiating, paying or other bank in accordance with such amendment or modification.

SECTION 17. SUCCESSORS AND ASSIGNS. The terms and conditions of this Agreement and each Application shall bind Applicant's heirs, executors, administrators, successors and assigns, and all rights, benefits and privileges conferred on you under or in connection with each L/C Document and each Loan Document shall be and hereby are extended to, conferred upon and may be enforced by your successors and assigns. Except as expressly permitted by the Credit Agreement or consented to by the Bank, Applicant will not assign this Agreement or Applicant's obligations or liabilities to you under or in connection with any L/C Document or Loan Document to any person or entity without your prior written approval.

SECTION 18. **GOVERNING LAW**. This Agreement and each Application, and Applicant's and your performance under this Agreement and each Application, shall be governed by, and construed in accordance with, the laws of the State of New York without regard to its conflicts of law principles (other than the provisions of 5-1401 and 5-1402 of the New York general obligations law). Unless you otherwise specifically agree in writing, each Credit, the opening of each Credit, the performance by you under each Credit, and the performance by the Beneficiary and any advising, confirming, negotiating, paying or other bank under each Credit, shall be governed by and be construed in accordance with the UCP in force on the date of the issuance of each Credit.

SECTION 19. **JURISDICTION AND SERVICE OF PROCESS**. Any suit, action or proceeding under or with respect to any L/C Document may be brought in (a) the courts of the State of New York, (b) the United States District Courts in New York, (c) the courts of Applicant's jurisdiction of incorporation or principal office, or (d) the courts of the jurisdiction where any Beneficiary, any advising, confirming, negotiating, paying or other bank, or any other person or entity has brought any suit, action or proceeding against you with respect to any Credit or any Demand, and

the parties hereto hereby submit to the nonexclusive jurisdiction of such courts for the purpose of any such suit, action, proceeding or judgment and waives any other preferential jurisdiction by reason of domicile. Applicant will accept joinder in any suit, action or proceeding brought in any court or jurisdiction against you by any Beneficiary, any advising, confirming, negotiating, paying or other bank or any other person or entity with respect to any Credit or any Demand. The parties hereto irrevocably waive trial by jury and any objection, including, without limitation, any objection of the laying of venue or any objection based on the grounds of *forum non conveniens*, which such party may now or hereafter have to the bringing of any such action or proceeding. Applicant further waives any right to transfer or change the venue from any court described above of any suit, action or proceeding brought against Applicant by you under or in connection with any L/C Document. The parties hereto irrevocably consent to the service of process in any action or proceeding in any court by the mailing of copies thereof by registered or certified mail, postage prepaid, to such party at its address specified next to its signature on this Agreement or at such other address as such party shall have notified to the other party in writing, such service to be effective ten (10) days after such mailing.

SECTION 20. **JOINT APPLICANTS**. If this Agreement is signed by more than one person and/or entity as an Applicant, this Agreement and the Applications shall be the joint and several agreement of all such persons and/or entities and that all references to "Applicant" or "Applicant's" in this Agreement and the Applications shall refer to all such persons and/or entities jointly and severally.

SECTION 21. **SEVERABILITY**. Any provision of any L/C Document which is prohibited or unenforceable in any jurisdiction shall be, only as to such jurisdiction, ineffective to the extent of such prohibition or unenforceability, but all the remaining provisions of such L/C Document and all the other L/C Documents shall remain valid.

SECTION 22. **HEADINGS**. The headings used in this Agreement are for convenience of reference only and shall not define or limit the provisions of this Agreement.

SECTION 23. **COMPLETE AGREEMENT**. The entire agreement with respect to each Credit will consist of the Loan Documents (including, without limitation, the Credit Agreement), this Agreement and the Application for such Credit, together with any written document or instrument signed by Applicant and you, or signed by Applicant and approved by you, which specifically references such Credit, the Application for such Credit, or this Agreement. Except as specifically provided in this Agreement (including, without limitation, Section 24 of this Agreement), in any Application or in any written document or instrument referred to in the preceding sentence, no statements or representations not contained in the Loan Documents, this Agreement, such Application or such written document or instrument or instrument.

SECTION 24. **INCONSISTENT PROVISIONS**. To the extent that any of the terms or provisions of this Agreement or any other L/C Document are inconsistent or conflict with, or otherwise violate, any of the terms or provisions of the Credit Agreement, the terms and provisions of the Credit Agreement shall control and be applicable. The phrases "notwithstanding any other provision of this Agreement" and "notwithstanding any other provision of any L/C Document or any Loan Document" and words of similar import used in this Agreement or any other L/C Document shall be deemed to be subject to the provisions of this Section 24.

SECTION 25. **COUNTERPARTS**. This Agreement may be executed in any number of counterparts and by the parties hereto on separate counterparts and each such counterpart shall be deemed to be an original, but all such counterparts shall together constitute but one and the same Agreement.

GREAT LAKES DREDGE & DOCK CORPORATION WELLS FARGO BANK, NATIONAL ASSOCIATION By: Name: Name: Deborah A. Wensel Title: _ Title: Senior Vice President, Chief Financial Officer and Treasurer Address: 1000 Louisiana St., 4th Floor Address: Great Lakes Dredge & Dock Corporation Houston, TX 77002, USA 2122 York Road Attention: Scott J. Gildea Oak Brook, IL 60523 Telephone No.: (713) 319-1374 Attention: Deborah A. Wensel

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered by the parties hereto as of the date first written above.

Fax No. (713) 739-1087

DATE:

9 of 9

DATE:

Telephone No.:(630) 574-2948 Fax No. (630) 574-3007

EXHIBIT "I"

REAFFIRMATION, RATIFICATION AND ASSUMPTION AGREEMENT

(International)

See attached.

1

Attachment to Exhibit I

FORM OF REAFFIRMATION, RATIFICATION AND ASSUMPTION AGREEMENT

THIS REAFFIRMATION, RATIFICATION AND ASSUMPTION AGREEMENT (this "<u>Agreement</u>"), dated as of ______, 200_, is made by Great Lakes Dredge & Dock Corporation (formerly named Great Lakes Dredge & Dock Holdings Corp.), a Delaware corporation ("<u>Successor Borrower</u>"), in favor of Wells Fargo HSBC Trade Bank, N.A., and the Secured Party (as defined in the International Security Agreement).

RECITALS:

- A. Great Lakes Dredge & Dock Corporation, a Delaware corporation (the "<u>Original Borrower</u>"), and Bank are parties to that certain International Letter of Credit Agreement dated as of September ___, 2006 (as amended, restated, supplemented or otherwise modified, the "<u>Credit Agreement</u>"). Capitalized terms used and not otherwise defined herein shall have the meanings ascribed to them in the Credit Agreement.
- B. The Original Borrower is also a party to (i) the Note, (ii) the International Security Agreement, (iii) the Standby Letter of Credit Agreement, (iv) the Borrower Agreement, (v) the Fast Track Borrower Agreement Supplement, (vi) the Acknowledgement of Country Limitation Schedule, and (vii) other International Loan Documents executed and delivered in connection with the Original Borrower's Obligations under the Credit Agreement.
- C. Effective as of even date herewith, the Original Borrower has merged with and into the Successor Borrower, with the Successor Borrower being the surviving Person of such merger (the "Borrower Merger").
- D. In order to obtain the consent of the Bank to the Borrower Merger, the Successor Borrower is required, among other things, to expressly assume the obligations of the Original Borrower under, and reaffirm and ratify the effectiveness of, (i) the Credit Agreement, (ii) the Note, (iii) the International Security Agreement, (iv) the Standby Letter of Credit Agreement, (v) the Borrower Agreement, (vi) the Fast Track Borrower Agreement Supplement, (vii) the Acknowledgement of Country Limitation Schedule, and (viii) other International Loan Documents executed and delivered in connection with the Original Borrower's Obligations under the Credit Agreement (collectively, the "Assumed Borrower Agreements").
 - E. Successor Borrower is executing and delivering this Agreement in order to satisfy such requirements.

NOW, THEREFORE, in consideration of the foregoing and to induce the Bank to consent to the Borrower Merger, Successor Borrower hereby agrees as follows:

1. Assumption of Liabilities.

- (a) As the successor by merger to the Original Borrower, Successor Borrower hereby expressly assumes all indebtedness, liabilities and obligations of the Original Borrower under the Assumed Borrower Agreements.
- (b) Any and all Liens, encumbrances, Collateral, security interests, UCC financing statements, transfers and any and all assignments of any right, claim or interest in and to property of any nature whatsoever heretofore given or granted to the Bank or any of the other Secured Parties by the Original Borrower are expressly assumed by Successor Borrower, and shall continue without interruption, in full force and effect, surviving the Borrower Merger and shall apply to any and all such property (i) owned by the Original Borrower prior to the Borrower Merger, (ii) acquired by Successor Borrower as a result of the Borrower Merger, or (iii) acquired hereafter by Successor Borrower.
- 2. **Reaffirmation and Ratification of Loan Documents**. Successor Borrower (i) hereby adopts, ratifies, reaffirms and confirms all of the Assumed Borrower Agreements executed by the Original Borrower, (ii) agrees that all Assumed Borrower Agreements shall remain in full force and effect, (iii) shall be substituted as a party to all such Assumed Borrower Agreements with the same force and effect as if Successor Borrower were originally a party thereto and (iv) makes all of the representations and warranties under the Assumed Borrower Agreements as a "Borrower" and a "Grantor" as defined thereunder.
- 3. **Representations and Warranties**. To induce the Bank to consent to the Borrower Merger, Successor Borrower represents and warrants to the Bank as follows:
- (a) Existence; Authority; Compliance with Law. Successor Borrower (i) is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware; (ii) has the power to own its Properties and to carry on its business as now conducted; (iii) is duly qualified to do business and is in good standing in all jurisdictions in which the failure to so qualify or be in good standing could reasonably be expected to have a Material Adverse Effect, (iv) has the power and authority to make, execute, deliver and carry out this Agreement and the Assumed Borrower Agreements to which it is a party (by assumption or otherwise); and (v) has complied in all material respects with all provisions of all applicable laws and regulations, including those relating to Successor Borrower's ownership of real or personal property, the conduct and licensing of Successor Borrower's business, the payment and withholding of taxes, ERISA and other employee matters, safety and environmental matters.
- (b) **Authorization; No Contravention**. The execution, delivery and performance by Successor Borrower of this Agreement and performance by Successor Borrower of its obligations under the Assumed Borrower Agreements to which it is a party (whether by assumption or otherwise) have been duly authorized by all necessary proceedings on its part and do not and will not (i) conflict with or result in a breach of any of the terms, conditions or provisions of, or constitute a default under, or result in any violation of, or result in the creation or imposition of any Lien on any of the Property of the Borrower or any Subsidiary (other than Liens created under the International Loan Documents and, in connection with the execution, delivery and performance of any Export Order, Permitted Liens) pursuant to (a) the Organization

Documents of the Successor Borrower; (b) any Governmental Requirement; or (c) the terms, conditions or provisions of any material contract to which the Successor Borrower is a party or by which it is bound or to which it is subject.

- (c) **No Consent**. No authorization or approval or other action by, and no notice to or filing with, any Person or any Governmental Authority is required for the due execution, delivery, and performance of this Agreement or any of the Assumed Borrower Agreements except (i) filings and recordings to perfect the Liens created under the International Loan Documents and (ii) those which, if not obtained, could not reasonably be expected to have a Material Adverse Effect.
 - (d) No Default. No Default or Event of Default has occurred and is continuing.
- (e) **Binding Effect**. This Agreement, the Credit Agreement and each other Assumed Borrower Agreements to which Successor Borrower is a party (whether by assumption or otherwise) constitute the validly and legally binding obligations of Successor Borrower, enforceable against Successor Borrower in accordance with their respective terms, except as limited by Debtor Laws. The Liens created by the International Security Documents constitute valid and perfected Liens on the Collateral described therein superior in right to all other Liens, except for Permitted Liens.
- (f) **Representations and Warranties**. The representations and warranties set forth in the Credit Agreement and the other Assumed Borrower Agreements are true and correct in all material respects on and as of the date hereof, both before and after giving effect to the Borrower Merger, as if such representations and warranties were being made on and as of the date hereof, except for any representation or warranty which is specified as being made as of an earlier date, in which case such representation or warranty shall only speak to such earlier date.
- (g) **Perfection of Liens**. All filings, registrations and recordings necessary or appropriate to create, preserve, protect and perfect the security interests granted by the Original Borrower to the Bank and hereby assumed by Successor Borrower in respect of the Collateral owned by Successor Borrower have been accomplished and the Liens and security interests granted to the Bank pursuant to the International Security Documents in and to such Collateral constitute a perfected security interest therein superior and prior to the rights of all other Persons therein and subject to no other Liens (other than Liens permitted by the Credit Agreement), and will be entitled to all the rights, priorities and benefits afforded by the Uniform Commercial Code or other relevant law as enacted in any relevant jurisdiction to perfected security interests.
- (h) **Chief Executive Office; Records**. The chief place of business and chief executive office of Successor Borrower and the office where Successor Borrower keeps the originals of all records concerning Successor Borrower's Export-Related Accounts Receivable and all originals of all chattel paper which evidence the Export-Related Accounts Receivable are and will remain located at the chief executive offices listed on <u>Schedule 4</u> to the International Security Agreement. The organizational number of Successor Borrower is as set forth on <u>Schedule A</u> attached hereto.

- 5. **Further Assurances.** Successor Borrower agrees to execute and deliver to Bank any and all documents requested by Bank and reasonably necessary to continue, without interruption, the Liens and security interests described in the International Security Documents hereby assumed and ratified by Successor Borrower, including, without limitation, UCC financing statements and amendments to reflect the Borrower Merger.
- 6. APPLICABLE LAW. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK, WITHOUT REGARD TO CONFLICTS OF LAW PRINCIPLES (OTHER THAN THE PROVISIONS OF 5-1401 AND 5-1402 OF THE NEW YORK GENERAL OBLIGATIONS LAW).

[Remainder of Page Intentionally Left Blank; Signature Page Follows]

IN WITNESS WHEREOF, the undersigned has caused this Agreement to be duly exas of the date first written above.	ecuted and delivered by its proper and duly authorized officer effective
	GREAT LAKES DREDGE & DOCK CORPORATION (formerly named Great Lakes Dredge & Dock Holdings Corp.)
	By: Name: Title:

SCHEDULE A TO REAFFIRMATION, RATIFICATION AND ASSUMPTION AGREEMENT

EXHIBIT "J"

DEPOSIT ACCOUNT CONTROL AGREEMENT

 $\underline{(International)}$

See attached.

1

Attachment to Exhibit J



RESTRICTED ACCOUNT AGREEMENT

(Access Restricted Immediately)

This **Restricted Account Agreement** (the "<u>Agreement</u>"), dated as of the date specified on the initial signature page of this Agreement, is entered into by and among Great Lakes Dredge & Dock Company, LLC ("<u>Company</u>"), Wells Fargo HSBC Trade Bank N.A. ("<u>Secured Party</u>") and **Wells Fargo Bank, National Association** ("<u>Bank</u>"), and sets forth the rights of Secured Party and the obligations of Bank with respect to the deposit account(s) of Company at Bank identified at the end of this Agreement as the "Restricted Account(s)". As used in this Agreement, the term "<u>Restricted Account</u>" refers, individually and collectively, to each such deposit account.

- 1. Secured Party's Interest in Restricted Account. Secured Party represents that it is either (i) a lender who has extended credit to Company and has been granted a security interest in the Restricted Account or (ii) such a lender and the agent for a group of such lenders (the "Lenders"). Company hereby confirms, and Bank hereby acknowledges, the security interest granted by Company to Secured Party in all of Company's right, title and interest in and to the Restricted Account and all sums now or hereafter on deposit in or payable or withdrawable from the Restricted Account (the "Account Funds"). Except as specifically provided otherwise in this Agreement, Company has given Secured Party complete control over the Account Funds. Secured Party hereby appoints Bank as agent for Secured Party only for the purpose of perfecting the security interest of Secured Party in the Account Funds while they are in the Restricted Account. Company and Secured Party would like to use the Restricted Account Service of Bank described in this Agreement (the "Service") to further the arrangements between Secured Party and Company regarding the Restricted Account and the Account Funds.
- 2. Access to Restricted Account. Company agrees that it will not be able to make debits or withdrawals from the Restricted Account, that it will not have access to the Restricted Account or any Account Funds, and that Secured Party will have exclusive access to the Restricted Account and Account Funds, except as specifically provided in this Agreement or as specifically agreed by Secured Party in writing.
- **Balance Reports.** Bank agrees, at the telephone request of Secured Party on any Business Day (a day on which Bank is open to conduct its regular banking business, other than a Saturday, Sunday or public holiday), to make available to Secured Party a report ("Balance Report") showing the opening available balance in the Restricted Account as of the beginning of such Business Day, either on-line or by facsimile transmission, at Bank's option. Company expressly consents to this transmission of information. Secured Party and Company understand and agree that the opening available balance in the Restricted Account at the beginning of any Business Day will be determined after deducting from the Restricted Account the face amount of all Returned Items (as defined in Section 5 of this Agreement).
- 4. **Transfers From Restricted Account.** Unless otherwise instructed by Secured Party, Bank agrees that on each Business Day it will transfer to Company's account specified at the end of this Agreement with the bank specified at the end of this Agreement (the "Company Remote")

RestAcctAgmt-Std-NS (Revised 08/09/05)

Account") the full amount of each payment received provided the funds comprising each payment are collected funds, such transfer to be made on the same Business Day as received if received on or before 2:00 p.m. (Central time) and for each payment received after 2:00 p.m. (Central time), shall be transferred the next Business Day. Bank will use the Fedwire system to make each funds transfer unless for any reason the Fedwire system is unavailable, in which case Bank will determine the funds transfer system to be used in making each funds transfer and the means by which each transfer will be made. Bank, Secured Party and Company each agree that Bank will comply with instructions given to Bank by Secured Party directing disposition of funds in the Restricted Account without further consent by Company, subject otherwise to the terms of this Agreement and Bank's standard policies, procedures and documentation in effect from time to time governing the type of disposition requested. Except as otherwise required by law, Bank will not agree with any third party to comply with instructions for disposition of funds in the Restricted Account originated by such third party. So long as no Event of Default has occurred and is continuing under the Letter of Credit Agreement (defined below), Secured Party will deliver orders to Bank providing for a daily transfer of funds from the Restricted Account to the Company Remote Account in accordance with the terms hereof. As used herein, "Letter of Credit Agreement" means that certain International Letter of Credit Agreement dated as of September 29, 2006, by and among the Company, Great Lakes Dredge and Dock Corporation and Secured Party.

- 5. **Returned Item Amounts.** Secured Party and Company understand and agree that the face amount ("Returned Item Amount") of each Returned Item will be paid by Bank debiting the Restricted Account, without prior notice to Secured Party or Company. As used in this Agreement, the term "Returned Item" means (i) any item deposited to the Restricted Account and returned unpaid, whether for insufficient funds or for any other reason, and without regard to the timeliness of such return or the occurrence or timeliness of any drawee's notice of non-payment; (ii) any item subject to a claim against Bank of breach of transfer or presentment warranty under the Uniform Commercial Code, as adopted in the applicable state; (iii) any automated clearing house ("ACH") entry credited to the Restricted Account and returned unpaid or subject to an adjustment entry under applicable clearing house rules, whether for insufficient funds or for any other reason, and without regard to the timeliness of such return or adjustment; (iv) any credit to the Restricted Account from a merchant card transaction, against which a contractual demand for chargeback has been made; and (v) any credit to the Restricted Account made in error. Company agrees to pay all Returned Item Amounts immediately on demand, without setoff or counterclaim, to the extent there are not sufficient funds in the Restricted Account to cover the Returned Item Amounts on the day they are to be debited from the Restricted Account. Secured Party agrees to pay all Returned Item Amounts within thirty (30) calendar days after demand, without setoff or counterclaim, to the extent that (i) the Returned Item Amounts are not paid in full by Company within fifteen (15) calendar days after demand on Company by Bank, and (ii) Secured Party has received proceeds from the corresponding Returned Items.
- 6. [Reserved.]
- 7. **Bank Fees.** Company agrees to pay all reasonable customary Bank's fees and charges for the maintenance and administration of the Restricted Account and for the treasury management and other account services provided with respect to the Restricted Account (collectively "Bank Fees"), including, but not limited to, the fees for (a) the Balance Reports provided on the Restricted Account, (b) the wire transfer services received with respect to the Restricted Account, (c) Returned Items, (d) funds advanced to cover overdrafts in the Restricted Account (but without Bank being in any way obligated to make any such advances), and (e) duplicate bank statements on the Restricted Account. The Bank Fees will be paid by Bank debiting the

RestAcctAgmt-Std-NS (Revised 08/09/05)

Restricted Account on the Business Day that the Bank Fees are due without notice to Secured Party or Company. If there are not sufficient funds in the Restricted Account to cover fully the Bank Fees on the Business Day they are to be debited from the Restricted Account, such shortfall or the amount of such Bank Fees will be paid by Company sending Bank a check in the amount of such shortfall or such Bank Fees, without setoff or counterclaim, within thirty (30) calendar days after written demand to Company by Bank. Secured Party agrees to pay the Bank Fees within forty-five (45) calendar days after demand, without setoff or counterclaim, to the extent such Bank Fees are not paid in full by Company within thirty (30) calendar days after demand on Company by Bank.

- **8. Account Documentation.** Secured Party and Company agree that, except as specifically provided in this Agreement, the Restricted Account will be subject to, and Bank's operation of the Restricted Account will be in accordance with, the terms and provisions of Bank's Commercial Account Agreement or other deposit account agreement governing the Restricted Account ("Account Agreement").
- **9. Bank Statements.** Bank will, upon receiving a written request from Secured Party, send to Secured Party by United States mail, at the address indicated for Secured Party after its signature to this Agreement, duplicate copies of all bank statements on the Restricted Account which are sent to Company. Company and/or Secured Party will have thirty (30) calendar days after receipt of a bank statement to notify Bank of an error in such statement. Bank's liability for such errors is limited as provided in the "Limitation of Liability" section of this Agreement.
- **10. Partial Subordination of Bank's Rights.** Bank hereby subordinates to the security interest of Secured Party in the Restricted Account (i) any security interest which Bank may have or acquire in the Restricted Account, and (ii) any right which Bank may have or acquire to set off or otherwise apply any Account Funds against the payment of any indebtedness from time to time owing to Bank from Company, except for debits to the Restricted Account permitted under this Agreement for the payment of Returned Item Amounts or Bank Fees.
- 11. Bankruptcy Notice; Effect of Filing. If Bank at any time receives notice of the commencement of a bankruptcy case or other insolvency or liquidation proceeding by or against Company (a "Bankruptcy Notice"), Bank will continue to comply with its obligations under this Agreement, except to the extent that any action required of Bank under this Agreement is prohibited under applicable bankruptcy laws or regulations or is stayed pursuant to the automatic stay imposed under the United States Bankruptcy Code or by order of any court or agency. With respect to any obligation of Secured Party hereunder which requires prior demand upon Company, the commencement of a bankruptcy case or other insolvency or liquidation proceeding by or against Company shall automatically eliminate the necessity of such demand upon Company by Bank, and shall immediately entitle Bank to make demand on Secured Party with the same effect as if demand had been made upon Company and the time for Company's performance had expired.
- 12. Legal Process, Legal Notices and Court Orders. Bank will comply with any legal process, legal notice or court order it receives if Bank determines in its sole discretion that the legal process, legal notice or court order is legally binding on it.
- 13. Indemnification for Following Instructions. Secured Party and Company each agree that, notwithstanding any other provision of this Agreement, Bank will not be liable to Secured Party or Company for any losses, liabilities, damages, claims (including, but not limited to, third party claims), demands, obligations, actions, suits, judgments, penalties, costs or expenses, including, but not limited to, attorneys' fees, (collectively, "Losses and Liabilities") suffered or

RestAcctAgmt-Std-NS (Revised 08/09/05)

incurred by Secured Party or Company as a result of or in connection with, (a) Bank complying with any binding legal process, legal notice or court order referred to in Section 12 of this Agreement, (b) Bank following any instruction or request of Secured Party, or (c) Bank complying with its obligations under this Agreement. Company will indemnify Bank against any Losses and Liabilities Bank may suffer or incur as a result of or in connection with any of the circumstances referred to in clauses (a) through (c) of this Section 13. To the extent not paid by Company within thirty (30) calendar days after written demand, Secured Party will indemnify Bank against any Losses and Liabilities Bank may suffer or incur as a result of or in connection with any of the circumstances referred to in clause (b) of this Section 13. Notwithstanding anything to the contrary contained herein, no party shall be entitled to the benefits of this Section 13 to the extent of any bad faith, gross negligence or willful misconduct by such party seeking indemnification.

- 14. No Representations or Warranties of Bank. Bank agrees to perform its obligations under this Agreement in a manner consistent with the quality provided when Bank performs similar services for its own account. However, Bank will not be responsible for the errors, acts or omissions of others, such as communications carriers, correspondents or clearinghouses through which Bank may perform its obligations under this Agreement or receive or transmit information in performing its obligations under this Agreement. Secured Party and Company also understand that Bank will not be responsible for any loss, liability or delay caused by wars, failures in communications networks, labor disputes, legal constraints, fires, power surges or failures, earthquakes, civil disturbances or other events beyond Bank's control. BANK MAKES NO EXPRESS OR IMPLIED REPRESENTATIONS OR WARRANTIES WITH RESPECT TO THE SERVICE OTHER THAN THOSE EXPRESSLY SET FORTH IN THIS AGREEMENT.
- 15. Limitation of Liability. Bank will not be responsible for any Losses and Liabilities due to any cause other than its own negligence or breach of this Agreement, in which case its liability to Secured Party and Company shall, unless otherwise provided by any law which cannot be varied by contract, be limited to direct money damages in an amount not to exceed ten (10) times all the Bank Fees charged or incurred during the calendar month immediately preceding the calendar month in which such Losses and Liabilities occurred (or, if no Bank Fees were charged or incurred in the preceding month, the Bank Fees charged or incurred in the month in which the Losses and Liabilities occurred). Company will indemnify Bank against all Losses and Liabilities suffered or incurred by Bank as a result of third party claims; provided, however, that Company shall not indemnify to extent such Losses and Liabilities arise from gross negligence, bad faith or willful misconduct from such party seeking indemnification. IN NO EVENT WILL BANK BE LIABLE FOR ANY SPECIAL, CONSEQUENTIAL, INCIDENTAL, INDIRECT OR PUNITIVE DAMAGES, WHETHER ANY CLAIM IS BASED ON CONTRACT OR TORT, WHETHER THE LIKELIHOOD OF SUCH DAMAGES WAS KNOWN TO BANK AND REGARDLESS OF THE FORM OF THE CLAIM OR ACTION, INCLUDING, BUT NOT LIMITED TO, ANY CLAIM OR ACTION ALLEGING GROSS NEGLIGENCE, WILLFUL MISCONDUCT, FAILURE TO EXERCISE REASONABLE CARE OR FAILURE TO ACT IN GOOD FAITH. Any action against Bank by Company or Secured Party under or related to this Agreement must be brought within twelve (12) months after the cause of action accrues.
- **16. Termination.** This Agreement and the Service may be terminated by Secured Party or Bank at any time by either of them giving thirty (30) calendar days prior written notice of such termination to the other parties to this Agreement at their contact addresses specified after their signatures to this Agreement; provided, however, that this Agreement and the Service may be terminated immediately upon written notice from Bank to Company and Secured Party should Company or Secured Party fail to make any payment when due to Bank from Company or Secured Party under the terms of this Agreement. Company's and Secured Party's obligation to

RestAcctAgmt-Std-NS (Revised 08/09/05)

report errors in funds transfers and bank statements and to pay Returned Item Amounts and Bank Fees, as well as the indemnifications made, and the limitations on the liability of Bank accepted, by Company and Secured Party under this Agreement will continue after the termination of this Agreement and/or the closure of the Restricted Account with respect to all the circumstances to which they are applicable existing or occurring before such termination or closure, and any liability of any party to this Agreement, as determined under the provisions of this Agreement, with respect to acts or omissions of such party prior to such termination or closure will also survive such termination or closure. Upon any termination of this Agreement and the Service or closure of the Restricted Account all collected and available balances in the Restricted Account on the date of such termination or closure will be transferred to Secured Party as requested by Secured Party in writing to Bank.

- **Modifications, Amendments, and Waivers.** This Agreement may not be modified or amended, or any provision thereof waived, except in a writing signed by all the parties to this Agreement.
- 18. Notices. All notices from one party to another shall be in writing, or be made by a telecommunications device capable of creating a written record, shall be delivered to Company, Secured Party and/or Bank at their contact addresses specified after their signatures to this Agreement, or any other address of any party notified to the other parties in writing, and shall be effective upon receipt. Any notice sent by a party to this Agreement to another party shall also be sent to all other parties to this Agreement. Bank is authorized by Company and Secured Party to act on any instructions or notices received by Bank if (a) such instructions or notices purport to be made in the name of Secured Party, (b) Bank reasonably believes that they are so made, and (c) they do not conflict with the terms of this Agreement as such terms may be amended from time to time, unless such conflicting instructions or notices are supported by a court order.
- 19. Successors and Assigns. Neither Company nor Secured Party may assign or transfer its rights or obligations under this Agreement to any person or entity without the prior written consent of Bank, which consent will not be unreasonably withheld. Bank may not assign its rights or obligations under this Agreement to any person or entity without the prior written consent of Secured Party, which consent will not be unreasonably withheld; provided, however, that no such consent will be required if such assignment or transfer takes place as part of a merger, acquisition or corporate reorganization affecting Bank.
- **20. Governing Law.** Company and Secured Party understand that Bank's provision of the Service under this Agreement is subject to federal laws and regulations. To the extent that such federal laws and regulations are not applicable this Agreement shall be governed by and be construed in accordance with the laws of the state in which the office of Bank that maintains the Restricted Account is located, without regard to conflict of laws principles.
- 21. Severability. To the extent that this Agreement or the Service to be provided under this Agreement are inconsistent with, or prohibited or unenforceable under, any applicable law or regulation, they will be deemed ineffective only to the extent of such prohibition or unenforceability and be deemed modified and applied in a manner consistent with such law or regulation. Any provision of this Agreement which is deemed unenforceable or invalid in any jurisdiction shall not affect the enforceability or validity of the remaining provisions of this Agreement or the same provision in any other jurisdiction.

RestAcctAgmt-Std-NS (Revised 08/09/05)

- **22. Counterparts.** This Agreement may be executed in any number of counterparts each of which shall be an original with the same effect as if the signatures thereto and hereto were upon the same instrument.
- **23. Entire Agreement.** This Agreement, together with the Account Agreement, contains the entire and only agreement among all the parties to this Agreement and between Bank and Company, and Bank and Secured Party, with respect to (a) the Service, (b) the interest of Secured Party and the Lenders in the Account Funds and the Restricted Account, and (c) Bank's obligations to Secured Party and the Lenders in connection with the Account Funds and the Restricted Account.

[SIGNATURE PAGE FOLLOWS]

RestAcctAgmt-Std-NS (Revised 08/09/05)

This Agreement has been signed by the duly authorized officers or representatives of Company, Secured Party and Bank on the date specified below.

Date: September 29, 2006

Restricted Account Number(s): 8252650166		
Company Remote Account:		
Bank of America		
Account No. 86661-01364		
ABA/Swift: 0260-0959-3 / BOFAUS3N		
Beneficiary Adress: 2122 York Road		
Chicago, Illinois 60523		
Account Name: Great Lakes Dredge & Dock Company, LLC		
Bank Address: 2231 S. LaSalle Street		
Chicago, Illinois 60697		
Great Lakes Dredge & Dock Company, LLC	Wells Fargo HSBC Trade Bank, N.A.	
Ву:	By:	
Name:	Name:	
Title:	Title:	
Address for Notices:	Address for Notices:	
2122 York Road	1000 Louisiana 4th Floor	
Oak Brook, Illinois	Houston, Texas	
60523	77002	
WELLS FARGO BANK, NATIONAL ASSOCIATION		
Ву:		
Name:		
Title:		
Address for Notices:		
1445 Ross Avenue		
Dallas, Texas		
75202		

RestAcctAgmt-Std-NS

(Revised 08/09/05)

Schedule 1.1

Existing Letters of Credit

	Amount	Term	LC#
ISSUED BY WELLS FARGO			
DIYAAR AL MUHARRAQ-AHLI UNITED BANK-MANAMA, BAHRAIN	\$ 1,633,919.81	09/18/06 - 12/01/06	NZS580323

Schedule 6.1

Subsidiaries

1. Great Lakes Dredge & Dock Corporation

100% of stock owned by GLDD Acquisitions Corp. Incorporated in the State of Delaware, July 31, 1991

2. Great Lakes Dredge & Dock Company, LLC

Sole member is Great Lakes Dredge & Dock Corporation

Organized in the State of Delaware, July 12, 2004

Great Lakes Dredge & Dock Company, LLC owns a 50% general partner interest in Amboy Aggregates Join Venture, a general partnership formed under New Jersey law

3. North American Site Developers, Inc.

Great Lakes Dredge & Dock Corporation owns (i) 800 shares of Common Stock, par value \$0.01 per share, representing 100% of the issued and outstanding shares of Common Stock, and (ii) 50 shares of Non-Voting Common Stock, par value \$0.01 per share, representing 25% of the issued and outstanding shares of Non-Voting Common Stock. Christopher Berardi owns Non-Voting Common Stock that represents 75% of the outstanding Non-Voting Common Stock and 15% of the fully-diluted common stock of North American Site Developers, Inc.

Incorporated in the State of Massachusetts, October 8, 1976

4. Fifty-Three Dredging Corporation

100% of stock owned by Great Lakes Dredge & Dock Company, LLC

Incorporated in the State of New Jersey, April 21, 1982

5. Dawson Marine Services Company

100% of stock owned by Great Lakes Dredge & Dock Corporation

Incorporated in the State of Delaware, March 23, 1987

6. Lydon Dredging & Construction Company, LTD

100% of stock owned by Great Lakes Dredge & Dock Company, LLC

Incorporated in Ontario, Canada

7. Great Lakes Caribbean Dredging, Inc.

100% of stock owned by Great Lakes Dredge & Dock Corporation

Incorporated in the State of Delaware, February 10, 2000

8. JDC Soil Management & Development, Inc.

100% of stock owned by Great Lakes Dredge & Dock Corporation

Incorporated in the State of Massachusetts, February 20, 2004.

9. NASDI Holdings Corporation

100% of stock owned by Great Lakes Dredge & Dock Corporation Incorporated in the State of Delaware, December 1, 2005

10. Other

Shell corporations in foreign jurisdictions that are used for bidding purposes and own no material assets.

Litigation

- 1. On April 24, 2006, a class action complaint was filed in the US District Court for the Eastern District of Louisiana, on behalf of citizens of Louisiana who suffered property damage from the waters that flooded New Orleans after Hurricane Katrina hit the area. The suit names Great Lakes Dredge & Dock Company, LLC (the "Company") along with numerous other dredging companies who have completed projects on behalf of the Army Corps of Engineers in the Mississippi River Gulf Outlet ("MRGO"), and the Federal government as defendants. The complaint alleges that dredging of MRGO caused the destruction of the Louisiana wetlands which provide a natural barrier against storms and hurricanes. This loss of natural barriers then contributed to the failure of the levees upon the impact of Hurricane Katrina which allowed the floodwaters to damage plaintiffs' property. Among other assertions is negligence in violation of the Water Pollution Control Act. The amount of damages is not stated but would be presumed to be significant. The Company believes it has meritorious defenses and believes this claim will not have a material adverse impact on the Company's business.
- 2. On February 10, 2004, the Company was served with a subpoena to produce documents in connection with a federal grand jury convened in the United States District Court for the District of South Carolina. The Company believes the grand jury has been convened to investigate the United States dredging industry in connection with work performed for the U.S. Army Corp of Engineers. The Company complied with the Justice Department's requests and Company management believes that it has provided substantially all of the documents that have been requested to this point. In addition to the documents requested, seven employees or former employees of the Company have now been interviewed or have testified before the grand jury.
- 3. On January 19, 2005, the Company, along with its joint-venture partners on the Port of Los Angeles Deepening Project, received a request for information from the United States Environmental Protection Agency ("EPA") pursuant to section 308(a) of the Clean Water Act. The EPA is investigating alleged dredging of unauthorized material and unauthorized discharge of that material at various locations in federally regulated waters of the U.S. relating to this project. The Company performed this project under a contract with the Los Angeles district of the Corps and believes it was in compliance with the contract specifications. The Company has complied with the request for information and has received no further response from the EPA. It is management's understanding that the Corps is currently addressing this matter with the EPA in Washington, as this is a national policy issue. Therefore, the Company believes the EPA request it has received will not result in any further action against the Company.

Schedule 6.3

Litigation

4. The Company or its former subsidiary, NATCO Limited Partnership, are named as defendants in approximately 280 lawsuits, the majority of which were filed between 1989 and 2000, and 18 of which were filed in the last three years. In these lawsuits, the plaintiffs allege personal injury, primarily fibrosis or asbestosis, from exposure to asbestos on our vessels. The vast majority of these lawsuits have been filed in the Northern District of Ohio and a few in the Eastern District of Michigan. All of the cases filed against the Company prior to 1996 were administratively dismissed in May 1996 and any cases filed since that time have similarly been administratively transferred to the inactive docket. Plaintiffs in these cases could seek to reinstate the cases at a future date without being barred by the statute of limitations. However, to date, no plaintiffs with claims against the Company have sought reinstatement. There are therefore no active pending cases against the Company. The Company believes that these cases will not have a material adverse impact on the Company's business.

Schedule 6.5

Capital Stock; Controlling Affiliates

I. Capital Stock of Great Lakes Dredge & Dock Corporation

The following are the names and ownership percentages (stated both on an outstanding basis and fully-diluted basis) of each legal, record owner of each class of the issued and outstanding shares of Capital Stock of Great Lakes Dredge & Dock Corporation, and of all issued and outstanding warrants, options, stock appreciation rights and other convertible interests with respect to Great Lakes Dredge & Dock Corporation:

	Shares		Shares Convertible				Fully Diluted	Outstanding
<u>Name</u>	Preferred	Common	Interests	Total	Percentage	Percentage		
GLDD Acquisitions Corp.	0	1000	0	1000	100%	100%		

II. Controlling Affiliates

GLDD Acquisitions Corp. owns all of the outstanding Capital Stock of Great Lakes Dredge & Dock Corporation. As of the Closing Date, the outstanding Capital Stock of GLDD Acquisitions Corp is owned as follows:

A. Series A Preferred Stock

Holder	Date Issued	Number of Shares
Madison Dearborn Capital Partners IV, L.P.	12/22/2003	76,807.55
Special Co-Invest I	12/22/2003	419.83
Randolph Street Partners VI	12/22/2003	227.18
Northwestern University	12/22/2003	45.44

Capital Stock; Controlling Affiliates

B. Series B Preferred Stock

<u>Holder</u>	Date Issued	Number of Shares
Steven R. Auernhamer	12/22/2003	9.00
Steven R. Becker	12/22/2003	32.00
David C. Cizek	12/22/2003	9.00
Arthur S. Fletcher	12/22/2003	9.00
James C. Gillespie	12/22/2003	32.00
William E. Hannum	12/22/2003	20.00
Bradley T.J. Hansen	12/22/2003	50.00
William H. Hanson	12/22/2003	16.00
Patrick C. Hughes	12/22/2003	9.00
Kyle D. Johnson	12/22/2003	32.00
John F. Karas	12/22/2003	32.00
Richard M. Lowry	12/22/2003	851.00
Donald J. Luce	12/22/2003	20.00
Michael J. Lueders	12/22/2003	9.00

Capital Stock; Controlling Affiliates

B. Series B Preferred Stock

<u>Holder</u>	Date Issued	Number of Shares
Robert F. Mackay	12/22/2003	9.00
Christopher T. Mackie 1998 Trust	12/22/2003	20.00
Douglas B. Mackie	12/22/2003	751.00
Kathleen J. Mackie 1998 Trust	12/22/2003	20.00
Madeline C. Mackie 1998 Trust	12/22/2003	20.00
Natalie A. Mackie 1998 Trust	12/22/2003	20.00
Philip D. Mackie 1998 Trust	12/22/2003	20.00
John R. Maszczyk	12/22/2003	9.00
Sam R. Morrison	12/22/2003	9.00
William A. Murchison	12/22/2003	9.00
John T. O'Brien	12/22/2003	20.00
Steven F. O'Hara	12/22/2003	50.00
William F. Pagendarm	12/22/2003	50.00
Seann M. Perez	12/22/2003	9.00

Capital Stock; Controlling Affiliates

B. Series B Preferred Stock

<u>Holder</u>	Date Issued	Number of Shares
Robert C. Ramsdell	12/22/2003	9.00
T. Christopher Roberts	12/22/2003	9.00
Michael R. Sayer	12/22/2003	9.00
David E. Simonelli	12/22/2003	46.00
George T. Strawn	12/22/2003	9.00
Mark R. Thomas	12/22/2003	9.00
Russell F. Zimmerman	12/22/2003	9.00
Madison Dearborn Capital Partners IV, L.P.	12/22/2003	7,125.00
Deborah A. Wensel Living Trust, dated December 7, 2004	12/7/2004	100.00

Capital Stock; Controlling Affiliates

C. Company Common Stock

<u>Holder</u>	Date Issued	Number of Shares
Madison Dearborn Capital Partners IV, L.P.	12/22/2003	843,044.83
Special Co-Invest I	12/22/2003	4,216.91
Randolph Street Partners VI	12/22/2003	2,281.88
Northwestern University	12/22/2003	456.38
Steven R. Auernhamer	12/22/2003	1,100.00
Steven W. Becker	12/22/2003	3,800.00
David C. Cizek	12/22/2003	1,100.00
Arthur S. Fletcher	12/22/2003	1,100.00
James C. Gillespie	12/22/2003	3,800.00
William E. Hannum	12/22/2003	2,500.00
Bradley T.J. Hansen	12/22/2003	6,000.00
William H. Hanson	12/22/2003	1,900.00
Patrick C. Hughes	12/22/2003	1,100.00

Capital Stock; Controlling Affiliates

C. Company Common Stock

<u>Holder</u>	Date Issued	Number of Shares
Kyle D. Johnson	12/22/2003	3,800.00
John F. Karas	12/22/2003	3,800.00
Richard M. Lowry	12/22/2003	31,900.00
Donald J. Luce	12/22/2003	2,500.00
Michael J. Lueders	12/22/2003	1,100.00
Robert F. Mackay	12/22/2003	1,100.00
Christopher T. Mackie 1998 Trust	12/22/2003	2,000.00
Douglas B. Mackie	12/22/2003	21,900.00
Kathleen J. Mackie 1998 Trust	12/22/2003	2,000.00
Madeline C. Mackie 1998 Trust	12/22/2003	2,000.00
Natalie A. Mackie 1998 Trust	12/22/2003	2,000.00
Philip D. Mackie 1998 Trust	12/22/2003	2,000.00
John R. Maszczyk	12/22/2003	1,100.00
Sam R. Morrison	12/22/2003	1,100.00
William A. Murchison	12/22/2003	1,100.00

Capital Stock; Controlling Affiliates

C. Company Common Stock

<u>Holder</u>	Date Issued	Number of Shares
John T. O'Brien	12/22/2003	2,500.00
Steven F. O'Hara	12/22/2003	6,000.00
William F. Pagendarm	12/22/2003	6,000.00
Seann M. Perez	12/22/2003	1,100.00
Robert C. Ramsdell	12/22/2003	1,100.00
T. Christopher Roberts	12/22/2003	1,100.00
Michael R. Sayer	12/22/2003	1,100.00
David E. Simonelli	12/22/2003	5,400.00
George T. Strawn	12/22/2003	1,100.00
Mark R. Thomas	12/22/2003	1,100.00
Russell F. Zimmerman	12/22/2003	1,100.00
Deborah A. Wensel Living Trust, dated December 7, 2004	12/7/2004	17,000.00

Schedule 8.3

Insurance

<u>Coverage</u> Marine Hull	Effective Date 31-Jul-06	Expiration Date 31-Jul-07	Insurer Lloyd's Syndicates, British Companies	Policy Number LH00648188	Maximum Limit \$43,680,000	_Maxi	mum Deductible 500,000
Increased Value	31-Jul-06	31-Jul-07	Lloyd's Syndicate No. 2003 (Catlin)	LH0648186	\$10,920,000	\$	500,000
Hull & Machinery (Bahrain)	31-Jul-06	31-Jul-07	Solidarity	t.b.a.	\$16,800,000	\$	500,000
Hull War Risks	31-Jul-06	31-Jul-07	Lloyd's Syndicates	LH0648185	\$21,000,000	\$	1,000,000
Business Property and Commercial Inland Marine	31-Jul-06	31-Jul-07	Travelers Property Casualty Company of America	QT-660-307D565A-TIL- 06	\$10,000,000	\$100 Prop (exc	ipment: 0,000 perty: \$10,000 ept flood/ nquake)
Commercial General Liability	1-Oct-05	1-Oct-06	American Home Assurance Company	GL 155-81-29	\$ 250,000	\$750 insureter	
Automobile Liability (All Other States)	1-Oct-05	1-Oct-06	American Home Assurance Company	CA 826-21-72	\$ 1,000,000	\$	500,000
Automobile Liability (Massachusetts)	1-Oct-05	1-Oct-06	American Home Assurance Company	CA 826-21-73	\$ 1,000,000	\$	500,000
Automobile Liability (Texas)	1-Oct-05	1-Oct-06	American Home Assurance Company	CA 826-21-74	\$ 1,000,000	\$	500,000
Workers' Compensation and Employer's Liability	1-Oct-05	1-Oct-06	American Home Assurance Company	WC 134-64-00	\$ 1,000,000	\$	500,000
Workers' Compensation and Employer's Liability (California)	1-Oct-05	1-Oct-06	American Home Assurance Company	WC 134-63-99	\$ 1,000,000	\$	500,000

U.S. Longshore and Harbor Workers' Compensation	1-Oct-05	1-Oct-06	Signal Mutual Indemnity Association, Ltd.	60045	\$ 10,000,000	\$ (
Workers' Compensation and Employer's Liability (Foreign)	1-Oct-05	1-Oct-06	AIG World Risk	WR10001555	\$ 1,000,000	\$ (
Excess Liability	31-Jul-06	31-Jul-07	Lloyd's & British Companies	LH0648183	\$99,000,000 excess of \$1,000,000	Underlying
Excess Liability	31-Jul-06	31-Jul-07	X.L. Insurance Company Ltd.	GB00007874EL06A	\$50,000,000 excess of \$100,000,000	Underlying
Protection & Indemnity	20-Feb-06	20-Feb-07	The West of England Ship Owners Mutual Insurance Association (Luxembourg)	00-0/2541P-1- 20/02/2006	Per Association Rules (approx. \$2.25 billion)	\$ 500,000
Commercial Crime	15-Dec-05	15-Dec-06	St. Paul Fire & Marine Insurance Company		\$ 5,000,000	\$ 125,000
Fiduciary Liability	15-Dec-05	15-Dec-06	St. Paul Mercury Insurance Company		\$ 5,000,000	
Directors' & Officers' Liability	5-Dec-05	5-Dec-06	St. Paul Mercury Insurance Company		\$ 10,000,000	\$ 500,000
Excess Directors' & Officers' Liability	5-Dec-05	5-Dec-06	Twin City Fire Insurance Company		\$10,000,000 excess of \$10,000,000	Underlying
Excess Directors' & Officers' Liability	5-Dec-05	5-Dec-06	XL Specialty Insurance Company		\$5,000,000 excess of \$20,000,000	Underlying
Marine Cargo			Lloyd's		Values shipped	\$ 50,000

Builder's Risk			Lloyd's, et al	Lloyd's - part of Hull policy	As required by contract; must be declared & negotiated	\$100,000
Contractor's All Risks			Lloyd's		As required by contract	
Loss of Hire			Lloyd's, et al	Part of Hull policy	As required by contract; must be declared & negotiated	
Contractor's Pollution Liability (NASDI)	1-Oct-05	1-Oct-06	American International Specialty Lines Insurance Company (AIG)	CPO 1366920	\$ 5,000,000	\$ 25,000

Schedule 8.7

Export Orders Excluded from the Cash Collateral Account

21346 Al Sukhna, Egypt
21525 HIDD Terminal, Bahrain
21772 Panama Navigation Study, Panama
21775 Doha Airport, Qatar
21813 Sheraton Hotel, Qatar
21834 Doha Airport JV, Qatar
21860 Juffair, Bahrain
21867 Al Sharq Resort, Qatar